# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

**WASHINGTON, DC 20549** 

## FORM S-8 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933

# THE RUBICON PROJECT, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization) 20-8881738 (I.R.S. Employer Identification No.)

12181 Bluff Creek Drive, 4th Floor Los Angeles, CA 90094 (Address of Principal Executive Offices, Zip Code)

isocket, inc. 2009 Equity Incentive Plan The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan (Full title of the plans)

> Brian W. Copple General Counsel The Rubicon Project, Inc. 12181 Bluff Creek Drive, 4<sup>th</sup> Floor Los Angeles, CA 90094 (Name and address of agent for service)

> > (310) 207-0272

(Telephone number, including area code, of agent for service)

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer 

Non-accelerated filer 

(Do not check if a smaller reporting company)

Smaller reporting company

#### **CALCULATION OF REGISTRATION FEE**

Title of Securities to be Registered	Amount to be Registered(2)	Proposed Maximum Offering Price Per Share	Proposed Maximum Aggregate Offering Price	Amount of Registration Fee
Common Stock of The Rubicon Project, Inc. issuable upon exercise of assumed stock options under the isocket inc. 2009 Equity				
Incentive Plan(1)	318,685	\$4.31(3)	\$1,373,532.35(3)	\$159.60
Common Stock of The Rubicon Project, Inc. to be issued pursuant to				
outstanding and future awards under The Rubicon Project, Inc.				
2014 Inducement Grant Equity Incentive Plan	1,000,000	\$15.19(4)	\$15,190,000(4)	\$1,765.08

- (1) In connection with the merger of a wholly owned subsidiary of The Rubicon Project, Inc. (the "Company" or "Registrant") with and into isocket, inc. ("isocket"), the Registrant has assumed outstanding options granted under isocket's 2009 Equity Incentive Plan.
- (2) In addition to the number of shares of the common stock, par value \$0.00001 per share (the "Common Stock") of the Registrant stated above, pursuant to Rule 416 under the Securities Act of 1933, as amended (the "Securities Act"), this registration statement also covers any additional shares of Common Stock that become issuable by reason of any stock dividend, stock split, recapitalization or other similar transaction effected without receipt of consideration that increases the number of the Registrant's outstanding shares of Common Stock.
- (3) Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(h) under the Securities Act, based upon the weighted-average exercise price per share of the options assumed by the Registrant.
- (4) Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(c) and Rule 457(h) under the Securities Act, based upon the average of the high and low prices of the Registrant's Common Stock on the New York Stock Exchange on December 17, 2014.

#### PART I

#### INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The information required by this Part I has been omitted from this Registration Statement pursuant to the Note to Part I of Form S-8.

#### **PART II**

#### INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

#### Item 3. Incorporation of Documents by Reference.

The following documents filed by the Registrant with the Securities and Exchange Commission (the "Commission"), are incorporated by reference herein and shall be deemed to be a part hereof:

- 1. The prospectus dated April 1, 2014, filed by the Registrant with the Commission pursuant to Rule 424(b)(4) under the Securities Act, filed on April 2, 2014, relating to the Registration Statement on Form S-1, as amended (Registration No. 333-193739), which contains the Registrant's audited financial statements for the latest fiscal year for which such statements have been filed;
- 2. The Registrant's Quarterly Reports on Form 10-Q filed with the Commission on May 15, 2014, August 1, 2014 and October 31, 2014;
- 3. The Registrant's Current Report on Form 8-K filed with the Commission on November 17, 2014 (other than with respect to Item 7.01 and the related exhibit); and
- 4. The description of the Common Stock contained in the Registrant's Registration Statement on Form 8-A filed with the Commission under Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), on March 28, 2014, including any amendments or reports filed for the purpose of updating such description.

In addition, all documents filed by the Registrant pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date hereof, but prior to the filing of a post-effective amendment to this Registration Statement that indicates that all securities offered hereby have been sold or that deregisters all such securities then remaining unsold, shall be deemed to be incorporated by reference in this Registration Statement and to be a part hereof from the date of filing of such documents. Notwithstanding the foregoing, unless specifically stated to the contrary in such filing, none of the information that the Registrant discloses under Items 2.02 or 7.01 of any Current Report on Form 8-K that it may from time to time furnish to the Commission will be incorporated by reference into, or otherwise be included in or deemed to be a part of, this Registration Statement.

For purposes of this Registration Statement, any document or any statement contained in a document incorporated or deemed to be incorporated herein by reference shall be deemed to be modified or superseded to the extent that a subsequently filed document or a statement contained herein or in any other subsequently filed document that also is or is deemed to be incorporated herein by reference modifies or supersedes such document or such statement in such document. Any statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement. Subject to the foregoing, all information in this Registration Statement is so qualified in its entirety by the information appearing in the documents incorporated herein by reference.

#### Item 4. Description of Securities.

Not applicable.

#### Item 5. Interests of Named Experts and Counsel.

Not applicable.

#### Item 6. Indemnification of Directors and Officers.

Registrant is a Delaware corporation. Section 145 of the General Corporation Law of the State of Delaware (the "DGCL") allows for the indemnification of officers, directors and other corporate agents in terms sufficiently broad to indemnify such persons under certain circumstances for liabilities (including reimbursement for expenses incurred) arising under the Securities Act. Article X of the Registrant's Sixth Amended and Restated Certificate of Incorporation and Article VI of the Registrant's Amended and Restated Bylaws authorize indemnification of the Company's directors, officers, employees and other agents to the extent and under the circumstances permitted by the DGCL.

The Registrant's Amended and Restated Bylaws also expressly authorize the Registrant to carry directors' and officers' insurance providing indemnification to its directors and officers for some liabilities. The Registrant maintains directors' and officers' liability insurance.

In addition, the Registrant has entered into agreements to indemnify its directors and executive officers.

The above discussion of the DGCL and of the Registrant's Sixth Amended and Restated Certificate of Incorporation, Amended and Restated Bylaws, and indemnification agreements is not intended to be exhaustive and is qualified in its entirety by such statute, certificate of incorporation, bylaws and indemnification agreements.

#### Item 7. Exemption from Registration Claimed.

Not applicable.

#### Item 8. Exhibits.

The Exhibit Index immediately preceding the exhibits is incorporated herein by reference.

#### Item 9. Undertakings.

- (a) The undersigned Registrant hereby undertakes:
  - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
    - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
- (ii) To reflect in the prospectus any facts or events arising after the effective date of the Registration Statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the Registration Statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement; and
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the Registration Statement or any material change to such information in the Registration Statement;

provided, however, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in reports filed with or furnished to the Commission by the Registrant pursuant to Section 13 or Section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the Registration Statement;

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof; and
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by

reference in the Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

(c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Act, the Registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Los Angeles, State of California, on December 19, 2014.

#### The Rubicon Project, Inc.

By: /s/ Todd Tappin
Name: Todd Tappin

Title: Chief Operating Officer and Chief Financial Officer

#### POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints Frank Addante, Todd Tappin, Brian W. Copple, and Jonathan Feldman, and each of them, his or her true and lawful attorneys-in-fact and agents, each with full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign any and all amendments, including post-effective amendments, to this Registration Statement, and any registration statement relating to the offering covered by this Registration Statement and filed pursuant to Rule 462 under the Securities Act, and to file the same, with exhibits thereto and other documents in connection therewith, with the Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done, as fully to all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that each of said attorneys-in-fact and agents or their substitute or substitutes may lawfully so or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act, the following persons have signed this Registration Statement in the capacities and on the date(s) indicated.

Signature

Title

Date

/s/ Frank Addante

Chief Executive Officer and Chairman (Principal Executive Officer)

December 19, 2014

/s/ Gregory R. Raifman Gregory R. Raifman	President and Director	December 19, 2014
/s/ Todd Tappin Todd Tappin	Chief Operating Officer and Chief Financial Officer ( <i>Principal Financial Officer</i> )	December 19, 2014
/s/ David Day David Day	Chief Accounting Officer (Principal Accounting Officer)	December 19, 2014
/s/ Sumant Mandal Sumant Mandal	Director	December 19, 2014
/s/ Jarl Mohn Jarl Mohn	Director	December 19, 2014
/s/ Lisa L. Troe Lisa. L. Troe	_ Director	December 19, 2014
/s/ Robert J. Frankenberg Robert J. Frankenberg		December 19, 2014
/s/ Robert F. Spillane Robert F. Spillane	Director	December 19, 2014

#### EXHIBIT INDEX

Exhibit No.	Exhibit Description
4.1*	isocket, inc. 2009 Equity Incentive Plan.
4.2*	Form of Stock Option Agreement under the isocket, inc. 2009 Equity Incentive Plan.
4.3*	Form of Notice of Grant of Stock Options under the isocket, inc. 2009 Equity Incentive Plan.
4.4*	Form of Notice of Grant of Stock Purchase Right under the isocket, inc. 2009 Equity Incentive Plan.
4.5*	Form of Stock Purchase Agreement under the isocket, inc. 2009 Equity Incentive Plan.
4.6*	The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan.
4.7*	Form of Restricted Stock Unit Grant Notice under The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan.
4.8*	Form of Stock Option Grant Notice under The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan.
4.9*	Form of Restricted Stock Grant Notice under The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan.
5.1*	Opinion of Jonathan Feldman, Esq.
23.1*	Consent of Jonathan Feldman, Esq. (included in Exhibit 5.1).
23.2*	Consent of PricewaterhouseCoopers LLP.
24.1*	Power of Attorney (included on signature page hereto).

<sup>\*</sup> Filed herewith.

#### ISOCKET, INC. 2009 EQUITY INCENTIVE PLAN

#### 1. ESTABLISHMENT, PURPOSE AND TERM OF PLAN.

- 1.1 Establishment. The isocket, inc. 2009 Equity Incentive Plan (the "Plan") is hereby established effective as of August 4, 2009.
- 1.2 **Purpose.** The purpose of the Plan is to advance the interests of the Participating Company Group and its stockholders by providing an incentive to attract, retain and reward persons performing services for the Participating Company Group and by motivating such persons to contribute to the growth and profitability of the Participating Company Group. The Company intends that Awards granted pursuant to the Plan be exempt from or comply with Section 409A of the Code (including any amendments or replacements of such section), and the Plan shall be so construed.
- 1.3 **Term of Plan.** The Plan shall continue in effect until its termination by the Board; provided, however, that all Awards shall be granted, if at all, within ten (10) years from the earlier of the date the Plan is adopted by the Board or the date the Plan is duly approved by the stockholders of the Company.

#### 2. **DEFINITIONS AND CONSTRUCTION.**

- 2.1 **Definitions.** Whenever used herein, the following terms shall have their respective meanings set forth below:
  - (a) "Award" means an Option, Restricted Stock Purchase Right or Restricted Stock Bonus granted under the Plan.
- (b) "Award Agreement" means a written or electronic agreement between the Company and a Participant setting forth the terms, conditions and restrictions of the Award granted to the Participant.
- (c) "Board" means the Board of Directors of the Company. If one or more Committees have been appointed by the Board to administer the Plan, "Board" also means such Committee(s).
- (d) "Cause" means, unless such term or an equivalent term is otherwise defined with respect to an Award by the Participant's Award Agreement or written contract of employment or service, any of the following: (i) the Participant's theft, dishonesty, willful misconduct, breach of fiduciary duty for personal profit, or falsification of any Participating Company documents or records; (ii) the Participant's material failure to abide by a Participating Company's code of conduct or other policies (including, without limitation, policies relating to confidentiality and reasonable workplace conduct); (iii) the Participant's unauthorized use, misappropriation, destruction or diversion of any tangible or intangible asset or corporate opportunity of a Participating Company (including, without limitation, the

Participant's improper use or disclosure of a Participating Company's confidential or proprietary information); (iv) any intentional act by the Participant which has a material detrimental effect on a Participating Company's reputation or business; (v) the Participant's repeated failure or inability to perform any reasonable assigned duties after written notice from a Participating Company of, and a reasonable opportunity to cure, such failure or inability; (vi) any material breach by the Participant of any employment or service agreement between the Participant and a Participating Company, which breach is not cured pursuant to the terms of such agreement; or (vii) the Participant's conviction (including any plea of guilty or nolo contendere) of any criminal act involving fraud, dishonesty, misappropriation or moral turpitude, or which impairs the Participant's ability to perform his or her duties with a Participating Company.

- (e) "Change in Control" means, unless such term or an equivalent term is otherwise defined with respect to an Award by the Participant's Award Agreement or written contract of employment or service, the occurrence of any of the following:
- (i) an Ownership Change Event or a series of related Ownership Change Events (collectively, a "*Transaction*") in which the stockholders of the Company immediately before the Transaction do not retain immediately after the Transaction direct or indirect beneficial ownership of more than fifty percent (50%) of the total combined voting power of the outstanding securities entitled to vote generally in the election of Directors or, in the case of an Ownership Change Event described in Section 2.1(v)(iii), the entity to which the assets of the Company were transferred (the "*Transferee*"), as the case may be; or
  - (ii) the liquidation or dissolution of the Company.

For purposes of the preceding sentence, indirect beneficial ownership shall include, without limitation, an interest resulting from ownership of the voting securities of one or more corporations or other business entities which own the Company or the Transferee, as the case may be, either directly or through one or more subsidiary corporations or other business entities. The Board shall have the right to determine whether multiple sales or exchanges of the voting securities of the Company or multiple Ownership Change Events are related, and its determination shall be final, binding and conclusive.

- (f) "Code" means the Internal Revenue Code of 1986, as amended, and any applicable regulations and administrative guidelines promulgated thereunder.
- (g) "Committee" means the compensation committee or other committee or subcommittee of the Board duly appointed to administer the Plan and having such powers as shall be specified by the Board. Unless the powers of the Committee have been specifically limited, the Committee shall have all of the powers of the Board granted herein, including, without limitation, the power to amend or terminate the Plan at any time, subject to the terms of the Plan and any applicable limitations imposed by law.
  - (h) "Company" means isocket, inc., a Delaware corporation, or any successor corporation thereto.
- (i) "Consultant" means a person engaged to provide consulting or advisory services (other than as an Employee or a Director) to a Participating Company,

provided that the identity of such person, the nature of such services or the entity to which such services are provided would not preclude the Company from offering or selling securities to such person pursuant to the Plan in reliance on either the exemption from registration provided by Rule 701 under the Securities Act or, if the Company is required to file reports pursuant to Section 13 or 15(d) of the Exchange Act, registration on a Form S-8 Registration Statement under the Securities Act.

- (j) "Director" means a member of the Board.
- (k) "Disability" means the inability of the Participant, in the opinion of a qualified physician acceptable to the Company, to perform the major duties of the Participant's position with the Participating Company Group because of the sickness or injury of the Participant.
- (l) "Employee" means any person treated as an employee (including an Officer or a Director who is also treated as an employee) in the records of a Participating Company and, with respect to any Incentive Stock Option granted to such person, who is an employee for purposes of Section 422 of the Code; provided, however, that neither service as a Director nor payment of a director's fee shall be sufficient to constitute employment for purposes of the Plan. The Company shall determine in good faith and in the exercise of its discretion whether an individual has become or has ceased to be an Employee and the effective date of such individual's employment or termination of employment, as the case may be. For purposes of an individual's rights, if any, under the terms of the Plan as of the time of the Company's determination of whether or not the individual is an Employee, all such determinations by the Company shall be final, binding and conclusive as to such rights, if any, notwithstanding that the Company or any court of law or governmental agency subsequently makes a contrary determination as to such individual's status as an Employee.
  - (m) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (n) "Fair Market Value" means, as of any date, the value of a share of Stock or other property as determined by the Board, in its discretion, or by the Company, in its discretion, if such determination is expressly allocated to the Company herein, subject to the following:
- (i) If, on such date, the Stock is listed on a national or regional securities exchange or market system, the Fair Market Value of a share of Stock shall be the closing price of a share of Stock as quoted on the national or regional securities exchange or market system constituting the primary market for the Stock, as reported in <a href="The Wall Street Journal">The Wall Street Journal</a> or such other source as the Company deems reliable. If the relevant date does not fall on a day on which the Stock has traded on such securities exchange or market system, the date on which the Fair Market Value shall be established shall be the last day on which the Stock was so traded prior to the relevant date, or such other appropriate day as shall be determined by the Board, in its discretion.

- (ii) If, on such date, the Stock is not listed on a national or regional securities exchange or market system, the Fair Market Value of a share of Stock shall be as determined by the Board in good faith without regard to any restriction other than a restriction which, by its terms, will never lapse, and in a manner consistent with the requirements of Section 409A of the Code.
- (o) "Incentive Stock Option" means an Option intended to be (as set forth in the Award Agreement) and which qualifies as an incentive stock option within the meaning of Section 422(b) of the Code.
  - (p) "Insider" means an Officer, a Director or other person whose transactions in Stock are subject to Section 16 of the Exchange Act.
- (q) "Insider Trading Policy" means the written policy of the Company pertaining to the purchase, sale, transfer or other disposition of the Company's equity securities by Directors, Officers, Employees or other service providers who may possess material, nonpublic information regarding the Company or its securities.
- (r) "Net-Exercise" means a procedure by which (i) the Company will deduct from the total number of shares of Stock for which a Participant elects to exercise an Option the largest number of whole shares of Stock that does not exceed the aggregate exercise price therefor, (ii) the Participant will pay to the Company in cash the remaining balance of such aggregate exercise price, and (iii) the Participant will be issued a whole number of shares of Stock upon such exercise determined in accordance with the following formula:

N = X\*((A-B)/A), where

- "N" = the number of shares of Stock (rounded down to the nearest whole number) to be issued to the Participant upon exercise of the Option;
- "X" = the total number of shares with respect to which the Participant has elected to exercise the Option;
- "A" = the Fair Market Value of one (1) share of Stock determined on the exercise date; and
- "B" = the exercise price per share (as defined in the Participant's Award Agreement)
- (s) "Nonstatutory Stock Option" means an Option not intended to be (as set forth in the Award Agreement) or which does not qualify as an Incentive Stock Option.
  - (t) "Officer" means any person designated by the Board as an officer of the Company.

- (u) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option granted pursuant to the Plan.
- (v) "Ownership Change Event" means the occurrence of any of the following with respect to the Company: (i) the direct or indirect sale or exchange in a single or series of related transactions by the stockholders of the Company of more than fifty percent (50%) of the voting stock of the Company; (ii) a merger or consolidation in which the Company is a party; or (iii) the sale, exchange, or transfer of all or substantially all of the assets of the Company (other than a sale, exchange or transfer to one or more subsidiaries of the Company).
  - (w) "Parent Corporation" means any present or future "parent corporation" of the Company, as defined in Section 424(e) of the Code.
  - (x) "Participant" means any eligible person who has been granted one or more Awards.
  - (y) "Participating Company" means the Company or any Parent Corporation or Subsidiary Corporation.
  - (z) "Participating Company Group" means, at any point in time, all entities collectively which are then Participating Companies.
  - (aa) "Restricted Stock Award" means an Award of a Restricted Stock Bonus or a Restricted Stock Purchase Right.
  - (bb) "Restricted Stock Bonus" means Stock granted to a Participant pursuant to Section 7.
  - (cc) "Restricted Stock Purchase Right" means a right to purchase Stock granted to a Participant pursuant to Section 7.
  - (dd) "Rule 16b-3" means Rule 16b-3 under the Exchange Act, as amended from time to time, or any successor rule or regulation.
  - (ee) "Securities Act" means the Securities Act of 1933, as amended.
- (ff) "Service" means a Participant's employment or service with the Participating Company Group, whether in the capacity of an Employee, a Director or a Consultant. Unless otherwise provided by the Board, a Participant's Service shall not be deemed to have terminated merely because of a change in the capacity in which the Participant renders such Service, provided that there is no interruption or termination of the Participant's Service. Furthermore, a Participant's Service shall not be deemed to have terminated if the Participant takes any military leave, sick leave, or other bona fide leave of absence approved by the Company. However, unless otherwise provided by the Board, if any such leave taken by a Participant exceeds ninety (90) days, then on the ninety-first (91st) day following the commencement of such leave the Participant's Service shall be deemed to have terminated, unless the Participant's right to return to Service is guaranteed by statute or contract.

Notwithstanding the foregoing, unless otherwise designated by the Company or required by law, an unpaid leave of absence shall not be treated as Service for purposes of determining vesting under the Participant's Award Agreement. Except as otherwise provided by the Board, in its discretion, the Participant's Service shall be deemed to have terminated either upon an actual termination of Service or upon the business entity for which the Participant performs Service ceasing to be a Participating Company. Subject to the foregoing, the Company, in its discretion, shall determine whether the Participant's Service has terminated and the effective date of and reason for such termination.

- (gg) "Stock" means the common stock of the Company, as adjusted from time to time in accordance with Section 4.2.
- (hh) "Subsidiary Corporation" means any present or future "subsidiary corporation" of the Company, as defined in Section 424(f) of the Code.
- (ii) "Ten Percent Stockholder" means a person who, at the time an Award is granted to such person, owns stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of a Participating Company within the meaning of Section 422(b)(6) of the Code.
- (jj) "Vesting Conditions" mean those conditions established in accordance with the Plan prior to the satisfaction of which shares subject to an Award remain subject to forfeiture or a repurchase option in favor of the Company exercisable for the Participant's monetary purchase price, if any, for such shares upon the Participant's termination of Service.
- 2.2 **Construction.** Captions and titles contained herein are for convenience only and shall not affect the meaning or interpretation of any provision of the Plan. Except when otherwise indicated by the context, the singular shall include the plural and the plural shall include the singular. Use of the term "or" is not intended to be exclusive, unless the context clearly requires otherwise.

#### 3. **ADMINISTRATION.**

- 3.1 Administration by the Board. The Plan shall be administered by the Board. All questions of interpretation of the Plan, of any Award Agreement or of any other form of agreement or other document employed by the Company in the administration of the Plan or of any Award shall be determined by the Board, and such determinations shall be final, binding and conclusive upon all persons having an interest in the Plan or such Award, unless fraudulent or made in bad faith. Any and all actions, decisions and determinations taken or made by the Board in the exercise of its discretion pursuant to the Plan or Award Agreement or other agreement thereunder (other than determining questions of interpretation pursuant to the preceding sentence) shall be final, binding and conclusive upon all persons having an interest therein.
- 3.2 Authority of Officers. Any Officer shall have the authority to act on behalf of the Company with respect to any matter, right, obligation, determination or election

which is the responsibility of or which is allocated to the Company herein, provided the Officer has apparent authority with respect to such matter, right, obligation, determination or election.

- 3.3 **Powers of the Board.** In addition to any other powers set forth in the Plan and subject to the provisions of the Plan, the Board shall have the full and final power and authority, in its discretion:
- (a) to determine the persons to whom, and the time or times at which, Awards shall be granted and the number of shares of Stock to be subject to each Award;
  - (b) to determine the type of Award granted;
  - (c) to determine the Fair Market Value of shares of Stock or other property;
- (d) to determine the terms, conditions and restrictions applicable to each Award (which need not be identical) and any shares acquired pursuant thereto, including, without limitation, (i) the exercise or purchase price of shares pursuant to any Award, (ii) the method of payment for shares purchased pursuant to any Award, (iii) the method for satisfaction of any tax withholding obligation arising in connection with any Award, including by the withholding or delivery of shares of Stock, (iv) the timing, terms and conditions of the exercisability or vesting of any Award or shares acquired pursuant thereto, (v) the time of expiration of any Award, (vii) the effect of any Participant's termination of Service on any of the foregoing, and (vii) all other terms, conditions and restrictions applicable to any Award or shares acquired pursuant thereto not inconsistent with the terms of the Plan;
  - (e) to approve one or more forms of Award Agreement;
- (f) to amend, modify, extend, cancel or renew any Award or to waive any restrictions or conditions applicable to any Award or any shares acquired pursuant thereto;
- (g) to accelerate, continue, extend or defer the exercisability or vesting of any Award or any shares acquired pursuant thereto, including with respect to the period following a Participant's termination of Service;
- (h) to prescribe, amend or rescind rules, guidelines and policies relating to the Plan, or to adopt sub-plans or supplements to, or alternative versions of, the Plan, including, without limitation, as the Board deems necessary or desirable to comply with the laws of, or to accommodate the tax policy, accounting principles or custom of, foreign jurisdictions whose citizens may be granted Awards; and
- (i) to correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award Agreement and to make all other determinations and take such other actions with respect to the Plan or any Award as the Board may deem advisable to the extent not inconsistent with the provisions of the Plan or applicable law.
- 3.4 **Administration with Respect to Insiders.** With respect to participation by Insiders in the Plan, at any time that any class of equity security of the Company is registered

pursuant to Section 12 of the Exchange Act, the Plan shall be administered in compliance with the requirements, if any, of Rule 16b-3.

3.5 **Indemnification.** In addition to such other rights of indemnification as they may have as members of the Board or as officers or employees of the Participating Company Group, members of the Board and any officers or employees of the Participating Company Group to whom authority to act for the Board or the Company is delegated shall be indemnified by the Company against all reasonable expenses, including attorneys' fees, actually and necessarily incurred in connection with the defense of any action, suit or proceeding, or in connection with any appeal therein, to which they or any of them may be a party by reason of any action taken or failure to act under or in connection with the Plan, or any right granted hereunder, and against all amounts paid by them in settlement thereof (provided such settlement is approved by independent legal counsel selected by the Company) or paid by them in satisfaction of a judgment in any such action, suit or proceeding, except in relation to matters as to which it shall be adjudged in such action, suit or proceeding that such person is liable for gross negligence, bad faith or intentional misconduct in duties; provided, however, that within sixty (60) days after the institution of such action, suit or proceeding, such person shall offer to the Company, in writing, the opportunity at its own expense to handle and defend the same.

#### 4. SHARES SUBJECT TO PLAN.

- 4.1 **Maximum Number of Shares Issuable.** Subject to adjustment as provided in Section 4.2, the maximum aggregate number of shares of Stock that may be issued under the Plan shall be 6,488,939 and shall consist of authorized but unissued or reacquired shares of Stock or any combination thereof. If an outstanding Award for any reason expires or is terminated or canceled or if shares of Stock are acquired pursuant to an Award subject to forfeiture or repurchase and are forfeited or repurchased by the Company for an amount not greater than the Participant's exercise or purchase price, the shares of Stock allocable to the terminated portion of such Award or such forfeited or repurchased shares of Stock shall again be available for issuance under the Plan. Notwithstanding the foregoing, at any such time as the offer and sale of securities pursuant to the Plan is subject to compliance with Section 260.140.45 of Title 10 of the California Code of Regulations ("Section 260.140.45"), the total number of shares of Stock issuable upon the exercise of all outstanding Awards (together with options outstanding under any other stock plan of the Company) and the total number of shares provided for under any stock bonus or similar plan of the Company shall not exceed thirty percent (30%) (or such other higher percentage limitation as may be approved by the stockholders of the Company pursuant to Section 260.140.45) of the then outstanding shares of the Company as calculated in accordance with the conditions and exclusions of Section 260.140.45.
- 4.2 Adjustments for Changes in Capital Structure. Subject to any required action by the stockholders of the Company and the requirements of Sections 409A and 424 of the Code to the extent applicable, in the event of any change in the Stock effected without receipt of consideration by the Company, whether through merger, consolidation, reorganization, reincorporation, recapitalization, reclassification, stock dividend, stock split, reverse stock split, split-up, split-off, spin-off, combination of shares, exchange of shares, or similar change in the capital structure of the Company, or in the event of payment of a dividend or distribution to the stockholders of the Company in a form other than Stock (excepting normal cash dividends) that

has a material effect on the Fair Market Value of shares of Stock, appropriate and proportionate adjustments shall be made in the number and kind of shares subject to the Plan and to any outstanding Awards, in the ISO Share Limit set forth in Section 5.3(a), and in the exercise or purchase price per share of any outstanding Awards in order to prevent dilution or enlargement of Participants' rights under the Plan. For purposes of the foregoing, conversion of any convertible securities of the Company shall not be treated as "effected without receipt of consideration by the Company." If a majority of the shares which are of the same class as the shares that are subject to outstanding Awards are exchanged for, converted into, or otherwise become (whether or not pursuant to an Ownership Change Event) shares of another corporation (the "New Shares"), the Board may unilaterally amend the outstanding Awards to provide that such Awards are for New Shares. In the event of any such amendment, the number of shares subject to, and the exercise or purchase price per share of, the outstanding Awards shall be adjusted in a fair and equitable manner as determined by the Board, in its discretion. Any fractional share resulting from an adjustment pursuant to this Section shall be rounded down to the nearest whole number, and the exercise price per share shall be rounded up to the nearest whole cent. In no event may the exercise or purchase price, if any, under any Award be decreased to an amount less than the par value, if any, of the stock subject to the Award. Such adjustments shall be determined by the Board, and its determination shall be final, binding and conclusive.

#### 5. ELIGIBILITY AND OPTION LIMITATIONS.

- 5.1 **Persons Eligible for Awards.** Awards may be granted only to Employees, Consultants and Directors.
- 5.2 **Participation in the Plan.** Awards are granted solely at the discretion of the Board. Eligible persons may be granted more than one Award. However, eligibility in accordance with this Section shall not entitle any person to be granted an Award, or, having been granted an Award, to be granted an additional Award.
  - 5.3 Incentive Stock Option Limitations.
- (a) **Maximum Number of Shares Issuable Pursuant to Incentive Stock Options.** Subject to Section 4.1 and adjustment as provided in Section 4.2, the maximum aggregate number of shares of Stock that may be issued under the Plan pursuant to the exercise of Incentive Stock Options shall not exceed 6,488,939 shares (the "ISO Share Limit"). The maximum aggregate number of shares of Stock that may be issued under the Plan pursuant to all Awards other than Incentive Stock Options shall be the number of shares determined in accordance with Section 4.1, subject to adjustment as provided in Section 4.2.
- (b) **Persons Eligible.** An Incentive Stock Option may be granted only to a person who, on the effective date of grant, is an Employee. Any person who is not an Employee on the effective date of the grant of an Option to such person may be granted only a Nonstatutory Stock Option.
- (c) Fair Market Value Limitation. To the extent that options designated as Incentive Stock Options (granted under all stock plans of the Participating

Company Group, including the Plan) become exercisable by a Participant for the first time during any calendar year for stock having a Fair Market Value greater than One Hundred Thousand Dollars (\$100,000), the portions of such options which exceed such amount shall be treated as Nonstatutory Stock Options. For purposes of this Section 5.3, options designated as Incentive Stock Options shall be taken into account in the order in which they were granted, and the Fair Market Value of stock shall be determined as of the time the option with respect to such stock is granted. If the Code is amended to provide for a limitation different from that set forth in this Section, such different limitation shall be deemed incorporated herein effective as of the date and with respect to such Options as required or permitted by such amendment to the Code. If an Option is treated as an Incentive Stock Option in part and as a Nonstatutory Stock Option in part by reason of the limitation set forth in this Section, the Participant may designate which portion of such Option the Participant is exercising. In the absence of such designation, the Participant shall be deemed to have exercised the Incentive Stock Option portion of the Option first. Upon exercise of the Option, Shares issued pursuant to each such portion shall be separately identified.

#### 6. STOCK OPTIONS.

Options shall be evidenced by Award Agreements specifying the number of shares of Stock covered thereby, in such form as the Board shall from time to time establish. Such Award Agreements may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

- 6.1 Exercise Price. The exercise price for each Option shall be established in the discretion of the Board; provided, however, that (a) the exercise price per share for an Option shall be not less than the Fair Market Value of a share of Stock on the effective date of grant of the Option and (b) no Incentive Stock Option granted to a Ten Percent Stockholder shall have an exercise price per share less than one hundred ten percent (110%) of the Fair Market Value of a share of Stock on the effective date of grant of the Option. Notwithstanding the foregoing, an Option (whether an Incentive Stock Option or a Nonstatutory Stock Option) may be granted with an exercise price lower than the minimum exercise price set forth above if such Option is granted pursuant to an assumption or substitution for another option in a manner qualifying under the provisions of Section 424(a) of the Code.
- 6.2 Exercisability and Term of Options. Options shall be exercisable at such time or times, or upon such event or events, and subject to such terms, conditions, performance criteria and restrictions as shall be determined by the Board and set forth in the Award Agreement evidencing such Option; provided, however, that (a) no Option shall be exercisable after the expiration of ten (10) years after the effective date of grant of such Option and (b) no Incentive Stock Option granted to a Ten Percent Stockholder shall be exercisable after the expiration of five (5) years after the effective date of grant of such Option. Subject to the foregoing, unless otherwise specified by the Board in the grant of an Option, any Option granted hereunder shall terminate ten (10) years after the effective date of grant of the Option, unless earlier terminated in accordance with its provisions.

#### 6.3 Payment of Exercise Price.

(a) Forms of Consideration Authorized. Except as otherwise provided below, payment of the exercise price for the number of shares of Stock being purchased pursuant to any Option shall be made (i) in cash, by check or in cash equivalent, (ii) by tender to the Company, or attestation to the ownership, of shares of Stock owned by the Participant having a Fair Market Value not less than the exercise price, (iii) by delivery of a properly executed notice of exercise together with irrevocable instructions to a broker providing for the assignment to the Company of the proceeds of a sale or loan with respect to some or all of the shares being acquired upon the exercise of the Option (including, without limitation, through an exercise complying with the provisions of Regulation T as promulgated from time to time by the Board of Governors of the Federal Reserve System) (a "Cashless Exercise"), (iv) by delivery of a properly executed notice electing a Net-Exercise, (v) by such other consideration as may be approved by the Board from time to time to the extent permitted by applicable law, or (vi) by any combination thereof. The Board may at any time or from time to time grant Options which do not permit all of the foregoing forms of consideration to be used in payment of the exercise price or which otherwise restrict one or more forms of consideration.

#### (b) Limitations on Forms of Consideration.

- (i) **Tender of Stock.** Notwithstanding the foregoing, an Option may not be exercised by tender to the Company, or attestation to the ownership, of shares of Stock to the extent such tender or attestation would constitute a violation of the provisions of any law, regulation or agreement restricting the redemption of the Company's stock.
- (ii) Cashless Exercise. The Company reserves, at any and all times, the right, in the Company's sole and absolute discretion, to establish, decline to approve or terminate any program or procedures for the exercise of Options by means of a Cashless Exercise, including with respect to one or more Participants specified by the Company notwithstanding that such program or procedures may be available to other Participants.

#### 6.4 Effect of Termination of Service.

- (a) *Option Exercisability*. Subject to earlier termination of the Option as otherwise provided by this Plan and unless a longer exercise period is provided by the Board, an Option shall terminate immediately upon the Participant's termination of Service to the extent that it is then unvested and shall be exercisable after the Participant's termination of Service to the extent it is then vested only during the applicable time period determined in accordance with this Section and thereafter shall terminate:
- (i) **Disability.** If the Participant's Service terminates because of the Disability of the Participant, the Option, to the extent unexercised and exercisable for vested shares on the date on which the Participant's Service terminated, may be exercised by the Participant (or the Participant's guardian or legal representative) at any time prior to the expiration of twelve (12) months after the date on which the Participant's Service terminated, but in any event no later than the date of expiration of the Option's term as set forth in the Award Agreement evidencing such Option (the "Option Expiration Date").

(ii) **Death.** If the Participant's Service terminates because of the death of the Participant, the Option, to the extent unexercised and exercisable for vested shares on the date on which the Participant's Service terminated, may be exercised by the Participant's legal representative or other person who acquired the right to exercise the Option by reason of the Participant's death at any time prior to the expiration of twelve (12) months after the date on which the Participant's Service terminated, but in any event no later than the Option Expiration Date. The Participant's Service shall be deemed to have terminated on account of death if the Participant dies within three (3) months after the Participant's termination of Service.

(iii) **Termination for Cause.** Notwithstanding any other provision of the Plan to the contrary, if the Participant's Service is terminated for Cause, the Option shall terminate in its entirety and cease to be exercisable immediately upon such termination of Service.

(iv) **Other Termination of Service.** If the Participant's Service terminates for any reason, except Disability, death or Cause, the Option, to the extent unexercised and exercisable for vested shares on the date on which the Participant's Service terminated, may be exercised by the Participant at any time prior to the expiration of three (3) months after the date on which the Participant's Service terminated, but in any event no later than the Option Expiration Date.

(b) Extension if Exercise Prevented by Law. Notwithstanding the foregoing other than termination of Service for Cause, if the exercise of an Option within the applicable time periods set forth in Section 6.4(a) is prevented by the provisions of Section 11 below, the Option shall remain exercisable until the later of (i) thirty (30) days after the date such exercise first would no longer be prevented by such provisions or (ii) the end of the applicable time period under Section 6.4(a), but in any event no later than the Option Expiration Date.

6.5 **Transferability of Options.** During the lifetime of the Participant, an Option shall be exercisable only by the Participant or the Participant's guardian or legal representative. An Option shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant's beneficiary, except transfer by will or by the laws of descent and distribution. Notwithstanding the foregoing, to the extent permitted by the Board, in its discretion, and set forth in the Award Agreement evidencing such Option, an Option shall be assignable or transferable subject to the applicable limitations, if any, described in Rule 701 under the Securities Act and the General Instructions to Form S-8 Registration Statement under the Securities Act or, in the case of an Incentive Stock Option, only as permitted by applicable regulations under Section 421 of the Code in a manner that does not disqualify such Option as an Incentive Stock Option.

#### 7. RESTRICTED STOCK AWARDS.

Restricted Stock Awards shall be evidenced by Award Agreements specifying whether the Award is a Restricted Stock Bonus or a Restricted Stock Purchase Right and the number of shares of Stock subject to the Award, in such form as the Board shall from time to

time establish. Such Award Agreements may incorporate all or any of the terms of the Plan by reference and shall comply with and be subject to the following terms and conditions:

- 7.1 **Types of Restricted Stock Awards Authorized.** Restricted Stock Awards may be granted in the form of either a Restricted Stock Bonus or a Restricted Stock Purchase Right. Restricted Stock Awards may be granted upon such conditions as the Board shall determine, including, without limitation, upon the attainment of one or more performance goals.
- 7.2 **Purchase Price.** The purchase price for shares of Stock issuable under each Restricted Stock Purchase Right shall be established by the Board in its discretion. No monetary payment (other than applicable tax withholding) shall be required as a condition of receiving shares of Stock pursuant to a Restricted Stock Bonus, the consideration for which shall be services actually rendered to a Participating Company or for its benefit. Notwithstanding the foregoing, if required by applicable state corporate law, the Participant shall furnish consideration in the form of cash or past services rendered to a Participating Company or for its benefit having a value not less than the par value of the shares of Stock subject to a Restricted Stock Award.
- 7.3 **Purchase Period.** A Restricted Stock Purchase Right shall be exercisable within a period established by the Board, which shall in no event exceed thirty (30) days from the effective date of the grant of the Restricted Stock Purchase Right.
- 7.4 **Payment of Purchase Price.** Except as otherwise provided below, payment of the purchase price for the number of shares of Stock being purchased pursuant to any Restricted Stock Purchase Right shall be made (a) in cash, by check or in cash equivalent, (b) by such other consideration as may be approved by the Board from time to time to the extent permitted by applicable law, or (c) by any combination thereof.
- 7.5 Vesting and Restrictions on Transfer. Shares issued pursuant to any Restricted Stock Award may (but need not) be made subject to Vesting Conditions based upon the satisfaction of such Service requirements, conditions, restrictions or performance criteria, as shall be established by the Board and set forth in the Award Agreement evidencing such Award. During any period in which shares acquired pursuant to a Restricted Stock Award remain subject to Vesting Conditions, such shares may not be sold, exchanged, transferred, pledged, assigned or otherwise disposed of other than pursuant to an Ownership Change Event or as provided in Section 7.8. The Board, in its discretion, may provide in any Award Agreement evidencing a Restricted Stock Award that, if the satisfaction of Vesting Conditions with respect to any shares subject to such Restricted Stock Award would otherwise occur on a day on which the sale of such shares would violate the provisions of the Insider Trading Policy, then satisfaction of the Vesting Conditions automatically shall be determined on the next trading day on which the sale of such shares would not violate the Insider Trading Policy. Upon request by the Company, each Participant shall execute any agreement evidencing such transfer restrictions prior to the receipt of shares of Stock hereunder and shall promptly present to the Company and all certificates representing shares of Stock acquired hereunder for the placement on such certificates of appropriate legends evidencing any such transfer restrictions.

- 7.6 **Voting Rights; Dividends and Distributions.** Except as provided in this Section, Section 7.5 and any Award Agreement, during any period in which shares acquired pursuant to a Restricted Stock Award remain subject to Vesting Conditions, the Participant shall have all of the rights of a stockholder of the Company holding shares of Stock, including the right to vote such shares and to receive all dividends and other distributions paid with respect to such shares; provided, however, that if so determined by the Board and provided by the Award Agreement, such dividends and distributions shall be subject to the same Vesting Conditions as the shares subject to the Restricted Stock Award with respect to which such dividends or distributions were paid. In the event of a dividend or distribution paid in shares of Stock or other property or any other adjustment made upon a change in the capital structure of the Company as described in Section 4.2, any and all new, substituted or additional securities or other property (other than normal cash dividends) to which the Participant is entitled by reason of the Participant's Restricted Stock Award shall be immediately subject to the same Vesting Conditions as the shares subject to the Restricted Stock Award with respect to which such dividends or distributions were paid or adjustments were made.
- 7.7 Effect of Termination of Service. Unless otherwise provided by the Board in the Award Agreement evidencing a Restricted Stock Award, if a Participant's Service terminates for any reason, whether voluntary or involuntary (including the Participant's death or disability), then (a) the Company shall have the option to repurchase for the purchase price paid by the Participant any shares acquired by the Participant pursuant to a Restricted Stock Purchase Right which remain subject to Vesting Conditions as of the date of the Participant's termination of Service and (b) the Participant shall forfeit to the Company any shares acquired by the Participant pursuant to a Restricted Stock Bonus which remain subject to Vesting Conditions as of the date of the Participant's termination of Service. The Company shall have the right to assign at any time any repurchase right it may have, whether or not such right is then exercisable, to one or more persons as may be selected by the Company.
- 7.8 Nontransferability of Restricted Stock Award Rights. Rights to acquire shares of Stock pursuant to a Restricted Stock Award shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance or garnishment by creditors of the Participant or the Participant's beneficiary, except transfer by will or the laws of descent and distribution. All rights with respect to a Restricted Stock Award granted to a Participant hereunder shall be exercisable during his or her lifetime only by such Participant or the Participant's guardian or legal representative.

#### 8. STANDARD FORMS OF AWARD AGREEMENTS.

8.1 **Award Agreements.** Each Award shall comply with and be subject to the terms and conditions set forth in the appropriate form of Award Agreement approved by the Board and as amended from time to time. No Award or purported Award shall be a valid and binding obligation of the Company unless evidenced by a fully executed Award Agreement. Any Award Agreement may consist of an appropriate form of Notice of Grant and a form of Agreement incorporated therein by reference, or such other form or forms, including electronic media, as the Board may approve from time to time.

8.2 **Authority to Vary Terms.** The Board shall have the authority from time to time to vary the terms of any standard form of Award Agreement either in connection with the grant or amendment of an individual Award or in connection with the authorization of a new standard form or forms; provided, however, that the terms and conditions of any such new, revised or amended standard form or forms of Award Agreement are not inconsistent with the terms of the Plan.

#### 9. CHANGE IN CONTROL.

- 9.1 **Effect of Change in Control on Awards.** Subject to the requirements and limitations of Section 409A of the Code, if applicable, the Board may provide for any one or more of the following:
- (a) Accelerated Vesting. The Board may, in its discretion, provide in any Award Agreement or, in the event of a Change in Control, may take such actions as it deems appropriate to provide for the acceleration of the exercisability and/or vesting in connection with such Change in Control of each or any outstanding Award or portion thereof and shares acquired pursuant thereto upon such conditions, including termination of the Participant's Service prior to, upon, or following such Change in Control, to such extent as the Board shall determine.
- (b) Assumption, Continuation or Substitution of Awards. In the event of a Change in Control, the surviving, continuing, successor, or purchasing corporation or other business entity or parent thereof, as the case may be (the "Acquiror"), may, without the consent of any Participant, assume or continue the Company's rights and obligations under each or any Award or portion thereof outstanding immediately prior to the Change in Control or substitute for each or any such outstanding Award or portion thereof a substantially equivalent award with respect to the Acquiror's stock. For purposes of this Section, if so determined by the Board, in its discretion, an Award or any portion thereof shall be deemed assumed if, following the Change in Control, the Award confers the right to receive, subject to the terms and conditions of the Plan and the applicable Award Agreement, for each share of Stock subject to such portion of the Award immediately prior to the Change in Control, the consideration (whether stock, cash, other securities or property or a combination thereof) to which a holder of a share of Stock on the effective date of the Change in Control was entitled; provided, however, that if such consideration is not solely common stock of the Acquiror, the Board may, with the consent of the Acquiror, provide for the consideration to be received upon the exercise of the Award for each share of Stock to consist solely of common stock of the Acquiror equal in Fair Market Value to the per share consideration received by holders of Stock pursuant to the Change in Control. If any portion of such consideration may be received by holders of Stock pursuant to the Change in Control on a contingent or delayed basis, the Board may, in its discretion, determine such Fair Market Value per share as of the time of the Change in Control on the basis of the Board's good faith estimate of the present value of the probable future payment of such consideration. Any Award or portion thereof which is neither assumed or continued by the Acquiror in connection with the Change in Control nor exercised as of the time of consummation of the Change in Control shall terminate and cease to be outstanding effective as of the time of consummation of the Change in Control. Notwithstanding the foregoing, shares acquired upon exercise of an Award prior to the Change in Control and any consideration received pursuant to the Change in Control with respect to such shares shall continue to be subject to all applicable provisions of the

Award Agreement evidencing such Award except as otherwise provided in such Award Agreement.

(c) Cash-Out of Outstanding Awards. The Board may, in its discretion and without the consent of any Participant, determine that, upon the occurrence of a Change in Control, each or any Award or portion thereof outstanding immediately prior to the Change in Control shall be canceled in exchange for a payment with respect to each vested share (and each unvested share, if so determined by the Board) of Stock subject to such canceled Award in (i) cash, (ii) stock of the Company or of a corporation or other business entity a party to the Change in Control, or (iii) other property which, in any such case, shall be in an amount having a Fair Market Value equal to the Fair Market Value of the consideration to be paid per share of Stock in the Change in Control, reduced by the exercise or purchase price per share, if any, under such Award. If any portion of such consideration may be received by holders of Stock pursuant to the Change in Control on a contingent or delayed basis, the Board may, in its sole discretion, determine such Fair Market Value per share as of the time of the Change in Control on the basis of the Board's good faith estimate of the present value of the probable amount of future payment of such consideration. In the event such determination is made by the Board, the amount of such payment (reduced by applicable withholding taxes, if any) shall be paid to Participants in respect of the vested portions of their canceled Awards as soon as practicable following the date of the Change in Control and in respect of the unvested portions of their canceled Awards in accordance with the vesting schedules applicable to such Awards.

#### 9.2 Federal Excise Tax Under Section 4999 of the Code.

(a) *Excess Parachute Payment*. If any acceleration of vesting pursuant to an Award and any other payment or benefit received or to be received by a Participant would subject the Participant to any excise tax pursuant to Section 4999 of the Code due to the characterization of such acceleration of vesting, payment or benefit as an "excess parachute payment" under Section 280G of the Code, then, provided such election would not subject the Participant to taxation under Section 409A of the Code, the Participant may elect, in his or her sole discretion, to reduce the amount of any acceleration of vesting called for under the Award in order to avoid such characterization.

(b) **Determination by Independent Accountants.** To aid the Participant in making any election called for under Section 9.2(a), no later than the date of the occurrence of any event that might reasonably be anticipated to result in an "excess parachute payment" to the Participant as described in Section 9.2(a), the Company shall request a determination in writing by independent public accountants selected by the Company (the "Accountants"). As soon as practicable thereafter, the Accountants shall determine and report to the Company and the Participant the amount of such acceleration of vesting, payments and benefits which would produce the greatest after-tax benefit to the Participant. For the purposes of such determination, the Accountants may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Company and the Participant shall furnish to the Accountants such information and documents as the Accountants may reasonably request in order to make their required determination. The Company shall bear all fees and expenses the Accountants may reasonably charge in connection with their services contemplated by this Section 9.2(b).

#### 10. TAX WITHHOLDING.

10.1 **Tax Withholding in General.** The Company shall have the right to deduct from any and all payments made under the Plan, or to require the Participant, through payroll withholding, cash payment or otherwise, including by means of a Cashless Exercise of an Option, to make adequate provision for, the federal, state, local and foreign taxes (including any social insurance tax), if any, required by law to be withheld by the Participating Company Group with respect to an Award or the shares acquired pursuant thereto. The Company shall have no obligation to deliver shares of Stock or to release shares of Stock from an escrow established pursuant to an Award Agreement until the Participating Company Group's tax withholding obligations have been satisfied by the Participant.

10.2 **Withholding in Shares.** The Company shall have the right, but not the obligation, to deduct from the shares of Stock issuable to a Participant upon the exercise or vesting of an Award, or to accept from the Participant the tender of, a number of whole shares of Stock having a Fair Market Value, as determined by the Company, equal to all or any part of the tax withholding obligations of the Participating Company Group. The Fair Market Value of any shares of Stock withheld or tendered to satisfy any such tax withholding obligations shall not exceed the amount determined by the applicable minimum statutory withholding rates.

#### 11. COMPLIANCE WITH SECURITIES LAW.

The grant of Awards and the issuance of shares of Stock pursuant to any Award shall be subject to compliance with all applicable requirements of federal, state and foreign law with respect to such securities and the requirements of any stock exchange or market system upon which the Stock may then be listed. In addition, no Award may be exercised or shares issued pursuant to an Award unless (a) a registration statement under the Securities Act shall at the time of such exercise or issuance be in effect with respect to the shares issuable pursuant to the Award or (b) in the opinion of legal counsel to the Company, the shares issuable pursuant to the Award may be issued in accordance with the terms of an applicable exemption from the registration requirements of the Securities Act. The inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance and sale of any shares hereunder shall relieve the Company of any liability in respect of the failure to issue or sell such shares as to which such requisite authority shall not have been obtained. As a condition to issuance of any Stock, the Company may require the Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company.

#### 12. AMENDMENT OR TERMINATION OF PLAN.

The Board may amend, suspend or terminate the Plan at any time. However, without the approval of the Company's stockholders, there shall be (a) no increase in the maximum aggregate number of shares of Stock that may be issued under the Plan (except by operation of the provisions of Section 4.2), (b) no change in the class of persons eligible to receive Incentive Stock Options, and (c) no other amendment of the Plan that would require approval of the Company's stockholders under any applicable law, regulation or rule, including

the rules of any stock exchange or market system upon which the Stock may then be listed. No amendment, suspension or termination of the Plan shall affect any then outstanding Award unless expressly provided by the Board. Except as provided by the next sentence, no amendment, suspension or termination of the Plan may adversely affect any then outstanding Award without the consent of the Participant. Notwithstanding any other provision of the Plan or any Award Agreement to the contrary, the Board may, in its sole and absolute discretion and without the consent of any Participant, amend the Plan or any Award Agreement, to take effect retroactively or otherwise, as it deems necessary or advisable for the purpose of conforming the Plan or such Award Agreement to any present or future law, regulation or rule applicable to the Plan, including, but not limited to, Section 409A of the Code.

#### 13. MISCELLANEOUS PROVISIONS.

- 13.1 **Repurchase Rights.** Shares issued under the Plan may be subject to a right of first refusal, one or more repurchase options, or other conditions and restrictions as determined by the Board in its discretion at the time the Award is granted. The Company shall have the right to assign at any time any repurchase right it may have, whether or not such right is then exercisable, to one or more persons as may be selected by the Company. Upon request by the Company, each Participant shall execute any agreement evidencing such transfer restrictions prior to the receipt of shares of Stock hereunder and shall promptly present to the Company any and all certificates representing shares of Stock acquired hereunder for the placement on such certificates of appropriate legends evidencing any such transfer restrictions.
- 13.2 **Provision of Information**. At least annually, copies of the Company's balance sheet and income statement for the just completed fiscal year shall be made available to each Participant and purchaser of shares of Stock upon the exercise of an Award; provided, however, that this requirement shall not apply if all offers and sales of securities pursuant to the Plan comply with all applicable conditions of Rule 701 under the Securities Act. The Company shall not be required to provide such information to key persons whose duties in connection with the Company assure them access to equivalent information. The Company shall deliver to each Participant such disclosures as are required in accordance with Rule 701 under the Securities Act.
- 13.3 **Rights as Employee, Consultant or Director.** No person, even though eligible pursuant to Section 5, shall have a right to be selected as a Participant, or, having been so selected, to be selected again as a Participant. Nothing in the Plan or any Award granted under the Plan shall confer on any Participant a right to remain an Employee, Consultant or Director or interfere with or limit in any way any right of a Participating Company to terminate the Participant's Service at any time. To the extent that an Employee of a Participating Company other than the Company receives an Award under the Plan, that Award shall in no event be understood or interpreted to mean that the Company is the Employee's employer or that the Employee has an employment relationship with the Company.
- 13.4 **Rights as a Stockholder.** A Participant shall have no rights as a stockholder with respect to any shares covered by an Award until the date of the issuance of such shares (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for dividends,

distributions or other rights for which the record date is prior to the date such shares are issued, except as provided in Section 4.2 or another provision of the Plan.

- 13.5 **Delivery of Title to Shares.** Subject to any governing rules or regulations, the Company shall issue or cause to be issued the shares of Stock acquired pursuant to an Award and shall deliver such shares to or for the benefit of the Participant by means of one or more of the following: (a) by delivering to the Participant evidence of book entry shares of Stock credited to the account of the Participant, (b) by depositing such shares of Stock for the benefit of the Participant with any broker with which the Participant has an account relationship, or (c) by delivering such shares of Stock to the Participant in certificate form.
  - 13.6 Fractional Shares. The Company shall not be required to issue fractional shares upon the exercise or settlement of any Award.
- 13.7 **Retirement and Welfare Plans**. Neither Awards made under this Plan nor shares of Stock or cash paid pursuant to such Awards shall be included as "compensation" for purposes of computing the benefits payable to any Participant under any Participating Company's retirement plans (both qualified and non-qualified) or welfare benefit plans unless such other plan expressly provides that such compensation shall be taken into account in computing such benefits.
- 13.8 **Severability**. If any one or more of the provisions (or any part thereof) of this Plan shall be held invalid, illegal or unenforceable in any respect, such provision shall be modified so as to make it valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions (or any part thereof) of the Plan shall not in any way be affected or impaired thereby.
- 13.9 **No Constraint on Corporate Action.** Nothing in this Plan shall be construed to: (a) limit, impair, or otherwise affect the Company's or another Participating Company's right or power to make adjustments, reclassifications, reorganizations, or changes of its capital or business structure, or to merge or consolidate, or dissolve, liquidate, sell, or transfer all or any part of its business or assets; or (b) limit the right or power of the Company or another Participating Company to take any action which such entity deems to be necessary or appropriate.
- 13.10 **Choice of Law.** Except to the extent governed by applicable federal law, the validity, interpretation, construction and performance of the Plan and each Award Agreement shall be governed by the laws of the State of California, without regard to its conflict of law rules.
- 13.11 **Stockholder Approval**. The Plan or any increase in the maximum aggregate number of shares of Stock issuable thereunder as provided in Section 4.1 (the "*Authorized Shares*") shall be approved by a majority of the outstanding securities of the Company entitled to vote by the later of (a) a period beginning twelve (12) months before and ending twelve (12) months after the date of adoption thereof by the Board or (b) the first issuance of any security pursuant to the Plan in the State of California (within the meaning of Section 25008 of the California Corporations Code). Awards granted prior to security holder

approval of the Plan or in excess of the Authorized Shares previously approved by the security holders shall become exercisable no earlier than the date of security holder approval of the Plan or such increase in the Authorized Shares, as the case may be, and such Awards shall be rescinded if such security holder approval is not received in the manner described in the preceding sentence.

### PLAN HISTORY

August 4, 2009 Board adopts Plan, with an initial reserve of 6,488,939 shares.

August 4, 2009 Stockholders of the Company approve Plan.

THE SECURITIES WHICH ARE THE SUBJECT OF THIS AGREEMENT HAVE NOT BEEN QUALIFIED WITH THE COMMISSIONER OF CORPORATIONS OF THE STATE OF CALIFORNIA AND THE ISSUANCE OF SUCH SECURITIES OR THE PAYMENT OR RECEIPT OF ANY PART OF THE CONSIDERATION THEREFOR PRIOR TO SUCH QUALIFICATION IS UNLAWFUL, UNLESS THE SALE OF SECURITIES IS EXEMPT FROM QUALIFICATION BY SECTION 25100, 25102, OR 25105 OF THE CALIFORNIA CORPORATIONS CODE. THE RIGHTS OF ALL PARTIES TO THIS AGREEMENT ARE EXPRESSLY CONDITIONED UPON SUCH QUALIFICATION BEING OBTAINED, UNLESS THE SALE IS SO EXEMPT.

THE SECURITIES WHICH ARE THE SUBJECT OF THIS AGREEMENT HAVE BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO, OR IN CONNECTION WITH, THE SALE OR DISTRIBUTION THEREOF. NO SUCH SALE OR DISPOSITION MAY BE EFFECTED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933.

#### ISOCKET, INC. STOCK OPTION AGREEMENT

isocket, inc. has granted to the Participant named in the *Notice of Grant of Stock Option* (the "*Grant Notice*") to which this Stock Option Agreement (the "*Option Agreement*") is attached an option (the "*Option*") to purchase certain shares of Stock upon the terms and conditions set forth in the Grant Notice and this Option Agreement. The Option has been granted pursuant to and shall in all respects be subject to the terms and conditions of the isocket, inc. 2009 Equity Incentive Plan (the "*Plan*"), as amended to the Date of Grant, the provisions of which are incorporated herein by reference. By signing the Grant Notice, the Participant: (a) acknowledges receipt of, and represents that the Participant has read and is familiar with, the Grant Notice, this Option Agreement and the Plan, (b) accepts the Option subject to all of the terms and conditions of the Grant Notice, this Option Agreement and the Plan, and (c) agrees to accept as binding, conclusive and final all decisions or interpretations of the Board upon any questions arising under the Grant Notice, this Option Agreement or the Plan.

#### 1. **DEFINITIONS AND CONSTRUCTION.**

1.1 **Definitions.** Unless otherwise defined herein, capitalized terms shall have the meanings assigned to such terms in the Grant Notice or the Plan.

1.2 **Construction.** Captions and titles contained herein are for convenience only and shall not affect the meaning or interpretation of any provision of this Option Agreement. Except when otherwise indicated by the context, the singular shall include the plural and the plural shall include the singular. Use of the term "or" is not intended to be exclusive, unless the context clearly requires otherwise.

#### 2. TAX CONSEQUENCES.

- 2.1 Tax Status of Option. This Option is intended to have the tax status designated in the Grant Notice.
- (a) *Incentive Stock Option*. If the Grant Notice so designates, this Option is intended to be an Incentive Stock Option within the meaning of Section 422(b) of the Code, but the Company does not represent or warrant that this Option qualifies as such. The Participant should consult with the Participant's own tax advisor regarding the tax effects of this Option and the requirements necessary to obtain favorable income tax treatment under Section 422 of the Code, including, but not limited to, holding period requirements. (NOTE TO PARTICIPANT: If the Option is exercised more than three (3) months after the date on which you cease to be an Employee (other than by reason of your death or permanent and total disability as defined in Section 22(e)(3) of the Code), the Option will be treated as a Nonstatutory Stock Option and not as an Incentive Stock Option to the extent required by Section 422 of the Code.)
- (b) *Nonstatutory Stock Option*. If the Grant Notice so designates, this Option is intended to be a Nonstatutory Stock Option and shall not be treated as an Incentive Stock Option within the meaning of Section 422(b) of the Code.
- 2.2 ISO Fair Market Value Limitation. If the Grant Notice designates this Option as an Incentive Stock Option, then to the extent that the Option (together with all Incentive Stock Options granted to the Participant under all stock option plans of the Participating Company Group, including the Plan) becomes exercisable for the first time during any calendar year for shares having a Fair Market Value greater than One Hundred Thousand Dollars (\$100,000), the portion of such options which exceeds such amount will be treated as Nonstatutory Stock Options. For purposes of this Section, options designated as Incentive Stock Options are taken into account in the order in which they were granted, and the Fair Market Value of stock is determined as of the time the option with respect to such stock is granted. If the Code is amended to provide for a different limitation from that set forth in this Section, such different limitation shall be deemed incorporated herein effective as of the date required or permitted by such amendment to the Code. If the Option is treated as an Incentive Stock Option in part and as a Nonstatutory Stock Option in part by reason of the limitation set forth in this Section, the Participant may designate which portion of such Option the Participant is exercising. In the absence of such designation, the Participant shall be deemed to have exercised the Incentive Stock Option portion of the Option first. Separate certificates representing each such portion shall be issued upon the exercise of the Option. (NOTE TO PARTICIPANT: If the aggregate Exercise Price of the Option (that is, the Exercise Price multiplied by the Number of Option Shares) plus the aggregate exercise price of any other Incentive Stock Options you hold (whether granted pursuant to the Plan or any other stock option plus of the Participating Company Group) is greater than \$100,000, you should contact the Chief Financial Officer of the Company to ascertain whether the entire Option qualifies as an Incentive Stock Option.)

#### 3. **ADMINISTRATION.**

All questions of interpretation concerning the Grant Notice, this Option Agreement, the Plan or any other form of agreement or other document employed by the Company in the administration of the Plan or the Option shall be determined by the Board. All such determinations by the Board shall be final, binding and conclusive upon all persons having an interest in the Option, unless fraudulent or made in bad faith. Any and all actions, decisions and determinations taken or made by the Board in the exercise of its discretion pursuant to the Plan or the Option or other agreement thereunder (other than determining questions of interpretation pursuant to the preceding sentence) shall be final, binding and conclusive upon all persons having an interest in the Option. Any Officer shall have the authority to act on behalf of the Company with respect to any matter, right, obligation, or election which is allocated to the Company herein, provided the Officer has apparent authority with respect to such matter, right, obligation, or election.

#### 4. EXERCISE OF THE OPTION.

- 4.1 **Right to Exercise**. Except as otherwise provided herein, the Option shall be exercisable on and after the Initial Vesting Date and prior to the termination of the Option (as provided in Section 6) in an amount not to exceed the number of Vested Shares less the number of shares previously acquired upon exercise of the Option, subject to the Company's repurchase rights set forth in Section 11. In no event shall the Option be exercisable for more shares than the Number of Option Shares, as adjusted pursuant to Section 9.
- 4.2 Method of Exercise. Exercise of the Option shall be by means of electronic or written notice (the "Exercise Notice") in a form authorized by the Company. An electronic Exercise Notice must be digitally signed or authenticated by the Participant in such manner as required by the notice and transmitted to the Company or an authorized representative of the Company (including a third-party administrator designated by the Company). In the event that the Participant is not authorized or is unable to provide an electronic Exercise Notice, the Option shall be exercised by a written Exercise Notice addressed to the Company, which shall be signed by the Participant and delivered in person, by certified or registered mail, return receipt requested, by confirmed facsimile transmission, or by such other means as the Company may permit, to the Company, or an authorized representative of the Company (including a third-party administrator designated by the Company). Each Exercise Notice, whether electronic or written, must state the Participant's election to exercise the Option, the number of whole shares of Stock for which the Option is being exercised and such other representations and agreements as to the Participant's investment intent with respect to such shares as may be required pursuant to the provisions of this Option Agreement. Further, each Exercise Notice must be received by the Company prior to the termination of the Option as set forth in Section 6 and must be accompanied by full payment of the aggregate Exercise Price for the number of shares of Stock being purchased. The Option shall be deemed to be exercised upon receipt by the Company of such electronic or written Exercise Notice and the aggregate Exercise Price.

#### 4.3 Payment of Exercise Price.

- (a) Forms of Consideration Authorized. Except as otherwise provided below, payment of the aggregate Exercise Price for the number of shares of Stock for which the Option is being exercised shall be made (i) in cash, by check or in cash equivalent, (ii) if permitted by the Company and subject to the limitations contained in Section 4.3(b), by means of (1) a Cashless Exercise, (2) a Net-Exercise, or (3) a Stock Tender Exercise; or (iii) by any combination of the foregoing.
- (b) *Limitations on Forms of Consideration.* The Company reserves, at any and all times, the right, in the Company's sole and absolute discretion, to establish, decline to approve or terminate any program or procedure providing for payment of the Exercise Price through any of the means described below, including with respect to the Participant notwithstanding that such program or procedures may be available to others.
- (i) **Cashless Exercise**. A "Cashless Exercise" means the delivery of a properly executed Exercise Notice together with irrevocable instructions to a broker in a form acceptable to the Company providing for the assignment to the Company of the proceeds of a sale or loan with respect to shares of Stock acquired upon the exercise of the Option in an amount not less than the aggregate Exercise Price for such shares (including, without limitation, through an exercise complying with the provisions of Regulation T as promulgated from time to time by the Board of Governors of the Federal Reserve System).
- (ii) **Net-Exercise.** A "Net-Exercise" means the delivery of a properly executed Exercise Notice electing a procedure pursuant to which (1) the Company will reduce the number of shares otherwise issuable to the Participant upon the exercise of the Option by the largest whole number of shares having a Fair Market Value that does not exceed the aggregate Exercise Price for the shares with respect to which the Option is exercised, and (2) the Participant shall pay to the Company in cash the remaining balance of such aggregate Exercise Price not satisfied by such reduction in the number of whole shares to be issued. Following a Net-Exercise, the number of shares remaining subject to the Option, if any, shall be reduced by the sum of (1) the net number of shares issued to the Participant upon such exercise, and (2) the number of shares deducted by the Company for payment of the aggregate Exercise Price.
- (iii) **Stock Tender Exercise.** A "**Stock Tender Exercise**" means the delivery of a properly executed Exercise Notice accompanied by (1) the Participant's tender to the Company, or attestation to the ownership, in a form acceptable to the Company of whole shares of Stock having a Fair Market Value that does not exceed the aggregate Exercise Price for the shares with respect to which the Option is exercised, and (2) the Participant's payment to the Company in cash of the remaining balance of such aggregate Exercise Price not satisfied by such shares' Fair Market Value. A Stock Tender Exercise shall not be permitted if it would constitute a violation of the provisions of any law, regulation or agreement restricting the redemption of the Company's stock. If required by the Company, the Option may not be exercised by tender to the Company, or attestation to the ownership, of shares of Stock unless such shares either have been owned by the Participant for a period of time required by the Company (and not used for another option exercise by attestation during such period) or were not acquired, directly or indirectly, from the Company.

#### 4.4 Tax Withholding.

- (a) *In General.* At the time the Option is exercised, in whole or in part, or at any time thereafter as requested by a Participating Company, the Participant hereby authorizes withholding from payroll and any other amounts payable to the Participant, and otherwise agrees to make adequate provision for (including by means of a Cashless Exercise to the extent permitted by the Company), any sums required to satisfy the federal, state, local and foreign tax (including any social insurance) withholding obligations of the Participating Company Group, if any, which arise in connection with the Option. The Company shall have no obligation to deliver shares of Stock until the tax withholding obligations of the Participating Company Group have been satisfied by the Participant.
- (b) *Withholding in Shares.* The Company shall have the right, but not the obligation, to require the Participant to satisfy all or any portion of a Participating Company's tax withholding obligations upon exercise of the Option by deducting from the shares of Stock otherwise issuable to the Participant upon such exercise a number of whole shares having a fair market value, as determined by the Company as of the date of exercise, not in excess of the amount of such tax withholding obligations determined by the applicable minimum statutory withholding rates.
- 4.5 **Beneficial Ownership of Shares; Certificate Registration.** The Participant hereby authorizes the Company, in its sole discretion, to deposit for the benefit of the Participant with any broker with which the Participant has an account relationship of which the Company has notice any or all shares acquired by the Participant pursuant to the exercise of the Option. Except as provided by the preceding sentence, a certificate for the shares as to which the Option is exercised shall be registered in the name of the Participant, or, if applicable, in the names of the heirs of the Participant.
- 4.6 **Restrictions on Grant of the Option and Issuance of Shares.** The grant of the Option and the issuance of shares of Stock upon exercise of the Option shall be subject to compliance with all applicable requirements of federal, state or foreign law with respect to such securities. The Option may not be exercised if the issuance of shares of Stock upon exercise would constitute a violation of any applicable federal, state or foreign securities laws or other law or regulations or the requirements of any stock exchange or market system upon which the Stock may then be listed. In addition, the Option may not be exercised unless (i) a registration statement under the Securities Act shall at the time of exercise of the Option be in effect with respect to the shares issuable upon exercise of the Option or (ii) in the opinion of legal counsel to the Company, the shares issuable upon exercise of the Option may be issued in accordance with the terms of an applicable exemption from the registration requirements of the Securities Act. THE PARTICIPANT IS CAUTIONED THAT THE OPTION MAY NOT BE EXERCISED UNLESS THE FOREGOING CONDITIONS ARE SATISFIED. ACCORDINGLY, THE PARTICIPANT MAY NOT BE ABLE TO EXERCISE THE OPTION WHEN DESIRED EVEN THOUGH THE OPTION IS VESTED. The inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance and sale of any shares subject to the Option shall relieve the Company of any liability in respect of the failure to issue or sell such shares as to which such requisite authority shall not have been obtained. As a condition to the exercise of the Option, the

Company may require the Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company.

4.7 Fractional Shares. The Company shall not be required to issue fractional shares upon the exercise of the Option.

#### 5. NONTRANSFERABILITY OF THE OPTION.

During the lifetime of the Participant, the Option shall be exercisable only by the Participant or the Participant's guardian or legal representative. The Option shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant's beneficiary, except transfer by will or by the laws of descent and distribution. Following the death of the Participant, the Option, to the extent provided in Section 7, may be exercised by the Participant's legal representative or by any person empowered to do so under the deceased Participant's will or under the then applicable laws of descent and distribution.

#### 6. TERMINATION OF THE OPTION.

The Option shall terminate and may no longer be exercised after the first to occur of (a) the close of business on the Option Expiration Date, (b) the close of business on the last date for exercising the Option following termination of the Participant's Service as described in Section 7, or (c) a Change in Control to the extent provided in Section 8.

#### 7. EFFECT OF TERMINATION OF SERVICE.

- 7.1 **Option Exercisability.** The Option shall terminate immediately upon the Participant's termination of Service to the extent that it is then unvested and shall be exercisable after the Participant's termination of Service to the extent it is then vested only during the applicable time period as determined below and thereafter shall terminate.
- (a) *Disability*. If the Participant's Service terminates because of the Disability of the Participant, the Option, to the extent unexercised and exercisable for Vested Shares on the date on which the Participant's Service terminated, may be exercised by the Participant (or the Participant's guardian or legal representative) at any time prior to the expiration of twelve (12) months after the date on which the Participant's Service terminated, but in any event no later than the Option Expiration Date.
- (b) **Death.** If the Participant's Service terminates because of the death of the Participant, the Option, to the extent unexercised and exercisable for Vested Shares on the date on which the Participant's Service terminated, may be exercised by the Participant's legal representative or other person who acquired the right to exercise the Option by reason of the Participant's death at any time prior to the expiration of twelve (12) months after the date on which the Participant's Service terminated, but in any event no later than the Option Expiration Date. The Participant's Service shall be deemed to have terminated on account of death if the Participant dies within three (3) months after the Participant's termination of Service.

(c) *Termination for Cause*. Notwithstanding any other provision of this Option Agreement, if the Participant's Service is terminated for Cause, the Option shall terminate in its entirety and cease to be exercisable immediately upon such termination of Service.

(d) *Other Termination of Service*. If the Participant's Service terminates for any reason, except Disability, death or Cause, the Option, to the extent unexercised and exercisable for Vested Shares by the Participant on the date on which the Participant's Service terminated, may be exercised by the Participant at any time prior to the expiration of three (3) months after the date on which the Participant's Service terminated, but in any event no later than the Option Expiration Date.

7.2 Extension if Exercise Prevented by Law. Notwithstanding the foregoing other than termination of the Participant's Service for Cause, if the exercise of the Option within the applicable time periods set forth in Section 7.1 is prevented by the provisions of Section 4.6, the Option shall remain exercisable until the later of (a) thirty (30) days after the date such exercise first would no longer be prevented by such provisions or (b) the end of the applicable time period under Section 7.1, but in any event no later than the Option Expiration Date.

## 8. EFFECT OF CHANGE IN CONTROL.

In the event of a Change in Control, except to the extent that the Board determines to cash out the Option in accordance with Section 9.1(c) of the Plan, the surviving, continuing, successor, or purchasing corporation or other business entity or parent thereof, as the case may be (the "Acquiror"), may, without the consent of the Participant, assume or continue in full force and effect the Company's rights and obligations under all or any portion of the Option or substitute for all or any portion of the Option a substantially equivalent option for the Acquiror's stock. For purposes of this Section, the Option or any portion thereof shall be deemed assumed if, following the Change in Control, the Option confers the right to receive, subject to the terms and conditions of the Plan and this Option Agreement, for each share of Stock subject to such portion of the Option immediately prior to the Change in Control, the consideration (whether stock, cash, other securities or property or a combination thereof) to which a holder of a share of Stock on the effective date of the Change in Control was entitled; provided, however, that if such consideration is not solely common stock of the Acquiror, the Board may, with the consent of the Acquiror, provide for the consideration to be received upon the exercise of the Option for each share of Stock to consist solely of common stock of the Acquiror equal in Fair Market Value to the per share consideration received by holders of Stock pursuant to the Change in Control. If any portion of such consideration may be received by holders of Stock pursuant to the Change in Control on a contingent or delayed basis, the Board may, in its discretion, determine such Fair Market Value per share as of the time of the Change in Control on the basis of the Board's good faith estimate of the present value of the probable future payment of such consideration. The Option shall terminate and cease to be outstanding effective as of the time of consummation of the Change in Control to the extent that the Option is neither assumed or continued by the Acquiror in connection with the Change in Control nor exercised as of the time of the Change in Control. Notwithstanding the foregoing, shares acquired upon exercise of the Option prior to the Change in Control and any consideration received pursuant to the Change in Control with respect

to such shares shall continue to be subject to all applicable provisions of this Option Agreement except as otherwise provided herein.

#### 9. ADJUSTMENTS FOR CHANGES IN CAPITAL STRUCTURE.

Subject to any required action by the stockholders of the Company and the requirements of Sections 409A and 424 of the Code to the extent applicable, in the event of any change in the Stock effected without receipt of consideration by the Company, whether through merger, consolidation, reorganization, reincorporation, recapitalization, reclassification, stock dividend, stock split, reverse stock split, split-up, split-off, spin-off, combination of shares, exchange of shares, or similar change in the capital structure of the Company, or in the event of payment of a dividend or distribution to the stockholders of the Company in a form other than Stock (excepting normal cash dividends) that has a material effect on the Fair Market Value of shares of Stock, appropriate and proportionate adjustments shall be made in the number, Exercise Price and kind of shares subject to the Option, in order to prevent dilution or enlargement of the Participant's rights under the Option. For purposes of the foregoing, conversion of any convertible securities of the Company shall not be treated as "effected without receipt of consideration by the Company." Any fractional share resulting from an adjustment pursuant to this Section shall be rounded down to the nearest whole number, and the Exercise Price shall be rounded up to the nearest whole cent. In no event may the Exercise Price be decreased to an amount less than the par value, if any, of the stock subject to the Option. Such adjustments shall be determined by the Board, and its determination shall be final, binding and conclusive.

## 10. RIGHTS AS A STOCKHOLDER, DIRECTOR, EMPLOYEE OR CONSULTANT.

The Participant shall have no rights as a stockholder with respect to any shares covered by the Option until the date of the issuance of the shares for which the Option has been exercised (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for dividends, distributions or other rights for which the record date is prior to the date the shares are issued, except as provided in Section 9. If the Participant is an Employee, the Participant understands and acknowledges that, except as otherwise provided in a separate, written employment agreement between a Participating Company and the Participant, the Participant's employment is "at will" and is for no specified term. Nothing in this Option Agreement shall confer upon the Participant any right to continue in the Service of a Participating Company or interfere in any way with any right of the Participating Company Group to terminate the Participant's Service as a Director, an Employee or Consultant, as the case may be, at any time.

# 11. RIGHT OF FIRST REFUSAL.

11.1 **Grant of Right of First Refusal.** Except as provided in Section 11.7 and Section 16 below, in the event the Participant, the Participant's legal representative, or other holder of shares acquired upon exercise of the Option proposes to sell, exchange, transfer, pledge, or otherwise dispose of any Vested Shares (the "*Transfer Shares*") to any person or entity, including, without limitation, any stockholder of a Participating Company, the Company shall have the right to repurchase the Transfer Shares under the terms and subject to the conditions set forth in this Section 11 (the "*Right of First Refusal*").

- 11.2 **Notice of Proposed Transfer.** Prior to any proposed transfer of the Transfer Shares, the Participant shall deliver written notice (the "*Transfer Notice*") to the Company describing fully the proposed transfer, including the number of Transfer Shares, the name and address of the proposed transferee (the "*Proposed Transferee*") and, if the transfer is voluntary, the proposed transfer price, and containing such information necessary to show the bona fide nature of the proposed transfer. In the event of a bona fide gift or involuntary transfer, the proposed transfer price shall be deemed to be the Fair Market Value of the Transfer Shares, as determined by the Board in good faith. If the Participant proposes to transfer any Transfer Shares to more than one Proposed Transferee, the Participant shall provide a separate Transfer Notice for the proposed transfer to each Proposed Transferee. The Transfer Notice shall be signed by both the Participant and the Proposed Transferee and must constitute a binding commitment of the Participant and the Proposed Transferee for the transfer of the Transfer Shares to the Proposed Transferee subject only to the Right of First Refusal.
- 11.3 **Bona Fide Transfer.** If the Company determines that the information provided by the Participant in the Transfer Notice is insufficient to establish the bona fide nature of a proposed voluntary transfer, the Company shall give the Participant written notice of the Participant's failure to comply with the procedure described in this Section 11, and the Participant shall have no right to transfer the Transfer Shares without first complying with the procedure described in this Section 11. The Participant shall not be permitted to transfer the Transfer Shares if the proposed transfer is not bona fide.
- 11.4 Exercise of Right of First Refusal. If the Company determines the proposed transfer to be bona fide, the Company shall have the right to purchase all, but not less than all, of the Transfer Shares (except as the Company and the Participant otherwise agree) at the purchase price and on the terms set forth in the Transfer Notice by delivery to the Participant of a notice of exercise of the Right of First Refusal within thirty (30) days after the date the Transfer Notice is delivered to the Company. The Company's exercise or failure to exercise the Right of First Refusal with respect to any proposed transfer described in a Transfer Notice shall not affect the Company's right to exercise the Right of First Refusal with respect to any proposed transfer described in any other Transfer Notice, whether or not such other Transfer Notice is issued by the Participant or issued by a person other than the Participant with respect to a proposed transfer to the same Proposed Transferee. If the Company exercises the Right of First Refusal, the Company and the Participant shall thereupon consummate the sale of the Transfer Shares to the Company on the terms set forth in the Transfer Notice within sixty (60) days after the date the Transfer Notice is delivered to the Company (unless a longer period is offered by the Proposed Transferee); provided, however, that in the event the Transfer Notice provides for the payment for the Transfer Shares other than in cash, the Company shall have the option of paying for the Transfer Shares by the present value cash equivalent of the consideration described in the Transfer Notice as reasonably determined by the Company. For purposes of the foregoing, cancellation of any indebtedness of the Participant to any Participating Company shall be treated as payment to the Participant in cash to the extent of the unpaid principal and any accrued interest canceled. Notwithstanding anything contained in this Section to the contrary, the period during which the Company may exercise the Right of Fir

- 11.5 Failure to Exercise Right of First Refusal. If the Company fails to exercise the Right of First Refusal in full (or to such lesser extent as the Company and the Participant otherwise agree) within the period specified in Section 11.4 above, the Participant may conclude a transfer to the Proposed Transferee of the Transfer Shares on the terms and conditions described in the Transfer Notice, provided such transfer occurs not later than ninety (90) days following delivery to the Company of the Transfer Notice. The Company shall have the right to demand further assurances from the Participant and the Proposed Transferee (in a form satisfactory to the Company) that the transfer of the Transfer Shares was actually carried out on the terms and conditions described in the Transfer Notice. No Transfer Shares shall be transferred on the books of the Company until the Company has received such assurances, if so demanded, and has approved the proposed transfer as bona fide. Any proposed transfer on terms and conditions different from those described in the Transfer Notice, as well as any subsequent proposed transfer by the Participant, shall again be subject to the Right of First Refusal and shall require compliance by the Participant with the procedure described in this Section 11.
- 11.6 **Transferes of Transfer Shares.** All transferees of the Transfer Shares or any interest therein, other than the Company, shall be required as a condition of such transfer to agree in writing (in a form satisfactory to the Company) that such transfere shall receive and hold such Transfer Shares or interest therein subject to all of the terms and conditions of this Option Agreement, including this Section 11 providing for the Right of First Refusal with respect to any subsequent transfer. Any sale or transfer of any shares acquired upon exercise of the Option shall be void unless the provisions of this Section 11 are met.
- 11.7 **Transfers Not Subject to Right of First Refusal.** The Right of First Refusal shall not apply to any transfer or exchange of the shares acquired upon exercise of the Option if such transfer or exchange is in connection with an Ownership Change Event. If the consideration received pursuant to such transfer or exchange consists of stock of a Participating Company, such consideration shall remain subject to the Right of First Refusal unless the provisions of Section 11.9 below result in a termination of the Right of First Refusal.
- 11.8 **Assignment of Right of First Refusal.** The Company shall have the right to assign the Right of First Refusal at any time, whether or not there has been an attempted transfer, to one or more persons as may be selected by the Company.
- 11.9 Early Termination of Right of First Refusal. The other provisions of this Option Agreement notwithstanding, the Right of First Refusal shall terminate and be of no further force and effect upon (a) the occurrence of a Change in Control, unless the Acquiror assumes the Company's rights and obligations under the Option or substitutes a substantially equivalent option for the Acquiror's stock for the Option, or (b) the existence of a public market for the class of shares subject to the Right of First Refusal. A "public market" shall be deemed to exist if (i) such stock is listed on a national securities exchange (as that term is used in the Exchange Act) or (ii) such stock is traded on the over-the-counter market and prices therefor are published daily on business days in a recognized financial journal.

# 12. STOCK DISTRIBUTIONS SUBJECT TO OPTION AGREEMENT.

If, from time to time, there is any stock dividend, stock split or other change, as described in Section 9, in the character or amount of any of the outstanding stock of the corporation the stock of which is subject to the provisions of this Option Agreement, then in such event any and all new, substituted or additional securities to which the Participant is entitled by reason of the Participant's ownership of the shares acquired upon exercise of the Option shall be immediately subject to the Right of First Refusal with the same force and effect as the shares subject to the Right of First Refusal immediately before such event.

## 13. NOTICE OF SALES UPON DISQUALIFYING DISPOSITION.

The Participant shall dispose of the shares acquired pursuant to the Option only in accordance with the provisions of this Option Agreement. In addition, if the Grant Notice designates this Option as an Incentive Stock Option, the Participant shall (a) promptly notify the Chief Financial Officer of the Company if the Participant disposes of any of the shares acquired pursuant to the Option within one (1) year after the date the Participant exercises all or part of the Option or within two (2) years after the Date of Grant and (b) provide the Company with a description of the circumstances of such disposition. Until such time as the Participant disposes of such shares in a manner consistent with the provisions of this Option Agreement, unless otherwise expressly authorized by the Company, the Participant shall hold all shares acquired pursuant to the Option in the Participant's name (and not in the name of any nominee) for the one-year period immediately after the exercise of the Option and the two-year period immediately after Date of Grant. At any time during the one-year or two-year periods set forth above, the Company may place a legend on any certificate representing shares acquired pursuant to the Option requesting the transfer agent for the Company's stock to notify the Company of any such transfers. The obligation of the Participant to notify the Company of any such transfer shall continue notwithstanding that a legend has been placed on the certificate pursuant to the preceding sentence.

# 14. **LEGENDS.**

The Company may at any time place legends referencing the Right of First Refusal and any applicable federal, state or foreign securities law restrictions on all certificates representing shares of stock subject to the provisions of this Option Agreement. The Participant shall, at the request of the Company, promptly present to the Company any and all certificates representing shares acquired pursuant to the Option in the possession of the Participant in order to carry out the provisions of this Section. Unless otherwise specified by the Company, legends placed on such certificates may include, but shall not be limited to, the following:

14.1 "THE SECURITIES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD, TRANSFERRED, ASSIGNED OR HYPOTHECATED UNLESS THERE IS AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT COVERING SUCH SECURITIES, THE SALE IS MADE IN ACCORDANCE WITH RULE 144 OR RULE 701 UNDER THE ACT, OR THE COMPANY RECEIVES AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE COMPANY, STATING THAT SUCH

SALE, TRANSFER, ASSIGNMENT OR HYPOTHECATION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF SUCH ACT."

14.2 "THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND REPURCHASE OPTIONS IN FAVOR OF THE CORPORATION OR ITS ASSIGNEE SET FORTH IN AN AGREEMENT BETWEEN THE CORPORATION AND THE REGISTERED HOLDER, OR SUCH HOLDER'S PREDECESSOR IN INTEREST, A COPY OF WHICH IS ON FILE AT THE PRINCIPAL OFFICE OF THIS CORPORATION."

14.3 "THE SHARES EVIDENCED BY THIS CERTIFICATE WERE ISSUED BY THE CORPORATION TO THE REGISTERED HOLDER UPON EXERCISE OF AN INCENTIVE STOCK OPTION AS DEFINED IN SECTION 422 OF THE INTERNAL REVENUE CODE OF 1986, AS AMENDED ("ISO"). IN ORDER TO OBTAIN THE PREFERENTIAL TAX TREATMENT AFFORDED TO ISOs, THE SHARES SHOULD NOT BE TRANSFERRED PRIOR TO THE LATTER OF ONE YEAR FROM THE DATE OF EXERCISE OR TWO YEARS FROM THE DATE OF GRANT. SHOULD THE REGISTERED HOLDER ELECT TO TRANSFER ANY OF THE SHARES PRIOR TO THIS DATE AND FOREGO ISO TAX TREATMENT, THE TRANSFER AGENT FOR THE SHARES SHALL NOTIFY THE CORPORATION IMMEDIATELY. THE REGISTERED HOLDER SHALL HOLD ALL SHARES PURCHASED UNDER THE INCENTIVE STOCK OPTION IN THE REGISTERED HOLDER'S NAME (AND NOT IN THE NAME OF ANY NOMINEE) PRIOR TO THIS DATE OR UNTIL TRANSFERRED AS DESCRIBED ABOVE."

## 15. LOCK-UP AGREEMENT.

The Participant hereby agrees that in the event of any underwritten public offering of stock, including an initial public offering of stock, made by the Company pursuant to an effective registration statement filed under the Securities Act, the Participant shall not offer, sell, contract to sell, pledge, hypothecate, grant any option to purchase or make any short sale of, or otherwise dispose of any shares of stock of the Company or any rights to acquire stock of the Company for such period of time from and after the effective date of such registration statement as may be established by the underwriter for such public offering; provided, however, that such period of time shall not exceed one hundred eighty (180) days from the effective date of the registration statement to be filed in connection with such public offering. The foregoing limitation shall not apply to shares registered in the public offering under the Securities Act. The Participant hereby agrees to enter into any agreement reasonably required by the underwriters to implement the foregoing within a reasonable timeframe if so requested by the Company.

# 16. RESTRICTIONS ON TRANSFER OF SHARES.

No shares acquired upon exercise of the Option may be sold, exchanged, transferred (including, without limitation, any transfer to a nominee or agent of the Participant), assigned, pledged, hypothecated or otherwise disposed of, including by operation of law in any manner which violates any of the provisions of this Option Agreement, and any such attempted disposition shall be void. The Company shall not be required (a) to transfer on its books any

shares which will have been transferred in violation of any of the provisions set forth in this Option Agreement or (b) to treat as owner of such shares or to accord the right to vote as such owner or to pay dividends to any transferred to whom such shares will have been so transferred.

# 17. MISCELLANEOUS PROVISIONS.

- 17.1 **Termination or Amendment.** The Board may terminate or amend the Plan or the Option at any time; provided, however, that except as provided in Section 8 in connection with a Change in Control, no such termination or amendment may adversely affect the Option or any unexercised portion hereof without the consent of the Participant unless such termination or amendment is necessary to comply with any applicable law or government regulation, including, but not limited to Section 409A of the Code. No amendment or addition to this Option Agreement shall be effective unless in writing.
- 17.2 Compliance with Section 409A. The Company intends that income realized by the Participant pursuant to the Plan and this Option Agreement will not be subject to taxation under Section 409A of the Code. The provisions of the Plan and this Option Agreement shall be interpreted and construed in favor of satisfying any applicable requirements of Section 409A of the Code. The Company, in its reasonable discretion, may amend (including retroactively) the Plan and this Agreement in order to conform to the applicable requirements of Section 409A of the Code, including amendments to facilitate the Participant's ability to avoid taxation under Section 409A of the Code. However, the preceding provisions shall not be construed as a guarantee by the Company of any particular tax result for income realized by the Participant pursuant to the Plan or this Option Agreement. In any event, and except for the responsibilities of the Company set forth in Section 4.4, no Participating Company shall be responsible for the payment of any applicable taxes incurred by the Participant on income realized by the Participant pursuant to the Plan or this Option Agreement.
- 17.3 **Further Instruments.** The parties hereto agree to execute such further instruments and to take such further action as may reasonably be necessary to carry out the intent of this Option Agreement.
- 17.4 **Binding Effect.** This Option Agreement shall inure to the benefit of the successors and assigns of the Company and, subject to the restrictions on transfer set forth herein, be binding upon the Participant and the Participant's heirs, executors, administrators, successors and assigns.
- 17.5 **Delivery of Documents and Notices.** Any document relating to participation in the Plan, or any notice required or permitted hereunder shall be given in writing and shall be deemed effectively given (except to the extent that this Option Agreement provides for effectiveness only upon actual receipt of such notice) upon personal delivery, electronic delivery at the e-mail address, if any, provided for the Participant by a Participating Company, or upon deposit in the U.S. Post Office or foreign postal service, by registered or certified mail, or with a nationally recognized overnight courier service, with postage and fees prepaid, addressed to the other party at the address of such party set forth in the Grant Notice or at such other address as such party may designate in writing from time to time to the other party.

- (a) **Description of Electronic Delivery.** The Plan documents, which may include but do not necessarily include: the Plan, the Grant Notice, this Option Agreement, and any reports of the Company provided generally to the Company's stockholders, may be delivered to the Participant electronically. In addition, if permitted by the Company, the Participant may deliver electronically the Grant Notice and Exercise Notice called for by Section 4.2 to the Company or to such third party involved in administering the Plan as the Company may designate from time to time. Such means of electronic delivery may include but do not necessarily include the delivery of a link to a Company intranet or the Internet site of a third party involved in administering the Plan, the delivery of the document via e-mail or such other means of electronic delivery specified by the Company.
- (b) **Consent to Electronic Delivery.** The Participant acknowledges that the Participant has read Section 17.5(a) of this Option Agreement and consents to the electronic delivery of the Plan documents and, if permitted by the Company, the delivery of the Grant Notice and Exercise Notice, as described in Section 17.5(a). The Participant acknowledges that he or she may receive from the Company a paper copy of any documents delivered electronically at no cost to the Participant by contacting the Company by telephone or in writing. The Participant further acknowledges that the Participant will be provided with a paper copy of any documents if the attempted electronic delivery of such documents fails. Similarly, the Participant understands that the Participant must provide the Company or any designated third party administrator with a paper copy of any documents if the attempted electronic delivery of such documents fails. The Participant may revoke his or her consent to the electronic delivery of documents described in Section 17.5(a) or may change the electronic mail address to which such documents are to be delivered (if Participant has provided an electronic mail address) at any time by notifying the Company of such revoked consent or revised e-mail address by telephone, postal service or electronic mail. Finally, the Participant understands that he or she is not required to consent to electronic delivery of documents described in Section 17.5(a).
- 17.6 **Integrated Agreement.** The Grant Notice, this Option Agreement and the Plan, together with any employment, service or other agreement with the Participant and a Participating Company referring to the Option, shall constitute the entire understanding and agreement of the Participant and the Participating Company Group with respect to the subject matter contained herein or therein and supersede any prior agreements, understandings, restrictions, representations, or warranties among the Participant and the Participating Company Group with respect to such subject matter. To the extent contemplated herein or therein, the provisions of the Grant Notice, the Option Agreement and the Plan shall survive any exercise of the Option and shall remain in full force and effect.
- 17.7 **Applicable Law.** This Option Agreement shall be governed by the laws of the State of California as such laws are applied to agreements between California residents entered into and to be performed entirely within the State of California.
- 17.8 **Counterparts.** The Grant Notice may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

™ Incentive Stock Option	Participant:
M Nonstatutory Stock Option	Date:
STOCI	K OPTION EXERCISE NOTICE
socket, inc. Attention: Chief Financial Officer 270 East Lane, Suite #2 Burlingame, CA 94010	
Ladies and Gentlemen:	
	ase shares of the common stock (the "Shares") of isocket, inc. (the "Company") pursuant to ice of Grant of Stock Option (the "Grant Notice") and my Stock Option Agreement (the
Date of Grant:	
Number of Option Shares:	
Exercise Price per Share:	\$
Exercise of Option. I hereby elect to exercise the Option with the Grant Notice and the Option Agreement:	to purchase the following number of Shares, all of which are Vested Shares, in accordance
Total Shares Purchased:	
Total Exercise Price (Total Shares X Price per Share)	\$
3. Payments. I enclose payment in full of the total exercise	price for the Shares in the following form(s), as authorized by my Option Agreement:
™ Cash:	\$
TM Check:	\$
™ Cashless Exercise:	Contact Plan Administrator
™ Net Exercise:	Contact Plan Administrator
TM Stock Tender Exercise:	Contact Plan Administrator
	therwise will make adequate provision for the federal, state, local and foreign tax ith the Option. If I am exercising a Nonstatutory Stock Option, I enclose payment in full of
(Contact Plan	n Administrator for amount of tax due.)
™ Cash:	\$
TM Check:	\$
	1

5. Participant Information.	
My address is:	
My Social Security Number is:	
6. Notice of Disqualifying Disposition. If the Option is an Incenti Company if I transfer any of the Shares within one (1) year from the date	ive Stock Option, I agree that I will promptly notify the Chief Financial Officer of the e I exercise all or part of the Option or within two (2) years of the Date of Grant.
7. <u>Binding Effect</u> . I agree that the Shares are being acquired in acthe Option Agreement, including the Right of First Refusal set forth the inure to the benefit of and be binding upon my heirs, executors, administration	cordance with and subject to the terms, provisions and conditions of the Grant Notice, rein, and the Plan, to all of which I hereby expressly assent. This Agreement shall strators, successors and assigns.
Act"), and that consequently the Shares must be held indefinitely unles registration is available, or they are sold in accordance with Rule 144 or Company is under no obligation to register the Shares. I understand that	not been registered under the Securities Act of 1933, as amended (the "Securities is they are subsequently registered under the Securities Act, an exemption from such reference Rule 701 under the Securities Act. I further understand and acknowledge that the the certificate or certificates evidencing the Shares will be imprinted with legends chargistration is not required in the opinion of legal counsel satisfactory to the
available with respect to the Shares and, in any event, is available only	limited public resale of securities acquired in a nonpublic offering, is not currently if certain conditions are satisfied. I understand that any sale of the Shares that might its in accordance with the terms and conditions of such rule and that a copy of
I understand that I am purchasing the Shares pursuant to the terms received and carefully read and understand.	s of the Plan, the Grant Notice and my Option Agreement, copies of which I have
	Very truly yours,
	(Signature)
Receipt of the above is hereby acknowledged.	
ISOCKET, INC.	
Ву:	
Title:	
Dated:	
	2

# ISOCKET, INC. NOTICE OF GRANT OF STOCK OPTION

The Participant has been granted an option (the "Option") to purchase certain shares of Stock of isocket, inc. pursuant to the isocket, inc. 2009 Equity Incentive Plan (the "Plan"), as follows:

«Participant»

Participant:

Date of Grant:	«Date_of_Grant»		
Number of Option Shares:	«Shares», subject to adjustment as provide	ded by the Option Agreement.	
Exercise Price:	\$ <u>«Exercise_Price»</u>		
Initial Vesting Date:	The date one (1) year after «Vesting_Con	nmencement_Date»	
Option Expiration Date:	The date ten (10) years after the Date of O	Grant	
Tax Status of Option:	$\label{eq:continuity} Stock\ Option.\ (Enter\ ``Incentive"\ Option.)$	or "Nonstatutory." If blank, this Option will be a Nonstatutory	Stock
Vested Shares:		Agreement, the number of Vested Shares (disregarding any resumed by multiplying the Number of Option Shares by the "Vest	
			Vested Ratio
	Prior to Initial Vesting Date		0
	On Initial Vesting Date, provided the Par	ticipant's Service has not terminated prior to such date	1/4
	<u>Plus</u>		
	For each additional full month of the Par until the Vested Ratio equals 1/1, an add	ticipant's continuous Service from Initial Vesting Date itional	1/48
in good faith in compliance with the requir with the Company's determination. A subsconsequences to the Participant. By signing	ements of Section 409A of the Code. How equent IRS determination that the Exercis g below, the Participant agrees that the Co y the Participant as a result of such determ	e fair market value of a share of Stock as of the Date of Grant, do vever, there is no guarantee that the Internal Revenue Service we Price is less than such fair market value could result in adversing and shareholders shall not be helpination by the IRS. The Participant is urged to consult with hit tion of Section 409A.	will agree rse tax ld liable for
Stock Option Agreement, both of which are	attached to and made a part of this docur	governed by this Grant Notice and by the provisions of the Pla nent. The Participant acknowledges receipt of copies of the Pl their provisions, and hereby accepts the Option subject to all	an and the
ISOCKET, INC.		PARTICIPANT	
Ву:			
		Signature	
Its:		Date	

Address:	270 East Lane, Suite #2	
	Burlingame, CA 94010	Address

ATTACHMENTS: 2009 Equity Incentive Plan, as amended to the Date of Grant; Stock Option Agreement and Exercise Notice

# ISOCKET, INC. NOTICE OF GRANT OF STOCK PURCHASE RIGHT

The Participant has been granted a right to purchase (the "*Purchase Right*") certain shares of Stock (the "*Shares*") of isocket, inc. pursuant to the isocket, inc. 2009 equity Incentive Plan (the "*Plan*"), as follows:

D 41 1				
Participant:				
Date of Grai	nt:			
Total Numb	er of Shares:	, subject to adjustment as provided by the Stock Purchase A	greement.	
Purchase Pr	ice:	\$		
Expiration I	Date:	(Not later than 30 days after Date of Grant)		
Initial Vestir	ng Date:	The date one (1) year after [vesting commencement date]		
Vested Share	es:	Except as provided in the Stock Purchase Agreement, the number of Ve fractional share) as of any date is determined by multiplying the Total Netermined as of such date as follows:		
				Vested Ratio
		Prior to Initial Vesting Date		0
		On Initial Vesting Date, provided the Participant's Service has not term	inated prior to such date	1/4
		<u>Plus</u>		
		For each additional full month of the Participant's continuous Service funtil the Vested Ratio equals 1/1, an additional	rom Initial Vesting Date	1/48
governed by document. Tl	this Grant Notice and by the ne Participant acknowledges	pany and the Participant agree that the Purchase Right and any shares acq provisions of the Plan and the Stock Purchase Agreement, both of which a receipt of copies of the Plan and the Stock Purchase Agreement, represent accepts the Purchase Right subject to all of their terms and conditions.	are attached to and made a pa	rt of this
ISOCKET, IN	IC.	PARTICIPANT		
Ву:				
		Signature		
Its:				
		Date		
Address: 22	35 California street, #199			
M	ountain View, CA 94040	Address		
				_
ATTACHME	1 2	entive Plan, as amended to the Date of Grant; Stock Purchase Agreement, and form of Section 83(b) Election	Exercise Notice, Assignment	Separate

THE SECURITIES WHICH ARE THE SUBJECT OF THIS AGREEMENT HAVE NOT BEEN QUALIFIED WITH THE COMMISSIONER OF CORPORATIONS OF THE STATE OF CALIFORNIA AND THE ISSUANCE OF SUCH SECURITIES OR THE PAYMENT OR RECEIPT OF ANY PART OF THE CONSIDERATION THEREFOR PRIOR TO SUCH QUALIFICATION IS UNLAWFUL, UNLESS THE SALE OF SECURITIES IS EXEMPT FROM QUALIFICATION BY SECTION 25100, 25102, OR 25105 OF THE CALIFORNIA CORPORATIONS CODE. THE RIGHTS OF ALL PARTIES TO THIS AGREEMENT ARE EXPRESSLY CONDITIONED UPON SUCH QUALIFICATION BEING OBTAINED, UNLESS THE SALE IS SO EXEMPT.

THE SECURITIES WHICH ARE THE SUBJECT OF THIS AGREEMENT HAVE BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO, OR IN CONNECTION WITH, THE SALE OR DISTRIBUTION THEREOF. NO SUCH SALE OR DISPOSITION MAY BE EFFECTED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933.

## ISOCKET, INC. STOCK PURCHASE AGREEMENT

isocket, inc. has granted to the Participant named in the Notice of Grant of Stock Purchase Right (the "Grant Notice") to which this Stock Purchase Agreement (the "Agreement") is attached a Purchase Right consisting of a right to purchase Shares upon the terms and conditions set forth in the Grant Notice and this Agreement. The Purchase Right has been granted pursuant to and shall in all respects be subject to the terms and conditions of the isocket, inc. 2009 Equity Incentive Plan (the "Plan"), as amended to the Date of Grant, the provisions of which are incorporated herein by reference. By signing the Grant Notice, the Participant: (a) acknowledges receipt of, and represents that the Participant has read and is familiar with, the Grant Notice, this Agreement and the Plan, (b) accepts the Purchase Right subject to all of the terms and conditions of the Grant Notice, this Agreement and the Plan, and (c) agrees to accept as binding, conclusive and final all decisions or interpretations of the Board upon any questions arising under the Grant Notice, this Agreement or the Plan.

## 1. **DEFINITIONS AND CONSTRUCTION.**

1.1 **Definitions.** Unless otherwise defined herein, capitalized terms shall have the meanings assigned to such terms in the Grant Notice or the Plan.

1.2 **Construction.** Captions and titles contained herein are for convenience only and shall not affect the meaning or interpretation of any provision of this Agreement. Except when otherwise indicated by the context, the singular shall include the plural and the plural shall include the singular. Use of the term "or" is not intended to be exclusive, unless the context clearly requires otherwise.

## 2. TAX MATTERS.

- 2.1 **Election under Section 83(b) of the Code.** The Participant understands that Section 83 of the Code taxes as ordinary income the difference between the amount paid for the Shares and the fair market value of the Shares as of the date on which the shares are "substantially vested," within the meaning of Section 83. In this context, "substantially vested" means that the right of the Company to reacquire the Shares pursuant to the Unvested Share Repurchase Option contained in Section 7 of this Agreement has lapsed. The Participant understands that he or she may elect to have his or her taxable income determined at the time he or she acquires the Shares rather than when and as the Unvested Share Repurchase Option lapses by filing an election under Section 83(b) of the Code with the Internal Revenue Service no later than thirty (30) days after the date of acquisition of the Shares. Even if the fair market value of the Shares equals the amount paid for the Shares, the election must be made to avoid adverse tax consequences in the future. The Participant understands that failure to make a timely filing under Section 83(b) will result his or her recognition of ordinary income as the Unvested Share Repurchase Option lapses on the difference between the purchase price paid for the Shares and the fair market value of the Shares at the time the Unvested Share Repurchase Option lapses.
- 2.2 **Notice to Company.** The Participant will notify the Company in writing if the Participant files an election pursuant to Section 83(b) of the Code. The Company intends, in the event it does not receive from the Participant evidence of such filing, to claim a tax deduction for any amount which would otherwise be taxable to the Participant in the absence of such an election.
- 2.3 Consultation with Tax Advisors. The Participant understands that he or she should consult with his or her tax advisor regarding the advisability of filing with the IRS an election under Section 83(b) of the Code, which must be filed no later than thirty (30) days after the date of the acquisition of the Shares pursuant to this Agreement. Failure to file an election under Section 83(b), if appropriate, may result in adverse tax consequences to the Participant. The Participant acknowledges that he or she has been advised to consult with a tax advisor regarding the tax consequences to the Participant of the purchase of Shares hereunder. ANY ELECTION UNDER SECTION 83(b) THE PARTICIPANT WISHES TO MAKE MUST BE FILED NO LATER THAN 30 DAYS AFTER THE DATE ON WHICH THE PARTICIPANT PURCHASES THE SHARES. THIS TIME PERIOD CANNOT BE EXTENDED. THE PARTICIPANT ACKNOWLEDGES THAT TIMELY FILING OF A SECTION 83(b) ELECTION IS PARTICIPANT'S SOLE RESPONSIBILITY, EVEN IF THE PARTICIPANT REQUESTS THE COMPANY OR ITS REPRESENTATIVE TO FILE SUCH ELECTION ON HIS OR HER BEHALF.

## 3. **ADMINISTRATION.**

All questions of interpretation concerning the Grant Notice, this Agreement, the Plan or any other form of agreement or other document employed by the Company in the administration of the Plan or the Award shall be determined by the Board. All such determinations by the Board shall be final, binding and conclusive upon all persons having an interest in the Award, unless fraudulent or made in bad faith. Any and all actions, decisions and determinations taken or made by the Board in the exercise of its discretion pursuant to the Plan

or the Award or other agreement thereunder (other than determining questions of interpretation pursuant to the preceding sentence) shall be final, binding and conclusive upon all persons having an interest in the Award. Any Officer shall have the authority to act on behalf of the Company with respect to any matter, right, obligation, or election which is the responsibility of or which is allocated to the Company herein, provided the Officer has apparent authority with respect to such matter, right, obligation, or election.

## 4. EXERCISE OF PURCHASE RIGHT.

- 4.1 **Exercise of Purchase Right.** Provided that the Participant's Service has not terminated (except as provided by Section 6), the Purchase Right shall be exercisable on and after the Date of Grant and prior to the Expiration Date in an amount not to exceed the Total Number of Shares, subject to the Company's repurchase rights set forth in Sections 7 and 8.
- 4.2 Method of Exercise of Purchase Right. Exercise of the Purchase Right shall be by means of electronic or written notice (the "Exercise Notice") in a form authorized by the Company. An electronic Exercise Notice must be digitally signed or authenticated by the Participant in such manner as required by the notice and transmitted to the Company or an authorized representative of the Company (including a third-party administrator designated by the Company). In the event that the Participant is not authorized or is unable to provide an electronic Exercise Notice, the Purchase Right shall be exercised by a written Exercise Notice addressed to the Company, which shall be signed by the Participant and delivered in person, by certified or registered mail, return receipt requested, by confirmed facsimile transmission, or by such other means as the Company may permit, to the Company, or an authorized representative of the Company (including a third-party administrator designated by the Company). Each Exercise Notice, whether electronic or written, must state the Participant's election to exercise the Purchase Right, the number of whole Shares for which the Purchase Right is being exercised and such other representations and agreements as to the Participant's investment intent with respect to such Shares as may be required pursuant to the provisions of this Agreement. Further, each Exercise Notice must be received by the Company prior to the Expiration Date and must be accompanied by (a) full payment of the aggregate Purchase Price for the number of Shares being purchased and (b) if the Purchase Right is exercised with respect to any Unvested Shares (as defined in Section 7.1), an Assignment Separate from Certificate duly endorsed (with date and number of shares blank) in the form attached to the Grant Notice. The Purchase Right shall be deemed to be exercised upon receipt by the Company of such written notice, the aggregate Purchase Price, and, if required hereby, such duly endorsed Assignment Separate from Certificate
- 4.3 **Payment of Purchase Price.** Payment of the aggregate Purchase Price for the number of Shares for which the Purchase Right is being exercised shall be made in cash or by check or cash equivalent.
- 4.4 **Tax Withholding.** At the time the Purchase Right is exercised, in whole or in part, or at any time thereafter as requested by the Company, the Participant hereby authorizes withholding from payroll and any other amounts payable to the Participant, and otherwise agrees to make adequate provision for any sums required to satisfy the federal, state, local and foreign tax withholding obligations of the Participating Company Group, if any, which

arise in connection with the Shares acquired pursuant to this Agreement, including, without limitation, obligations arising upon (i) the exercise, in whole or in part, of the Purchase Right, (ii) the transfer, in whole or in part, of any Shares acquired, or (ii) the lapsing of any restriction with respect to any Shares acquired. The Company shall have no obligation to deliver Shares or to release Shares from an escrow established pursuant to this Agreement until the tax withholding obligations of the Participating Company Group have been satisfied by the Participant.

- 4.5 **Beneficial Ownership of Shares; Certificate Registration.** The Participant hereby authorizes the Company, in its sole discretion, to deposit the Shares with the Company's transfer agent, including any successor transfer agent, to be held in book entry form during the term of the Escrow pursuant to Section 9. Furthermore, the Participant hereby authorizes the Company, in its sole discretion, to deposit, following the term of such Escrow, for the benefit of the Participant with any broker with which the Participant has an account relationship of which the Company has notice any or all Shares which are no longer subject to such Escrow. Except as provided by the foregoing, a certificate for the Shares shall be registered in the name of the Participant, or, if applicable, in the names of the Participant.
- 4.6 Restrictions on Sale and Issuance of Shares. The sale and issuance of Shares shall be subject to compliance with all applicable requirements of federal, state or foreign law with respect to such securities. The Purchase Right may not be exercised if the issuance of Shares upon exercise would constitute a violation of any applicable federal, state or foreign securities laws or other law or regulations or the requirements of any stock exchange or market system upon which the Stock may then be listed. In addition, the Purchase Right may not be exercised unless (i) a registration statement under the Securities Act shall at the time of exercise of the Purchase Right be in effect with respect to the Shares issuable upon exercise of the Purchase Right or (ii) in the opinion of legal counsel to the Company, the Shares issuable upon exercise of the Purchase Right may be issued in accordance with the terms of an applicable exemption from the registration requirements of the Securities Act. The inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance and sale of any Shares subject to the Purchase Right shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained. As a condition to the exercise of the Purchase Right, the Company may require the Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company.
  - 4.7 Fractional Shares. The Company shall not be required to issue fractional Shares upon the exercise of the Purchase Right.

# 5. **VESTING OF SHARES.**

Shares acquired pursuant to this Agreement shall become Vested Shares as provided in the Grant Notice. For purposes of determining the number of Vested Shares following an Ownership Change Event, credited Service shall include all Service with any

corporation which is a Participating Company at the time the Service is rendered, whether or not such corporation is a Participating Company both before and after the Ownership Change Event.

## 6. NONTRANSFERABILITY OF PURCHASE RIGHT.

During the lifetime of the Participant, the Purchase Right shall be exercisable only by the Participant or the Participant's guardian or legal representative. The Purchase Right shall not be subject in any manner to anticipation, alienation, sale, exchange, transfer, assignment, pledge, encumbrance, or garnishment by creditors of the Participant or the Participant's beneficiary, except transfer by will or by the laws of descent and distribution. Following the death of the Participant, the Purchase Right may be exercised prior to the Expiration Date by the Participant's legal representative or by any person empowered to do so under the deceased Participant's will or under the then applicable laws of descent and distribution.

## 7. <u>Unvested Share Repurchase Option</u>.

- 7.1 **Grant of Unvested Share Repurchase Option.** In the event the Participant's Service is terminated for any reason or no reason, with or without cause, or, if the Participant, the Participant's legal representative, or other holder of Shares acquired pursuant to this Agreement, attempts to sell, exchange, transfer, pledge, or otherwise dispose of (other than pursuant to an Ownership Change Event) any Unvested Shares, as defined in Section 7.2 below (the "Unvested Shares"), the Company shall have the right to repurchase the Unvested Shares under the terms and subject to the conditions set forth in this Section 7 (the "Unvested Share Repurchase Option").
- 7.2 **Unvested Shares Defined.** The "*Unvested Shares*" shall mean, on any given date, the number of Shares acquired upon exercise of the Purchase Right which exceed the Vested Shares determined as of such date.
- 7.3 Exercise of Unvested Share Repurchase Option. The Company may exercise the Unvested Share Repurchase Option by written notice to the Participant within sixty (60) days after (a) termination of the Participant's Service or (b) the Company has received notice of the attempted disposition of Unvested Shares. If the Company fails to give notice within such sixty (60) day period, the Unvested Share Repurchase Option shall terminate unless the Company and the Participant have extended the time for the exercise of the Unvested Share Repurchase Option. Notwithstanding the preceding sentence, the period during which the Company may exercise the Unvested Share Repurchase Option shall terminate no sooner than the completion of a period of eight (8) months following the date on which the Participant acquired the Unvested Shares upon exercise of the Purchase Right. The Unvested Share Repurchase Option must be exercised, if at all, for all of the Unvested Shares, except as the Company and the Participant otherwise agree.
- 7.4 **Payment for Shares and Return of Shares to Company.** The purchase price per Share being repurchased by the Company shall be an amount equal to the Participant's original cost per Share, as adjusted pursuant to Section 11 (the "*Repurchase Price*"). The Company shall pay the aggregate Repurchase Price to the Participant in cash within thirty (30) days after the date of the written notice to the Participant of the Company's exercise of the

Unvested Share Repurchase Option. For purposes of the foregoing, cancellation of any purchase money indebtedness of the Participant to any Participating Company for the Shares shall be treated as payment to the Participant in cash to the extent of the unpaid principal and any accrued interest canceled. The Shares being repurchased shall be delivered to the Company by the Participant at the same time as the delivery of the Repurchase Price to the Participant.

- 7.5 **Assignment of Unvested Share Repurchase Option.** The Company shall have the right to assign the Unvested Share Repurchase Option at any time, whether or not such option is then exercisable, to one or more persons as may be selected by the Company.
- 7.6 Ownership Change Event. Upon the occurrence of an Ownership Change Event, any and all new, substituted or additional securities or other property to which the Participant is entitled by reason of the Participant's ownership of Unvested Shares shall be immediately subject to the Unvested Share Repurchase Option and included in the terms "Stock" and "Unvested Shares" for all purposes of the Unvested Share Repurchase Option with the same force and effect as the Unvested Shares immediately prior to the Ownership Change Event. While the aggregate Repurchase Price shall remain the same after such Ownership Change Event, the Repurchase Price per Unvested Share upon exercise of the Unvested Share Repurchase Option following such Ownership Change Event shall be adjusted as appropriate.

# 8. RIGHT OF FIRST REFUSAL.

- 8.1 **Grant of Right of First Refusal.** Except as provided in Section 8.7 and Section 13 below, in the event the Participant, the Participant's legal representative, or other holder of Shares acquired upon exercise of the Purchase Right proposes to sell, exchange, transfer, pledge, or otherwise dispose of any Vested Shares (the "*Transfer Shares*") to any person or entity, including, without limitation, any shareholder of a Participating Company, the Company shall have the right to repurchase the Transfer Shares under the terms and subject to the conditions set forth in this Section 8 (the "*Right of First Refusal*").
- 8.2 Notice of Proposed Transfer. Prior to any proposed transfer of the Transfer Shares, the Participant shall deliver written notice (the "Transfer Notice") to the Company describing fully the proposed transfer, including the number of Transfer Shares, the name and address of the proposed transferee (the "Proposed Transferee") and, if the transfer is voluntary, the proposed transfer price, and containing such information necessary to show the bona fide nature of the proposed transfer. In the event of a bona fide gift or involuntary transfer, the proposed transfer price shall be deemed to be the Fair Market Value of the Transfer Shares, as determined by the Board in good faith. If the Participant proposes to transfer any Transfer Shares to more than one Proposed Transferee, the Participant shall provide a separate Transfer Notice for the proposed transfer to each Proposed Transferee. The Transfer Notice shall be signed by both the Participant and the Proposed Transferee and must constitute a binding commitment of the Participant and the Proposed Transferee for the transfer of the Transfer Shares to the Proposed Transferee subject only to the Right of First Refusal.
- 8.3 **Bona Fide Transfer.** If the Company determines that the information provided by the Participant in the Transfer Notice is insufficient to establish the bona fide nature of a proposed voluntary transfer, the Company shall give the Participant written notice of the

Participant's failure to comply with the procedure described in this Section 8, and the Participant shall have no right to transfer the Transfer Shares without first complying with the procedure described in this Section 8. The Participant shall not be permitted to transfer the Transfer Shares if the proposed transfer is not bona fide.

8.4 Exercise of Right of First Refusal. If the Company determines the proposed transfer to be bona fide, the Company shall have the right to purchase all, but not less than all, of the Transfer Shares (except as the Company and the Participant otherwise agree) at the purchase price and on the terms set forth in the Transfer Notice by delivery to the Participant of a notice of exercise of the Right of First Refusal within thirty (30) days after the date the Transfer Notice is delivered to the Company. The Company's exercise or failure to exercise the Right of First Refusal with respect to any proposed transfer described in a Transfer Notice shall not affect the Company's right to exercise the Right of First Refusal with respect to any proposed transfer described in any other Transfer Notice, whether or not such other Transfer Notice is issued by the Participant or issued by a person other than the Participant with respect to a proposed transfer to the same Proposed Transferee. If the Company exercises the Right of First Refusal, the Company and the Participant shall thereupon consummate the sale of the Transfer Shares to the Company on the terms set forth in the Transfer Notice within sixty (60) days after the date the Transfer Notice is delivered to the Company (unless a longer period is offered by the Proposed Transferee); provided, however, that in the event the Transfer Notice provides for the payment for the Transfer Shares other than in cash, the Company shall have the option of paying for the Transfer Shares by the present value cash equivalent of the consideration described in the Transfer Notice as reasonably determined by the Company. For purposes of the foregoing, cancellation of any indebtedness of the Participant to any Participating Company shall be treated as payment to the Participant in cash to the extent of the unpaid principal and any accrued interest canceled. Notwithstanding anything contained in this Section to the contrary, the period during which the Company may exercise the Right of Firs

8.5 Failure to Exercise Right of First Refusal. If the Company fails to exercise the Right of First Refusal in full (or to such lesser extent as the Company and the Participant otherwise agree) within the period specified in Section 8.4 above, the Participant may conclude a transfer to the Proposed Transfere of the Transfer Shares on the terms and conditions described in the Transfer Notice, provided such transfer occurs not later than ninety (90) days following delivery to the Company of the Transfer Notice. The Company shall have the right to demand further assurances from the Participant and the Proposed Transferee (in a form satisfactory to the Company) that the transfer of the Transfer Shares was actually carried out on the terms and conditions described in the Transfer Notice. No Transfer Shares shall be transferred on the books of the Company until the Company has received such assurances, if so demanded, and has approved the proposed transfer as bona fide. Any proposed transfer on terms and conditions different from those described in the Transfer Notice, as well as any subsequent proposed transfer by the Participant, shall again be subject to the Right of First Refusal and shall require compliance by the Participant with the procedure described in this Section 8.

- 8.6 **Transferes of Transfer Shares.** All transferees of the Transfer Shares or any interest therein, other than the Company, shall be required as a condition of such transfer to agree in writing (in a form satisfactory to the Company) that such transferee shall receive and hold such Transfer Shares or interest therein subject to all of the terms and conditions of this Agreement, including this Section 6 providing for the Right of First Refusal with respect to any subsequent transfer. Any sale or transfer of any Shares acquired upon exercise of the Purchase Right shall be void unless the provisions of this Section 8 are met.
- 8.7 Transfers Not Subject to Right of First Refusal. The Right of First Refusal shall not apply to any transfer or exchange of the Shares acquired upon exercise of the Purchase Right if such transfer or exchange is in connection with an Ownership Change Event. If the consideration received pursuant to such transfer or exchange consists of stock of a Participating Company, such consideration shall remain subject to the Right of First Refusal unless the provisions of Section 8.9 below result in a termination of the Right of First Refusal.
- 8.8 **Assignment of Right of First Refusal.** The Company shall have the right to assign the Right of First Refusal at any time, whether or not there has been an attempted transfer, to one or more persons as may be selected by the Company.
- 8.9 Early Termination of Right of First Refusal. The other provisions of this Agreement notwithstanding, the Right of First Refusal shall terminate and be of no further force and effect upon (a) the occurrence of a Change in Control, unless the Acquiror assumes the Company's rights and obligations under this Agreement, or (b) the existence of a public market for the class of shares subject to the Right of First Refusal. A "public market" shall be deemed to exist if (i) such stock is listed on a national securities exchange (as that term is used in the Exchange Act) or (ii) such stock is traded on the over-the-counter market and prices therefor are published daily on business days in a recognized financial journal.

# 9. ESCROW.

- 9.1 Appointment of Agent. To ensure that Shares subject to the Unvested Share Repurchase Option will be available for repurchase, the Participant and the Company hereby appoint the Secretary of the Company, or any other person designated by the Company, as their agent and as attorney-in-fact for the Participant (the "Agent") to hold any and all Unvested Shares and to sell, assign and transfer to the Company any such Unvested Shares repurchased by the Company pursuant to the Unvested Share Repurchase Option. The Participant understands that appointment of the Agent is a material inducement to make this Agreement and that such appointment is coupled with an interest and is irrevocable. The Agent shall not be personally liable for any act the Agent may do or omit to do hereunder as escrow agent, agent for the Company, or attorney in fact for the Participant while acting in good faith and in the exercise of the Agent's own good judgment, and any act done or omitted by the Agent pursuant to the advice of the Agent's own attorneys shall be conclusive evidence of such good faith. The Agent may rely upon any letter, notice or other document executed by any signature purporting to be genuine and may resign at any time.
- 9.2 **Establishment of Escrow.** The Participant authorizes the Company to deposit the Unvested Shares with the Company's transfer agent to be held in book entry form, as

provided by Section 4.5, and the Participant agrees to deliver to and deposit with the Agent each certificate, if any, evidencing the Shares and an Assignment Separate from Certificate with respect to such book entry shares and each such certificate duly endorsed (with date and number of Shares blank) in the form attached to this Agreement, to be held by the Agent under the terms and conditions of this Section (the "Escrow"). Upon the occurrence of an Ownership Change Event, a dividend or distribution to the shareholders of the Company paid in shares of Stock or other property (other than regular, periodic dividends paid on Stock pursuant to the Company's dividend policy), or any other adjustment upon a change in the capital structure of the Company, as described in Section 11, any and all new, substituted or additional securities or other property to which the Participant is entitled by reason of his or her ownership of the Shares that remain, following such Ownership Change Event, dividend, distribution or change described in Section 11, subject to the Unvested Share Repurchase Option shall be immediately subject to the Escrow to the same extent as the Shares immediately before such event. The Company shall bear the expenses of the Escrow.

- 9.3 **Delivery of Shares to Participant.** The Escrow shall continue with respect to any Shares for so long as such Shares remain subject to the Unvested Share Repurchase Option. Upon termination of the Unvested Share Repurchase Option with respect to Shares, the Company shall so notify the Agent and direct the Agent to deliver such number of Shares to the Participant. As soon as practicable after receipt of such notice, the Agent shall the Shares specified by such notice to be delivered to the Participant, and the Escrow shall terminate with respect to such Shares.
- 9.4 **Notices and Payments.** In the event the Shares and any other property held in escrow are subject to the Company's exercise of the Unvested Share Repurchase Option or the Right of First Refusal, the notices required to be given to the Participant shall be given to the Agent, and any payment required to be given to the Participant shall be given to the Agent Shares and any other property which the Company has purchased to the Company and shall deliver the payment received from the Company to the Participant.

## 10. EFFECT OF CHANGE IN CONTROL.

In the event of a Change in Control, the surviving, continuing, successor, or purchasing corporation or other business entity or parent thereof, as the case may be (the "Acquiror"), may, without the consent of the Participant, assume or continue in full force and effect the Company's rights and obligations under the Purchase Right or substitute for the Purchase Right a substantially equivalent purchase right for the Acquiror's stock. For purposes of this Section, the Purchase Right shall be deemed assumed if, following the Change in Control, the Purchase Right confers the right to receive, subject to the terms and conditions of the Plan and this Agreement, for each Share subject to the Purchase Right immediately prior to the Change in Control, the consideration (whether stock, cash, other securities or property or a combination thereof) to which a holder of a share of Stock on the effective date of the Change in Control was entitled; provided, however, that if such consideration is not solely common stock of the Acquiror, the Board may, with the consent of the Acquiror, provide for the consideration to be received upon the exercise of the Purchase Right, for each Share subject to the Purchase Right, to consist solely of common stock of the Acquiror equal in Fair Market Value to the per

share consideration received by holders of Stock pursuant to the Change in Control. If any portion of such consideration may be received by holders of Stock pursuant to the Change in Control on a contingent or delayed basis, the Board may, in its sole discretion, determine such Fair Market Value per share as of the time of the Change in Control on the basis of the Board's good faith estimate of the present value of the probable future payment of such consideration. The Purchase Right shall terminate and cease to be outstanding effective as of the time of consummation of the Change in Control to the extent that the Purchase Right is neither assumed or continued by the Acquiror in connection with the Change in Control nor exercised as of the time of the Change in Control. Notwithstanding the foregoing, Shares acquired upon exercise of the Purchase Right prior to the Change in Control and any consideration received pursuant to the Change in Control with respect to such Shares shall continue to be subject to all applicable provisions of this Agreement except as otherwise provided herein.

#### 11. ADJUSTMENTS FOR CHANGES IN CAPITAL STRUCTURE.

Subject to any required action by the shareholders of the Company, in the event of any change in the Stock effected without receipt of consideration by the Company, whether through merger, consolidation, reorganization, reincorporation, recapitalization, reclassification, stock dividend, stock split, reverse stock split, split-up, split-off, spin-off, combination of shares, exchange of shares, or similar change in the capital structure of the Company, or in the event of payment of a dividend or distribution to the shareholders of the Company in a form other than Stock (excepting normal cash dividends) that has a material effect on the Fair Market Value of shares of Stock, appropriate and proportionate adjustments shall be made in the number, purchase price and class of shares of stock or other property subject to this Agreement, in order to prevent dilution or enlargement of the Participant's rights under the Agreement. Any fractional share resulting from an adjustment pursuant to this Section shall be rounded down to the nearest whole number and the purchase price shall be rounded up to the nearest whole cent. For purposes of the foregoing, conversion of any convertible securities of the Company shall not be treated as "effected without receipt of consideration by the Company." Any and all new, substituted or additional securities or other property to which Participant is entitled by reason of his ownership of Shares acquired pursuant to this Agreement will be immediately subject to the provisions of this Agreement on the same basis as all Shares originally purchased hereunder. For purposes of Sections 7 and 8 hereof, while the total price payable to exercise the rights provided in such sections will remain the same after each such event, the price payable per share to exercise such rights will be appropriately adjusted.

# 12. RIGHTS AS A SHAREHOLDER, DIRECTOR, EMPLOYEE OR CONSULTANT.

The Participant shall have no rights as a shareholder with respect to any Shares covered by the Purchase Right until the date of the issuance of the Shares for which the Purchase Right has been exercised (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for dividends, distributions or other rights for which the record date is prior to the date the Shares are issued, except as provided in Section 11. If the Participant is an Employee, the Participant understands and acknowledges that, except as otherwise provided in a separate, written employment agreement between a Participating Company and the Participant, the Participant's employment is "at will" and is for no specified term. Nothing in this Agreement shall confer upon the

Participant any right to continue in the Service of a Participating Company or interfere in any way with any right of the Participating Company Group to terminate the Participant's Service as a Director, an Employee or Consultant, as the case may be, at any time.

## 13. **LEGENDS.**

The Company may at any time place legends referencing the Unvested Share Repurchase Option, the Right of First Refusal and any applicable federal, state or foreign securities law restrictions on all certificates representing Shares subject to the provisions of this Agreement. The Participant shall, at the request of the Company, promptly present to the Company any and all certificates representing Shares acquired pursuant to this Agreement in the possession of the Participant in order to carry out the provisions of this Section. Unless otherwise specified by the Company, legends placed on such certificates may include, but shall not be limited to, the following:

13.1 "THE SECURITIES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD, TRANSFERRED, ASSIGNED OR HYPOTHECATED UNLESS THERE IS AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT COVERING SUCH SECURITIES, THE SALE IS MADE IN ACCORDANCE WITH RULE 144 OR RULE 701 UNDER THE ACT, OR THE COMPANY RECEIVES AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE COMPANY, STATING THAT SUCH SALE, TRANSFER, ASSIGNMENT OR HYPOTHECATION IS EXEMPT FROM THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF SUCH ACT."

13.2 "THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER AND REPURCHASE OPTIONS IN FAVOR OF THE CORPORATION OR ITS ASSIGNEE SET FORTH IN AN AGREEMENT BETWEEN THE CORPORATION AND THE REGISTERED HOLDER, OR SUCH HOLDER'S PREDECESSOR IN INTEREST, A COPY OF WHICH IS ON FILE AT THE PRINCIPAL OFFICE OF THIS CORPORATION."

# 14. LOCK-UP AGREEMENT.

The Participant hereby agrees that in the event of any underwritten public offering of stock, including an initial public offering of stock, made by the Company pursuant to an effective registration statement filed under the Securities Act, the Participant shall not offer, sell, contract to sell, pledge, hypothecate, grant any option to purchase or make any short sale of, or otherwise dispose of any shares of stock of the Company or any rights to acquire stock of the Company for such period of time from and after the effective date of such registration statement as may be established by the underwriter for such public offering; provided, however, that such period of time shall not exceed one hundred eighty (180) days from the effective date of the registration statement to be filed in connection with such public offering. The foregoing limitation shall not apply to shares registered in the public offering under the Securities Act. The Participant hereby agrees to enter into any agreement reasonably required by the underwriters to implement the foregoing within a reasonable timeframe if so requested by the Company.

# 15. RESTRICTIONS ON TRANSFER OF SHARES.

No Shares acquired upon exercise of the Purchase Right may be sold, exchanged, transferred (including, without limitation, any transfer to a nominee or agent of the Participant), assigned, pledged, hypothecated or otherwise disposed of, including by operation of law, in any manner which violates any of the provisions of this Agreement and, except pursuant to an Ownership Change Event, until the date on which such Shares become Vested Shares, and any such attempted disposition shall be void. The Company shall not be required (a) to transfer on its books any Shares which will have been transferred in violation of any of the provisions set forth in this Agreement or (b) to treat as owner of such Shares or to accord the right to vote as such owner or to pay dividends to any transferee to whom such Shares will have been so transferred.

## 16. MISCELLANEOUS PROVISIONS.

- 16.1 **Termination or Amendment.** The Board may terminate or amend the Plan or this Agreement at any time; provided, however, that except as provided in Section 10 in connection with a Change in Control, no such termination or amendment may adversely affect the Participant's rights under this Agreement without the consent of the Participant unless such termination or amendment is necessary to comply with any applicable law or government regulation, including, but not limited to, Section 409A of the Code. No amendment or addition to this Agreement shall be effective unless in writing.
- 16.2 Compliance with Section 409A. The Company intends that income realized by the Participant pursuant to the Plan and this Agreement will not be subject to taxation under Section 409A of the Code. The provisions of the Plan and this Agreement shall be interpreted and construed in favor of satisfying any applicable requirements of Section 409A of the Code. The Company, in its reasonable discretion, may amend (including retroactively) the Plan and this Agreement in order to conform to the applicable requirements of Section 409A of the Code, including amendments to facilitate the Participant's ability to avoid taxation under Section 409A of the Code. However, the preceding provisions shall not be construed as a guarantee by the Company of any particular tax result for income realized by the Participant pursuant to the Plan or this Agreement. In any event, and except for the responsibilities of the Company set forth in Section 4.4, no Participating Company shall be responsible for the payment of any applicable taxes incurred by the Participant on income realized by the Participant to the Plan or this Agreement.
- 16.3 **Further Instruments.** The parties hereto agree to execute such further instruments and to take such further action as may reasonably be necessary to carry out the intent of this Agreement.
- 16.4 **Binding Effect.** This Agreement shall inure to the benefit of the successors and assigns of the Company and, subject to the restrictions on transfer set forth herein, be binding upon the Participant and the Participant's heirs, executors, administrators, successors and assigns.

- 16.5 **Delivery of Documents and Notices.** Any document relating to participation in the Plan, or any notice required or permitted hereunder shall be given in writing and shall be deemed effectively given (except to the extent that this Agreement provides for effectiveness only upon actual receipt of such notice) upon personal delivery, electronic delivery at the e-mail address, if any, provided for the Participant by a Participating Company, or upon deposit in the U.S. Post Office or foreign postal service, by registered or certified mail, or with a nationally recognized overnight courier service, with postage and fees prepaid, addressed to the other party at the address of such party set forth in the Grant Notice or at such other address as such party may designate in writing from time to time to the other party.
- (a) **Description of Electronic Delivery.** The Plan documents, which may include but do not necessarily include: the Plan, the Grant Notice, this Agreement, and any reports of the Company provided generally to the Company's shareholders, may be delivered to the Participant electronically. In addition, if permitted by the Company, the Participant may deliver electronically the Grant Notice and Exercise Notice called for by Section 4.2 to the Company or to such third party involved in administering the Plan as the Company may designate from time to time. Such means of electronic delivery may include but do not necessarily include the delivery of a link to a Company intranet or the Internet site of a third party involved in administering the Plan, the delivery of the document via e-mail or such other means of electronic delivery specified by the Company.
- (b) **Consent to Electronic Delivery.** The Participant acknowledges that the Participant has read Section 16.5(a) of this Agreement and consents to the electronic delivery of the Plan documents and, if permitted by the Company, the delivery of the Grant Notice, Exercise Notice and notices in connection with the Escrow, as described in Section 16.5(a). The Participant acknowledges that he or she may receive from the Company a paper copy of any documents delivered electronically at no cost to the Participant by contacting the Company by telephone or in writing. The Participant further acknowledges that the Participant will be provided with a paper copy of any documents if the attempted electronic delivery of such documents fails. Similarly, the Participant understands that the Participant must provide the Company or any designated third party administrator with a paper copy of any documents if the attempted electronic delivery of such documents fails. The Participant may revoke his or her consent to the electronic delivery of documents described in Section 16.5(a) or may change the electronic mail address to which such documents are to be delivered (if Participant has provided an electronic mail address) at any time by notifying the Company of such revoked consent or revised e-mail address by telephone, postal service or electronic mail. Finally, the Participant understands that he or she is not required to consent to electronic delivery of documents described in Section 16.5(a).
- 16.6 **Integrated Agreement.** The Grant Notice, this Agreement and the Plan, together with any employment, service or other agreement between the Participant and a Participating Company referring to the Award, shall constitute the entire understanding and agreement of the Participant and the Participating Company Group with respect to the subject matter contained herein or therein and supersede any prior agreements, understandings, restrictions, representations, or warranties among the Participant and the Participating Company Group with respect to such subject matter. To the extent contemplated herein or therein, the

provisions of the Grant Notice, this Agreement and the Plan shall survive any exercise of the Purchase Right and shall remain in full force and effect.

- 16.7 **Applicable Law.** The Agreement shall be governed by the laws of the State of California as such laws are applied to agreements between California residents entered into and to be performed entirely within the State of California.
- 16.8 **Counterparts.** The Grant Notice may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

	Participant:	
	Date:	
NOTICE OF EXERCISE OF STO	CK PURCHASE RIGHT	
socket, inc. Attention: Chief Financial Officer 2235 Califomia Street, #199 Mountain View, CA 94040		
Ladies and Gentlemen:		
1. <u>Stock Purchase Right</u> . I was granted a right to purchase (the " <i>Purchase RightCompany</i> ") pursuant to the Company's 2009 Equity Incentive Plan (the " <i>Plan</i> "), a Stock Purchase Agreement (the " <i>Agreement</i> ") as follows:		
Date of Grant:		
Total Number of Shares:		
Purchase Price per Share:	\$	
2. Exercise of Purchase Right. I hereby elect to exercise my Purchase Right f	or the following number of Shares:	
Vested Shares:		
Unvested Shares:		
Total Shares Purchased:		
Total Purchase Price (Total Shares X Price per Share)	\$	
3. Payments. I enclose payment in full of the total purchase price for the Share	es in the following form(s), as authorized by my Agreement:	
□ Cash:	\$	
☐ Check:	\$	
4. <b>Tax Withholding.</b> I authorize payroll withholding and otherwise will make withholding obligations of the Company, if any, in connection with my purchase of follows:		
(Contact Plan Administrator fo	r amount of tax due.)	
☐ Cash:	\$	
☐ Check:	\$	
5 Participant Information		

1

My address is:

My Social Security Number is:			
including the Unvested Share Repurchase Option and the Right of shall inure to the benefit of and be binding upon my heirs, execute	in accordance with and subject to the terms, provisions and conditions of the Agreement, if First Refusal set forth therein, to all of which I hereby expressly assent. This Agreement ors, administrators, successors and assigns. If required by the Company, I agree to deposit assignment separate from certificate executed by me, with an escrow agent designated by the scrow Instructions.		
Act"), and that consequently the Shares must be held indefinitely registration is available, or they are sold in accordance with Rule I Company is under no obligation to register the Shares. I understan	have not been registered under the Securities Act of 1933, as amended (the "Securities unless they are subsequently registered under the Securities Act, an exemption from such 144 or Rule 701 under the Securities Act. I further understand and acknowledge that the dt that the certificate or certificates evidencing the Shares will be imprinted with legends or such registration is not required in the opinion of legal counsel satisfactory to the		
available with respect to the Shares and, in any event, is available	rmits limited public resale of securities acquired in a nonpublic offering, is not currently only if certain conditions are satisfied. I understand that any sale of the Shares that might mounts in accordance with the terms and conditions of such rule and that a copy of		
8. Election Under Section 83(b) of the Code. I understand and acknowledge that if I am exercising the Purchase Right to purchase Unvested Shares (i.e., shares that remain subject to the Company's Unvested Share Repurchase Option), that I should consult with my tax advisor regarding the advisability of filing with the Internal Revenue Service an election under Section 83(b) of the Code, which must be filed no later than thirty (30) days after the date on which I purchase the Shares. I acknowledge that I have been advised to consult with a tax advisor prior to the exercise of the Purchase Right regarding the tax consequences to me of exercising the Purchase Right. AN ELECTION UNDER SECTION 83(b) MUST BE FILED WITHIN 30 DAYS AFTER THE DATE ON WHICH I PURCHASE SHARES. THIS TIME PERIOD CANNOT BE EXTENDED. I ACKNOWLEDGE THAT TIMELY FILING OF A SECTION 83(b) ELECTION IS MY SOLE RESPONSIBILITY, EVEN IF I REQUEST THE COMPANY OR ITS REPRESENTATIVES TO FILE SUCH ELECTION ON MY BEHALF.			
I understand that I am purchasing the Shares pursuant to the carefully read and understand.	terms of the Plan, the Grant Notice and my Agreement, copies of which I have received and		
	Very truly yours,		
	(Signature)		
Receipt of the above is hereby acknowledged.			
ISOCKET, INC.			
Ву:			
Title:			
Dated:			
	2		

ASSIGNMEN	TI SEPARATE FROM CERTIFICATE
FOR VALUE RECEIVED the undersigned does hereby sell, a	assign and transfer unto
(	) shares of the Capital Stock of isocket, inc. standing in the undersigned's name on the
books of said corporation represented by Certificate No	herewith and does hereby irrevocably constitute and appoint
Attorney to transfer the said stock on the books of said con	rporation with full power of substitution in the premises.
Dated:	
	·
	Signature
	Print Name

<u>Instructions</u>: Please do not fill in any blanks other than the signature line. The purpose of this assignment is to enable the Company to exercise its Company Reacquisition Right set forth in the Restricted Stock Agreement without requiring additional signatures on the part of the Participant.

	SAMPLE
Inter	nal Revenue Service
_	Service Center re Form 1040 is Filed]
Re:	Section 83(b) Election
Dear	Sir or Madam:
	following information is submitted pursuant to section 1.83-2 of the Treasury Regulations in connection with this election by the undersigned under on 83(b) of the Internal Revenue Code of 1986, as amended (the "Code").
1.	The name, address and taxpayer identification number of the taxpayer are:
	Name:
	Address:
	Social Security Number:
2.	The following is a description of each item of property with respect to which the election is made:
	shares of common stock of isocket, inc. (the "Shares"), purchased from isocket, inc. (the "Company") pursuant to a stock purchase righ
3.	The property was transferred to the undersigned on:
	Stock purchase right exercise date:
	The taxable year for which the election is made is:
	Calendar Vear

4. The nature of the restriction to which the property is subject:

The Shares are subject to repurchase by the Company or its assignee upon the occurrence of certain events. This repurchase right lapses with regard to a portion of the Shares based upon the continued performance of services by the taxpayer over time.

5.	The following is the fair market value at the time of transfer (determined without regard to any restriction other than a restriction which by its terms will never lapse) of each property with respect to which the election is made:			
	\$ (	(	Shares at \$	per share).
	The pro	oper	ty was transferr	red to the taxpayer pursuant to the exercise of a stock purchase right.
6.	The following is the amount paid for the property:			
	\$ (	(	Shares at \$	per share).
7.	A copy of thi	is ele	ection has been	furnished to the Company, the corporation for which the services were performed by the undersigned.
Please acknowledge receipt of this election by date or received-stamping the enclosed copy of this letter and returning it to the undersigned. A self-addressed stamped envelope is provided for your convenience.				
Very truly yours,				
				Date:
Encl	osures			

cc: isocket, inc.

# THE RUBICON PROJECT, INC.

# 2014 INDUCEMENT GRANT EQUITY INCENTIVE PLAN

# ADOPTED BY THE BOARD: November 12, 2014

## 1. GENERAL.

- (a) Eligible Award Recipients. Any Eligible Individual shall be eligible to receive Awards.
- (c) Available Awards. The Plan provides for the grant of the following Awards: (i) Options; (ii) Stock Appreciation Rights; (iii) Restricted Stock Awards; (iv) Restricted Stock Unit Awards; (v) Performance Stock Awards; (vi) Performance Cash Awards; and (vii) Other Stock Awards.
- (d) **Purpose**. The purpose of the Plan is to provide terms and conditions to govern inducement grants made by the Company under NYSE Listed Company Manual Section 303A.08 ("Inducement Grants"). Such Inducement Grants are intended to assist the Company and any of its Affiliates in recruiting and retaining selected individuals to serve as employees of the Company and any Affiliate who are expected to contribute to the Company's success and to achieve long-term objectives that will benefit stockholders of the Company through the additional incentives inherent in the Inducement Grants.

## 2. ADMINISTRATION.

- (a) **Administration by Board**. The Board will administer the Plan. The Board may delegate administration of the Plan to a Committees, as provided in <u>Section 2(c)</u>.
  - (b) Powers of Board. The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:
- (i) To determine: (A) who will be granted Awards; (B) when and how each Award will be granted; (C) what type of Award will be granted; (D) the provisions of each Award (which need not be identical), including when a person will be permitted to exercise or otherwise receive cash or Common Stock under the Award; (E) the number of shares of Common Stock subject to, or the cash value of, an Award; and (F) the Fair Market Value applicable to a Stock Award.

- (ii) To construe and interpret the Plan and Awards granted under it, and to establish, amend and revoke rules and regulations for administration of the Plan and Awards. The Board, in the exercise of these powers, may correct any defect, omission or inconsistency in the Plan or in any Award Document or in the written terms of a Performance Cash Award, in a manner and to the extent it will deem necessary or expedient to make the Plan or Award fully effective.
  - (iii) To settle all controversies regarding the Plan and Awards granted under it.
- (iv) To accelerate, in whole or in part, or to extend, in whole or in part, the time during which an Award may be exercised or vest, or at which cash or shares of Common Stock may be issued.
- (v) To suspend or terminate the Plan at any time. Except as otherwise provided in the Plan, an employment agreement or offer letter, or an Award Document, suspension or termination of the Plan will not materially impair a Participant's rights under his or her then-outstanding Award without his or her written consent except as provided in subsection (vii) below.
- (vi) To amend the Plan in any respect the Board deems necessary or advisable, including, without limitation, adopting amendments relating to nonqualified deferred compensation under Section 409A of the Code and/or making the Plan or Awards granted under the Plan exempt from or compliant with the requirements for nonqualified deferred compensation under Section 409A of the Code, subject to the limitations, if any, of applicable law. Except as otherwise provided in the Plan (including subsection (vii) below), an employment agreement or offer letter, or an Award Document, no amendment of the Plan will materially impair a Participant's rights under an outstanding Award without the Participant's written consent.
- (vii) To approve forms of Award Documents for use under the Plan and to amend the terms of any one or more outstanding Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Award Documents for such Awards, subject to any specified limits in the Plan that are not subject to Board discretion. A Participant's rights under any Award will not be impaired by any such amendment unless the Company requests the consent of the affected Participant, and the Participant consents in writing. However, a Participant's rights will not be deemed to have been impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights. In addition, subject to the limitations of applicable law, if any, the Board may amend the terms of any one or more Awards without the affected Participant's consent (A) to clarify the manner of exemption from, or to bring the Award into compliance with, Section 409A of the Code, or (B) to comply with other applicable laws or listing requirements.
- (viii) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan and/or Award Documents.

(ix) To adopt such procedures and sub-plans as are necessary or appropriate (A) to permit or facilitate participation in the Plan by persons eligible to receive Awards under the Plan who are foreign nationals or employed outside the United States or (B) allow Awards to qualify for special tax treatment in a foreign jurisdiction; provided that Board approval will not be necessary for immaterial modifications to the Plan or any Award Document that are required for compliance with the laws of the relevant foreign jurisdiction.

# (c) Delegation to Committee.

- (i) General. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee of the Committee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee). Any delegation of administrative powers will be reflected in the charter of the Committee to which the delegation is made, or resolutions, not inconsistent with the provisions of the Plan, adopted from time to time by the Board or Committee (as applicable). The Committee may, at any time, abolish the subcommittee and/or revest in the Committee any powers delegated to any subcommittee. Unless otherwise provided by the Board, delegation of authority by the Board to a Committee, or to an Officer pursuant to Section 2(d), does not limit the authority of the Board, which may continue to exercise any authority so delegated and may concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated.
- (ii) Rule 16b-3 Compliance. The Committee may consist solely of two or more Non-Employee Directors, in accordance with Rule 16b-3 of the Exchange Act.
- (d) **Delegation to an Officer**. The Board may delegate to one (1) or more Officers the authority to do one or both of the following: (i) designate Eligible Individuals to be recipients of Options and SARs (and, to the extent permitted by applicable law, other Stock Awards) and, to the extent permitted by applicable law, the terms of such Awards; and (ii) determine the number of shares of Common Stock to be subject to such Stock Awards granted to such Eligible Individuals; provided, however, that the Board resolutions regarding such delegation will specify the total number of shares of Common Stock that may be subject to the Stock Awards granted by such Officer and that such Officer may not grant a Stock Award to himself or herself. Any such Stock Awards will be granted on the form of Stock Award Document approved by the Committee or the Board for use in connection with such Stock Awards, unless otherwise provided for in the resolutions approving the delegation authority. The Board may not delegate authority to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) to determine the Fair Market Value (as defined below).
- (e) **Effect of Board's Decision**. All determinations, interpretations and constructions made by the Board (or a duly authorized Committee, subcommittee or Officer exercising powers delegated by the Board under this <u>Section 2</u>) in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons.

## 3. SHARES SUBJECT TO THE PLAN.

## (a) Share Reserve.

- (i) Subject to Section 9(a) relating to Capitalization Adjustments, the aggregate maximum number of shares of Common Stock that may be issued pursuant to Stock Awards from and after the Effective Date (the "Share Reserve") will initially be 1,000,000 shares of Common Stock. The Share Reserve will increase from time to time by the Returning Shares (if and when the Returning Shares ever become available for grant under this Plan). The Board may in its discretion increase the number of Shares available for Awards under the Plan from time to time.
- (ii) For clarity, the Share Reserve is a limitation on the number of shares of Common Stock that may be issued under to the Plan. As a single share may be subject to grant more than once (e.g., if a share subject to a Stock Award is forfeited, it may be made subject to grant again as provided in Section 3(b) below), the Share Reserve is not a limit on the number of Stock Awards that can be granted.
- (b) Reversion of Shares to the Share Reserve. If a Stock Award or any portion of a Stock Award (i) expires or otherwise terminates without all of the shares covered by the Stock Award having been issued or (ii) is settled in cash (i.e., the Participant receives cash rather than stock), such expiration, termination or settlement will not reduce (or otherwise offset) the number of shares of Common Stock that are available for issuance under the Plan. If any shares of Common Stock issued under a Stock Award are forfeited back to or repurchased by the Company because of the failure to meet a contingency or condition required to vest such shares in the Participant, then the shares that are forfeited or repurchased will revert to and again become available for issuance under the Plan. Any shares retained or reacquired by the Company in satisfaction of tax withholding obligations on a Stock Award or as consideration for the exercise or purchase price of a Stock Award will again become available for issuance under the Plan.
- (c) **Source of Shares**. The stock issuable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.
- **4. ELIGIBILITY.** Any Eligible Individual shall be eligible to be selected as a Participant.

## 5. PROVISIONS RELATING TO OPTIONS AND STOCK APPRECIATION RIGHTS.

Each Option or SAR will be in such form and will contain such terms and conditions as the Board deems appropriate. The provisions of separate Options or SARs need not be identical; provided, however, that each Award Document will conform to (through incorporation of provisions hereof by reference in the applicable Award Document or otherwise) the substance of each of the following provisions:

(a) **Term**. No Option or SAR will be exercisable after the expiration of 10 years from the date of its grant or such shorter period specified in the Award Document.

- (b) Exercise Price. The exercise or strike price of each Option or SAR will be not less than 100% of the Fair Market Value of the Common Stock subject to the Option or SAR on the date the Award is granted. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise or strike price lower than 100% of the Fair Market Value of the Common Stock subject to the Award if such Award is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a corporate transaction and in a manner consistent with the provisions of Section 409A of the Code and, if applicable, Section 424(a) of the Code. Each SAR will be denominated in shares of Common Stock equivalents.
- (c) **Purchase Price for Options**. The purchase price of Common Stock acquired pursuant to the exercise of an Option may be paid, to the extent permitted by applicable law and as determined by the Board in its sole discretion, by any combination of the methods of payment set forth below. The Board will have the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to use a particular method of payment. The permitted methods of payment are as follows:
  - (i) by cash, check, bank draft or money order payable to the Company;
- (ii) pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board or a successor regulation, or a similar rule in a foreign jurisdiction of domicile of a Participant, that, prior to or contemporaneously with the issuance of the stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the proceeds of sale of such stock;
  - (iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock;
- (iv) by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issuable upon exercise by the largest whole number of shares with a Fair Market Value that does not exceed the aggregate exercise price; provided, however, that the Company will accept cash or other payment from the Participant to the extent of any remaining balance of the aggregate exercise price not satisfied by such reduction in the number of whole shares to be issued. Shares of Common Stock will no longer be subject to an Option and will not be exercisable thereafter to the extent that (A) shares issuable upon exercise are used to pay the exercise price pursuant to the "net exercise," (B) shares are delivered to the Participant as a result of such exercise, and (C) shares are withheld to satisfy tax withholding obligations; or
  - (v) in any other form of legal consideration that may be acceptable to the Board and specified in the applicable Award Document.
- (d) Exercise and Payment of a SAR. To exercise any outstanding SAR, the Participant must provide written notice of exercise to the Company in compliance with the provisions of the Stock Appreciation Right Award Document evidencing such SAR. The appreciation distribution payable on the exercise of a SAR will be not greater than an amount

equal to the excess of (A) the aggregate Fair Market Value (on the date of the exercise of the SAR) of a number of shares of Common Stock equal to the number of Common Stock equivalents in which the Participant is vested under such SAR (with respect to which the Participant is exercising the SAR on such date), over (B) the aggregate strike price of the number of Common Stock equivalents with respect to which the Participant is exercising the SAR on such date. The appreciation distribution may be paid in Common Stock, in cash, in any combination of the two or in any other form of consideration, as determined by the Board and contained in the Award Document evidencing such SAR.

- (e) **Transferability of Options and SARs**. The Board may, in its sole discretion, impose such limitations on the transferability of Options and SARs as the Board determines. In the absence of such a determination by the Board to the contrary, the following restrictions on the transferability of Options and SARs will apply:
- (i) <u>Restrictions on Transfer</u>. An Option or SAR will not be transferable except by will or by the laws of descent and distribution (or pursuant to subsections (ii) and (iii) below), and will be exercisable during the lifetime of the Participant only by the Participant. The Board may permit transfer of the Option or SAR in a manner that is not prohibited by applicable tax and securities laws. Except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration.
- (ii) <u>Domestic Relations Orders</u>. Subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to the terms of a domestic relations order, official marital settlement agreement or other divorce or separation instrument as permitted by U.S. Treasury Regulation 1.421-1(b)(2).
- (iii) <u>Beneficiary Designation</u>. Subject to the approval of the Board or a duly authorized Officer, a Participant may, by delivering written notice to the Company, in a form approved by the Company (or the designated broker), designate a third party who, on the death of the Participant, will thereafter be entitled to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise. In the absence of such a designation, the executor or administrator of the Participant's estate will be entitled to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise. However, the Company may prohibit designation of a beneficiary at any time, including due to any conclusion by the Company that such designation would be inconsistent with the provisions of applicable laws.
- (f) **Vesting Generally**. The total number of shares of Common Stock subject to an Option or SAR may vest and therefore become exercisable in periodic installments that may or may not be equal. The Option or SAR may be subject to such other terms and conditions on the time or times when it may or may not be exercised (which may be based on the satisfaction of Performance Goals or other criteria) as the Board may deem appropriate. The vesting provisions of individual Options or SARs may vary. The provisions of this <u>Section 5(f)</u> are subject to any Option or SAR provisions governing the minimum number of shares of Common Stock as to which an Option or SAR may be exercised.

- (g) **Termination of Continuous Service**. Except as otherwise provided in the applicable Award Document, or other agreement between the Participant and the Company, if a Participant's Continuous Service terminates (other than for Cause and other than upon the Participant's death or Disability), the Participant may exercise his or her Option or SAR (to the extent that the Participant was entitled to exercise such Award as of the date of termination of Continuous Service) within the period of time ending on the earlier of (i) the date three (3) months following the termination of the Participant's Continuous Service and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Award Document. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR within the applicable time frame, the Option or SAR will terminate.
- (h) Extension of Termination Date. Except as otherwise provided in the applicable Award Document, or other agreement between the Participant and the Company, if the exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause and other than upon the Participant's death or Disability) would be prohibited at any time solely because the issuance of shares of Common Stock would violate the registration requirements under the Securities Act, then the Option or SAR will terminate on the earlier of (i) the expiration of a total period of three (3) months (that need not be consecutive) after the termination of the Participant's Continuous Service during which the exercise of the Option or SAR would not be in violation of such registration requirements, and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Award Document. In addition, unless otherwise provided in a Participant's applicable Award Document or other agreement between the Participant and the Company, if the sale of any Common Stock received upon exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, and the Company does not waive the potential violation of the policy or otherwise permit the sale, or allow the Participant to surrender shares of Common Stock to the Company in satisfaction of any exercise price and/or any withholding obligations under Section 8(g), then the Option or SAR will terminate on the earlier of (i) the expiration of a period of months (that need not be consecutive) equal to the applicable post-termination exercise period after the termination of the Participant's Continuous Service during which the sale of the Common Stock received upon exercise of the Option or SAR would not be in violation of the Company's insider trading policy, or (ii) the expiration of the term of the Option or SAR as set forth in the applicable Award Document.
- (i) **Disability of Participant**. Except as otherwise provided in the applicable Award Document, or other agreement between the Participant and the Company, if a Participant's Continuous Service terminates as a result of the Participant's Disability, the Participant may exercise his or her Option or SAR (to the extent that the Participant was entitled to exercise such Option or SAR as of the date of termination of Continuous Service), but only within such period of time ending on the earlier of (i) the date 12 months following such termination of Continuous Service, and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Award Document. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR within the applicable time frame, the Option or SAR (as applicable) will terminate.
- (j) **Death of Participant**. Except as otherwise provided in the applicable Award Document, or other agreement between the Participant and the Company, if (i) a Participant's

Continuous Service terminates as a result of the Participant's death, or (ii) the Participant dies within the period (if any) specified in this Plan or the applicable Award Document, or other agreement between the Participant and the Company, for exercisability after the termination of the Participant's Continuous Service (for a reason other than death), then the Option or SAR may be exercised (to the extent the Participant was entitled to exercise such Option or SAR as of the date of death) by the Participant's estate, by a person who acquired the right to exercise the Option or SAR by bequest or inheritance or by a person designated to exercise the Option or SAR upon the Participant's death, but only within the period ending on the earlier of (i) the date 18 months following the date of death, and (ii) the expiration of the term of such Option or SAR as set forth in the applicable Award Document. If, after the Participant's death, the Option or SAR is not exercised within the applicable time frame, the Option or SAR will terminate.

- (k) **Termination for Cause**. Except as explicitly provided otherwise in a Participant's Award Document or other individual written agreement between the Company or any Affiliate and the Participant, if a Participant's Continuous Service is terminated for Cause, the Option or SAR will terminate upon the date on which the event giving rise to the termination for Cause first occurred, and the Participant will be prohibited from exercising his or her Option or SAR from and after the date on which the event giving rise to the termination for Cause first occurred (or, if required by law, the date of termination of Continuous Service). If a Participant's Continuous Service is suspended pending an investigation of the existence of Cause, all of the Participant's rights under the Option or SAR will also be suspended during the investigation period.
- (1) Non-Exempt Employees. If an Option or SAR is granted to an Employee who is a non-exempt employee for purposes of the U.S. Fair Labor Standards Act of 1938, as amended, the Option or SAR will not be first exercisable for any shares of Common Stock until at least 6 months following the date of grant of the Option or SAR (although the Award may vest prior to such date). Consistent with the provisions of the U.S. Worker Economic Opportunity Act, (i) if such non-exempt Employee dies or suffers a Disability, (ii) upon a Corporate Transaction in which such Option or SAR is not assumed, continued, or substituted, (iii) upon a Change in Control, or (iv) upon the non-exempt Employee's retirement (as such term may be defined in the non-exempt Employee's applicable Award Document, in another agreement between the non-exempt Employee and the Company, or, if no such definition, in accordance with the Company's then current employment policies and guidelines), the vested portion of any Options and SARs may be exercised earlier than 6 months following the date of grant. The foregoing provision is intended to operate so that any income derived by a non-exempt Employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay. To the extent permitted and/or required for compliance with the U.S. Worker Economic Opportunity Act to ensure that any income derived by a non-exempt Employee in connection with the exercise, vesting or issuance of any shares under any other Stock Award will be exempt from such Employee's regular rate of pay, the provisions of this paragraph will apply to all Stock Awards and are hereby incorporated by reference into such Stock Award Documents.

# 6. PROVISIONS OF STOCK AWARDS OTHER THAN OPTIONS AND SARS.

(a) **Restricted Stock Awards**. Each Restricted Stock Award Document will be in such form and will contain such terms and conditions as the Board deems appropriate. To the extent

consistent with the Company's bylaws, at the Board's election, shares of Common Stock may be (x) held in book entry form subject to the Company's instructions until any restrictions relating to the Restricted Stock Award lapse, or (y) evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. The terms and conditions of Restricted Stock Award Documents may change from time to time, and the terms and conditions of separate Restricted Stock Award Documents need not be identical. Each Restricted Stock Award Document will conform to (through incorporation of the provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

- (i) <u>Consideration</u>. A Restricted Stock Award may be awarded in consideration for (A) cash, check, bank draft or money order payable to the Company, (B) past services to the Company or an Affiliate, or (C) any other form of legal consideration (including future services) that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.
- (ii) <u>Vesting</u>. Shares of Common Stock awarded under the Restricted Stock Award Document may be subject to forfeiture to the Company in accordance with a vesting schedule and subject to such conditions as may be determined by the Board.
- (iii) <u>Termination of Participant's Continuous Service</u>. If a Participant's Continuous Service terminates, the Company may receive through a forfeiture condition or a repurchase right, any or all of the shares of Common Stock held by the Participant that have not vested as of the date of termination of Continuous Service under the terms of the Restricted Stock Award Document.
- (iv) <u>Transferability</u>. Common Stock issued pursuant to an Award, and rights to acquire shares of Common Stock under the Restricted Stock Award Document, will be transferable by the Participant only upon such terms and conditions as are set forth in the Restricted Stock Award Document, as the Board determines in its sole discretion, so long as such Common Stock remains subject to the terms of the Restricted Stock Award Document.
- (v) <u>Dividends</u>. A Restricted Stock Award Document may provide that any dividends paid on Restricted Stock will be subject to the same vesting and forfeiture restrictions as apply to the shares subject to the Restricted Stock Award to which they relate.
- (b) Restricted Stock Unit Awards. Each Restricted Stock Unit Award Document will be in such form and will contain such terms and conditions as the Board deems appropriate. The terms and conditions of Restricted Stock Unit Award Documents may change from time to time, and the terms and conditions of separate Restricted Stock Unit Award Documents need not be identical. Each Restricted Stock Unit Award Document will conform to (through incorporation of the provisions hereof by reference in the Agreement or otherwise) the substance of each of the following provisions:
- (i) <u>Consideration</u>. At the time of grant of a Restricted Stock Unit Award, the Board will determine the consideration, if any, to be paid by the Participant upon delivery of each share of Common Stock subject to the Restricted Stock Unit Award. The consideration to be paid (if any) by the Participant for each share of Common Stock subject to a Restricted Stock

Unit Award may be paid in any form of legal consideration that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.

- (ii) <u>Vesting</u>. At the time of the grant of a Restricted Stock Unit Award, the Board may impose such restrictions on or conditions to the vesting of the Restricted Stock Unit Award as it, in its sole discretion, deems appropriate.
- (iii) <u>Payment</u>. A Restricted Stock Unit Award may be settled by the delivery of shares of Common Stock, their cash equivalent, any combination thereof or in any other form of consideration, as determined by the Board and contained in the Restricted Stock Unit Award Document.
- (iv) <u>Additional Restrictions</u>. At the time of the grant of a Restricted Stock Unit Award, the Board, as it deems appropriate, may impose such restrictions or conditions that delay the delivery of the shares of Common Stock (or their cash equivalent) subject to a Restricted Stock Unit Award to a time after the vesting of such Restricted Stock Unit Award.
- (v) <u>Dividend Equivalents</u>. Dividend equivalents may be credited in respect of shares of Common Stock covered by a Restricted Stock Unit Award, as determined by the Board and contained in the Restricted Stock Unit Award Document. At the sole discretion of the Board, such dividend equivalents may be converted into additional shares of Common Stock covered by the Restricted Stock Unit Award in such manner as determined by the Board. Any additional shares covered by the Restricted Stock Unit Award credited by reason of such dividend equivalents will be subject to all of the same terms and conditions of the underlying Restricted Stock Unit Award Document to which they relate.
- (vi) <u>Termination of Participant's Continuous Service</u>. Except as otherwise provided in the applicable Restricted Stock Unit Award Document, or other agreement between the Participant and the Company, such portion of the Restricted Stock Unit Award that has not vested will be forfeited upon the Participant's termination of Continuous Service.

# (c) Performance Awards.

- (i) <u>Performance Stock Awards</u>. A Performance Stock Award is a Stock Award that is payable (including that may be granted, vest or exercised) contingent upon the attainment during a Performance Period of certain Performance Goals. A Performance Stock Award may, but need not, require the completion of a specified period of Continuous Service. The length of any Performance Period, the Performance Goals to be achieved during the Performance Period, and the measure of whether and to what degree such Performance Goals have been attained will be conclusively determined by the Committee, in its sole discretion. In addition, to the extent permitted by applicable law and the applicable Award Document, the Board may determine that cash may be used in payment of Performance Stock Awards.
- (ii) <u>Performance Cash Awards</u>. A Performance Cash Award is a cash award that is granted and/or becomes payable contingent upon the attainment during a Performance Period of certain Performance Goals. A Performance Cash Award may also require the completion of a specified period of Continuous Service. At the time of grant of a Performance Cash Award, the length of any Performance Period, the Performance Goals to be achieved during the Performance

Period, and the measure of whether and to what degree such Performance Goals have been attained will be conclusively determined by the Committee, in its sole discretion. The Board may specify the form of payment of Performance Cash Awards, which may be cash or other property, or may provide for a Participant to have the option for his or her Performance Cash Award, or such portion thereof as the Board may specify, to be paid in whole or in part in cash or other property.

- (iii) <u>Board Discretion</u>. The Committee retains the discretion to reduce or eliminate the compensation or economic benefit due upon attainment of Performance Goals and to define the manner of calculating the Performance Criteria it selects to use for a Performance Period.
- (d) Other Stock Awards. Other forms of Stock Awards valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof (e.g., options or stock rights with an exercise price or strike price less than 100% of the Fair Market Value of the Common Stock at the time of grant) may be granted either alone or in addition to Stock Awards provided for under Section 5 and the preceding provisions of this Section 6. Subject to the provisions of the Plan, the Board will have sole and complete authority to determine the persons to whom and the time or times at which such Other Stock Awards will be granted, the number of shares of Common Stock (or the cash equivalent thereof) to be granted pursuant to such Other Stock Awards and all other terms and conditions of such Other Stock Awards.

## 7. COVENANTS OF THE COMPANY.

- (a) **Availability of Shares**. The Company will keep available at all times the number of shares of Common Stock reasonably required to satisfy thenoutstanding Stock Awards.
- (b) Securities Law Compliance. The Company will seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Stock Awards and to issue and sell shares of Common Stock upon exercise of the Stock Awards; provided, however, that this undertaking will not require the Company to register under the Securities Act the Plan, any Stock Award or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company will be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained. A Participant will not be eligible for the grant of an Award or the subsequent issuance of cash or Common Stock pursuant to the Award if such grant or issuance would be in violation of any applicable securities law.
- (c) No Obligation to Notify or Minimize Taxes. The Company will have no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Stock Award. Furthermore, the Company will have no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of an Award or a possible period in which the Award may not be exercised. The Company has no duty or obligation to, and does not

undertake to, provide tax advice or to minimize the tax consequences of an Award to the holder of such Award.

## 8. MISCELLANEOUS.

- (a) Use of Proceeds from Sales of Common Stock. Proceeds from the sale of shares of Common Stock pursuant to Stock Awards will constitute general funds of the Company.
- (b) Corporate Action Constituting Grant of Awards. Corporate action constituting a grant by the Company of an Award to any Participant will be deemed completed as of the latest date that all necessary corporate action has occurred and all terms of the Award (including, in the case of stock options, the exercise price thereof) are fixed, unless otherwise determined by the Board, regardless of when the documentation evidencing the Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action constituting the grant contain terms (e.g., exercise price, vesting schedule or number of shares) that are inconsistent with those in the Award Document as a result of a clerical error in the papering of the Award Document, the corporate records will control and the Participant will have no legally binding right to the incorrect term in the Award Document.
- (c) Stockholder Rights. No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to a Stock Award unless and until (i) such Participant has satisfied all requirements for exercise of, or the issuance of shares of Common Stock under, the Stock Award pursuant to its terms, and (ii) the issuance of the Common Stock subject to such Stock Award has been entered into the books and records of the Company.
- (d) No Employment or Other Service Rights. Nothing in the Plan, any Award Document or any other instrument executed thereunder or in connection with any Award granted pursuant thereto will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or any other capacity or will affect the right of the Company or an Affiliate to terminate the employment of an Employee with or without notice and with or without cause, including, but not limited to, Cause.
- (e) Change in Time Commitment. If after the date of grant of any Award to the Participant, the Participant's regular level of time commitment in the performance of his or her services for the Company and any Affiliates is reduced (for example, and without limitation, if the Employee has a change in status from a full-time Employee to a part-time Employee or takes an extended leave of absence), or the Participant's role or primary responsibilities are changed to a level that, in the good faith determination by the Board does not justify the Participant's unvested Awards, the Board has the right in its sole discretion to (i) make a corresponding reduction in the number of shares or cash amount subject to any portion of such Award that is scheduled to vest or become payable after the date of such change in time commitment, and (ii) in lieu of or in combination with such a reduction, extend the vesting or payment schedule applicable to such Award. In the event of any such reduction, the Participant will have no right with respect to any portion of the Award that is so reduced or extended.

- (f) Investment Assurances. The Company may require a Participant, as a condition of exercising or acquiring Common Stock under any Stock Award, (i) to give written assurances satisfactory to the Company as to the Participant's knowledge and experience in financial and business matters and/or to employ a purchaser representative reasonably satisfactory to the Company who is knowledgeable and experienced in financial and business matters and that he or she is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Stock Award, and (ii) to give written assurances satisfactory to the Company stating that the Participant is acquiring Common Stock subject to the Stock Award for the Participant's own account and not with any present intention of selling or otherwise distributing the Common Stock. The foregoing requirements, and any assurances given pursuant to such requirements, will be inoperative if (i) the issuance of the shares upon the exercise of a Stock Award or acquisition of Common Stock under the Stock Award has been registered under a then currently effective registration statement under the Securities Act, or (ii) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities laws. The Company may, upon advice of counsel to the Company, place legends on stock certificates issued under the Plan as such counsel deems necessary or appropriate in order to comply with applicable securities laws, including, but not limited to, legends restricting the transfer of the Common Stock.
- (g) Withholding Obligations. Unless prohibited by the terms of an Award Document, the Company may, in its sole discretion, satisfy any U.S. federal, state, local, foreign or other tax withholding obligation relating to an Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Award; provided, however, that no shares of Common Stock are withheld with a value exceeding the minimum amount of tax required to be withheld by law (or such other amount as may be necessary to avoid classification of the Stock Award as a liability for financial accounting purposes); (iii) withholding cash from an Award settled in cash; (iv) withholding payment from any amounts otherwise payable to the Participant, including proceeds from the sale of shares of Common Stock issued pursuant to a Stock Award; or (v) by such other method as may be set forth in the Award Document.
- (h) **Electronic Delivery**. Any reference herein to a "written" agreement or document will include any agreement or document delivered electronically, filed publicly at www.sec.gov (or any successor website thereto), or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access).
- (i) **Deferrals**. To the extent permitted by applicable law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Award may be deferred and may establish programs and procedures for deferral elections to be made by Participants. Deferrals by Participants will be made in accordance with Section 409A of the Code (to the extent applicable to a Participant). Consistent with Section 409A of the Code, the Board may provide for distributions while a Participant is still an employee or otherwise providing services to the Company. The Board is authorized to make deferrals of Awards and determine when, and in what annual percentages, Participants may receive payments, including lump sum payments,

following the Participant's termination of Continuous Service, and implement such other terms and conditions consistent with the provisions of the Plan and in accordance with applicable law.

- (j) Compliance with Section 409A. Unless otherwise expressly provided for in an Award Document, or other agreement between the Participant and the Company, the Plan and Award Documents will be interpreted to the greatest extent possible in a manner that makes the Plan and the Awards granted hereunder exempt from Section 409A of the Code, and, to the extent not so exempt, in compliance with Section 409A of the Code. If the Board determines that any Award granted hereunder is not exempt from and is therefore subject to Section 409A of the Code, the Award Document evidencing such Award will incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code, and to the extent an Award Document is silent on terms necessary for compliance, such terms are hereby incorporated by reference into the Award Document. Notwithstanding anything to the contrary in this Plan (and unless the Award Document specifically provides otherwise), if the shares of Common Stock are publicly traded, and if a Participant holding an Award that constitutes "deferred compensation" under Section 409A of the Code is a "specified employee" for purposes of Section 409A of the Code, no distribution or payment of any amount that is due because of a "separation from service" (as defined in Section 409A of the Code without regard to alternative definitions thereunder) will be issued or paid before the date that is six (6) months following the date of such Participant's "separation from service" or, if earlier, the date of the Participant's death, unless such distribution or payment can be made in a manner that complies with Section 409A of the Code, and any amounts so deferred will be paid in a lump sum on the day after such six (6) month period elapses, with the balance paid thereafter on the original schedule.
- (k) Clawback/Recovery. All Awards granted under the Plan will be subject to recoupment in accordance with any clawback policy that the Company is required to adopt pursuant to the listing standards of any national securities exchange or association on which the Company's securities are listed or as is otherwise required by the Dodd-Frank Wall Street Reform and Consumer Protection Act or other applicable law. In addition, the Board may impose such other clawback, recovery or recoupment provisions in an Award Document as the Board determines necessary or appropriate, including, but not limited to, a reacquisition right in respect of previously acquired shares of Common Stock or other cash or property upon the occurrence of Cause. No recovery of compensation under such a clawback policy will be an event giving rise to a right to resign for "good reason" or "constructive termination" (or similar term) under any agreement with the Company or an Affiliate.

# 9. ADJUSTMENTS UPON CHANGES IN COMMON STOCK; OTHER CORPORATE EVENTS.

(a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board will appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to Section 3(a); and (ii) the class(es) and number of securities or other property and value (including price per share of stock) subject to outstanding Stock Awards. The Board will make such adjustments, and its determination will be final, binding and conclusive.

- (b) **Dissolution or Liquidation**. Except as otherwise provided in the Stock Award Document, or other agreement between the Participant and the Company, in the event of a dissolution or liquidation of the Company, all outstanding Stock Awards (other than Stock Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the shares of Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Stock Award is providing Continuous Service; provided, however, that the Board may, in its sole discretion, cause some or all Stock Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Stock Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.
- (c) Corporate Transaction. The following provisions will apply to Stock Awards in the event of a Corporate Transaction unless otherwise provided in the instrument evidencing the Stock Award or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of a Stock Award. In the event of a Corporate Transaction, then, notwithstanding any other provision of the Plan, the Board will take one or more of the following actions with respect to each outstanding Stock Award, contingent upon the closing or completion of the Corporate Transaction:
- (i) arrange for the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) to assume or continue the Stock Award or to substitute a similar stock award for the Stock Award (including, but not limited to, an award to acquire the same consideration per share paid to the stockholders of the Company pursuant to the Corporate Transaction);
- (ii) arrange for the assignment of any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to the Stock Award to the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company);
- (iii) accelerate the vesting, in whole or in part, of the Stock Award (and, if applicable, the time at which the Stock Award may be exercised) to a date prior to the effective time of such Corporate Transaction as the Board will determine (or, if the Board will not determine such a date, to the date that is 5 days prior to the effective date of the Corporate Transaction), with such Stock Award terminating if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction, and with such exercise reversed if the Corporate Transaction does not become effective;
  - (iv) arrange for the lapse, in whole or in part, of any reacquisition or repurchase rights held by the Company with respect to the Stock Award;
- (v) cancel or arrange for the cancellation of the Stock Award, to the extent not vested or not exercised prior to the effective time of the Corporate Transaction, in exchange for such cash consideration, if any, as the Board, in its reasonable determination, may consider appropriate as an approximation of the value of the canceled Stock Award, taking into account

the value of the Common Stock subject to the canceled Stock Award, the possibility that the Stock Award might not otherwise vest in full, and such other factors as the Board deems relevant; and

(vi) cancel or arrange for the cancellation of the Stock Award, to the extent not vested or not exercised prior to the effective time of the Corporate Transaction, in exchange for a payment, in such form as may be determined by the Board equal to the excess, if any, of (A) the value in the Corporate Transaction of the property the Participant would have received upon the exercise of the Stock Award immediately prior to the effective time of the Corporate Transaction, over (B) any exercise price payable by such holder in connection with such exercise.

The Board need not take the same action or actions with respect to all Stock Awards or portions thereof or with respect to all Participants. The Board may take different actions with respect to the vested and unvested portions of a Stock Award.

In the absence of any affirmative determination by the Board at the time of a Corporate Transaction, each outstanding Stock Award will be assumed or an equivalent Stock Award will be substituted by such successor corporation or a parent or substidiary of such successor corporation (the "Successor Corporation"), unless the Successor Corporation does not agree to assume the Stock Award or to substitute an equivalent Stock Award, in which case the vesting of such Stock Award will accelerate in its entirety (along with, if applicable, the time at which the Stock Award may be exercised) to a date prior to the effective time of such Corporate Transaction as the Board will determine (or, if the Board will not determine such a date, to the date that is 5 days prior to the effective date of the Corporate Transaction), with such Stock Award terminating if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction, and with such exercise reversed if the Corporate Transaction does not become effective.

(d) Change in Control. A Stock Award may be subject to additional acceleration of vesting and exercisability upon or after a Change in Control as may be provided in the Stock Award Document for such Stock Award or as may be provided in any other written agreement between the Company or any Affiliate and the Participant, but in the absence of such provision, no such acceleration will occur.

# 10. TERMINATION OR SUSPENSION OF THE PLAN.

The Board may suspend or terminate the Plan at any time. No Awards may be granted after the tenth (10th) anniversary of the Effective Date. No Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

## 11. EFFECTIVE DATE OF PLAN; TIMING OF FIRST GRANT OR EXERCISE.

The Plan came into existence on the Effective Date. No Award may be granted or exercised under the Plan prior to the Effective Date.

## 12. CHOICE OF LAW.

The laws of the State of Delaware will govern all questions concerning the construction, validity and interpretation of this Plan, without regard to that state's conflict of laws rules.

- 13. DEFINITIONS. As used in the Plan, the following definitions will apply to the capitalized terms indicated below:
- (a) "Affiliate" means, at the time of determination, any "parent" or "subsidiary" of the Company, as such terms are defined in Rule 405 of the Securities Act. The Board will have the authority to determine the time or times at which "parent" or "subsidiary" status is determined within the foregoing definition.
  - (b) "Award" means a Stock Award or a Performance Cash Award.
- (c) "Award Document" means a written agreement between the Company and a Participant, or a written notice issued by the Company to a Participant, evidencing the terms and conditions of an Award.
  - (d) "Board" means the Board of Directors of the Company.
  - (e) "Capital Stock" means each and every class of common stock of the Company, regardless of the number of votes per share.
- (f) "Capitalization Adjustment" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Stock Award after the Effective Date without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other similar equity restructuring transaction, as that term is used in Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.
- (g) "Cause" will have the meaning ascribed to such term in any written agreement between the Participant and the Company or any Affiliate defining such term and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) Participant's willful failure substantially to perform his or her duties and responsibilities to the Company or any Affiliate or deliberate material violation of a policy of the Company or any Affiliate; (ii) Participant's commission of any material act of fraud, embezzlement, dishonesty or any other willful misconduct that has caused or is reasonably expected to result in material injury to the Company or any Affiliate; (iii) unauthorized use or disclosure by Participant of any proprietary information or trade secrets of the Company or any other party to whom the Participant owes an obligation of nondisclosure as a result of his or her relationship with the Company or any Affiliate; or (iv) Participant's willful material breach of any of his or her obligations under any written agreement or covenant with the Company or any Affiliate. The determination as to whether a Participant is being terminated for Cause will be

made in good faith by the Company and will be final and binding on the Participant. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company, any Affiliate or such Participant for any other purpose.

- (h) "Change in Control" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control will not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities or (C) solely because the level of Ownership held by any Exchange Act Person (the "Subject Person") exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company reducing the number of shares outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control will be deemed to occur;
- (ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than 50% of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction or (B) more than 50% of the combined outstanding voting power of the parent of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction;
- (iii) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than 50% of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition; or

(iv) individuals who, on the Effective Date, are members of the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the appointment or election (or nomination for election) of any new Board member was approved or recommended by a majority vote of the members of the Incumbent Board then still in office, such new member will, for purposes of this Plan, be considered as a member of the Incumbent Board.

Notwithstanding the foregoing definition or any other provision of this Plan, (A) the term Change in Control will not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, and (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant will supersede the foregoing definition with respect to Awards subject to such agreement; provided, however, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition will apply.

If required for compliance with Section 409A of the Code, in no event will a Change in Control be deemed to have occurred if such transaction is not also a "change in the ownership or effective control of" the Company or "a change in the ownership of a substantial portion of the assets of" the Company as determined under U.S. Treasury Regulation Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder). The Board may, in its sole discretion and without a Participant's consent, amend the definition of "Change in Control" to conform to the definition of "Change in Control" under Section 409A of the Code, and the regulations thereunder.

- (i) "Code" means the U.S. Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.
- (j) "Committee" means a committee of one (1) or more Directors to whom authority has been delegated by the Board in accordance with Section 2(c).
- (k) "Common Stock" means the common stock, par value \$0.00001 per share, of the Company.
- (1) "Company" means The Rubicon Project, Inc., a Delaware corporation.
- (m) "Consultant" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan. Notwithstanding the foregoing, a person is treated as a Consultant under this Plan only if a Form Registration Statement on Form S-8 or a successor form under the Securities Act is available to register either the offer or the sale of the Company's securities to such person.
- (n) "Continuous Service" means that the Participant's service with the Company or an Affiliate is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee or a change in the Entity for

which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service. If the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board in its sole discretion, such Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. In addition, if required for exemption from or compliance with Section 409A of the Code, the determination of whether there has been a termination of Continuous Service will be made, and such term will be construed, in a manner that is consistent with the definition of "separation from service" as defined under U.S. Treasury Regulation Section 1.409A-1(h) (without regard to any alternative definition thereunder). A leave of absence will be treated as Continuous Service for purposes of vesting in a Stock Award only to such extent as may be provided in the applicable Award Document, the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law.

- (o) "Corporate Transaction" means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) a sale or other disposition of all or substantially all, as determined by the Board in its sole discretion, of the consolidated assets of the Company and its Subsidiaries;
  - (ii) a sale or other disposition of at least 90% of the outstanding securities of the Company;
  - (iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or
- (iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.

To the extent required for compliance with Section 409A of the Code, in no event will an event be deemed a Corporate Transaction if such transaction is not also a "change in the ownership or effective control of" the Company or "a change in the ownership of a substantial portion of the assets of" the Company as determined under U.S. Treasury Regulation Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder).

(p) "Director" means a member of the Board.

- (q) "Disability" means, with respect to a Participant, the inability of such Participant to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment that can be expected to result in death or that has lasted or can be expected to last for a continuous period of not less than 12 months as provided in Sections 22(e)(3) and 409A(a)(2)(C)(i) of the Code, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.
  - (r) "Effective Date" means the date the Plan is adopted by the Board.
- (s) "Eligible Individual" mans any individual who was not previously an Employee or a Non-Employee Director of the Company or any of its Subsidiaries (or who has had a bona fide period of non-employment with the Company and its Subsidiaries) who is hired by the Company or one of its Subsidiaries other than any individual who is being hired into a position within the Company or its Subsidiaries that will be subject to the reporting requirements of Section 16 of the Exchange Act.
- (t) "Employee" means any person providing services to the Company or an Affiliate. However, services solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.
  - (u) "Entity" means a corporation, partnership, limited liability company or other entity.
  - (v) "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
- (w) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company, or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities.
  - (x) "Fair Market Value" means, as of any date, the value of the Common Stock determined as follows:
- (i) If the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value of a share of Common Stock as of any date of determination will be, unless otherwise determined by the Board, the closing sales price for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) on the date of determination, as reported in a source the Board deems reliable.

- (ii) Unless otherwise provided by the Board, if there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing selling price on the last preceding date for which such quotation exists.
- (iii) In the absence of such markets for the Common Stock, the Fair Market Value will be determined by the Board in good faith and in a manner that complies with Sections 409A and 422 of the Code.
- (y) "Non-Employee Director" means a Director who either (i) is not a current employee or officer of the Company or an Affiliate, does not receive compensation, either directly or indirectly, from the Company or an Affiliate for services rendered as a consultant or in any capacity other than as a Director (except for an amount as to which disclosure would not be required under Item 404(a) of Regulation S-K promulgated pursuant to the Securities Act ("Regulation S-K")), does not possess an interest in any other transaction for which disclosure would be required under Item 404(a) of Regulation S-K, and is not engaged in a business relationship for which disclosure would be required pursuant to Item 404(b) of Regulation S-K; or (ii) is otherwise considered a "non-employee director" for purposes of Rule 16b-3 of the Exchange Act.
  - (z) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act.
- (aa) "Option" means an option to purchase shares of Common Stock granted pursuant to the Plan. For the avoidance of doubt, no Option granted pursuant to the Plan will be an "incentive stock option" within the meaning of Section 422 of the Code.
- (bb) "Option Agreement" means an Award Document evidencing the terms and conditions of an Option grant. Each Option Agreement will be subject to the terms and conditions of the Plan.
- (cc) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.
- (dd) "Other Stock Award" means an award based in whole or in part by reference to the Common Stock which is granted pursuant to the terms and conditions of Section 6(d).
- (ee) "Other Stock Award Document" means an Award Document evidencing the terms and conditions of an Other Stock Award grant. Each Other Stock Award Document will be subject to the terms and conditions of the Plan.
- (ff) "Own," "Owned," "Owner," "Ownership" means a person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.
- (gg) "Participant" means a person to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Stock Award.

- (hh) "Performance Cash Award" means an award of cash granted pursuant to the terms and conditions of Section 6(c)(ii).
- (ii) "Performance Criteria" means the one or more criteria that the Board will select for purposes of establishing the Performance Goals for a Performance Period. The Performance Criteria that will be used to establish such Performance Goals may be based on any one of, or combination of, the following as determined by the Board: (1) profit before tax; (2) billings; (3) revenue; (4) net revenue; (5) earnings (which may include earnings before interest, taxes, depreciation and amortization, or some of them, or and net earnings); (6) operating income; (7) operating margin; (8) operating profit; (9) controllable operating profit, or net operating profit; (10) net profit; (11) gross margin; (12) operating expenses or operating expenses as a percentage of revenue; (13) net income; (14) earnings per share; (15) total stockholder return, calculated either solely with respect to the Company's performance or relative to a benchmark; (16) market share; (17) return on assets or net assets; (18) the Company's stock price; (19) growth in stockholder value relative to a predetermined index; (20) return on equity; (21) return on invested capital; (22) cash flow (including free cash flow or operating cash flows); (23) cash conversion cycle; (24) economic value added; (25) individual confidential business objectives; (26) contract awards or backlog; (27) overhead or other expense reduction; (28) credit rating; (29) strategic plan development and implementation; (30) succession plan development and implementation; (31) improvement in workforce diversity; (32) customer indicators; (33) new product invention or innovation; (34) attainment of research and development milestones; (35) improvements in productivity; (36) bookings; (37) managed revenue; and (38) marketplace fees.
- (jj) "Performance Goals" means, for a Performance Period, the one or more goals established by the Board for the Performance Period based upon the Performance Criteria. Performance Goals may be based on a Company-wide basis or with respect to one or more parts of the Company's business, or with respect to individual Participant performance, as appropriate, and in either absolute terms or relative to the performance of one or more comparable companies or the performance of one or more relevant indices. Performance Goals for financial Performance Criteria may be determined on either a GAAP or non-GAAP basis. Unless specified otherwise by the Board (i) in the Award Document at the time the Award is granted or (ii) in such other document setting forth the Performance Goals at the time the Performance Goals are established, the Board will appropriately make adjustments in the method of calculating the attainment of Performance Goals for a Performance Period as follows: (1) to exclude restructuring and/or other nonrecurring charges; (2) to exclude exchange rate effects; (3) to exclude the effects of changes to generally accepted accounting principles; (4) to exclude the effects of any statutory adjustments to corporate tax rates; (5) to exclude the effects of any "extraordinary items" as determined under generally accepted accounting principles; (6) to exclude the dilutive effects of acquisitions or joint ventures; (7) to assume that any business divested by the Company achieved performance objectives at targeted levels during the balance of a Performance Period following such divestiture; (8) to exclude the effect of any change in the outstanding shares of common stock of the Company by reason of any stock dividend or split, stock repurchase, reorganization, recapitalization, merger, consolidation, spin-off, combination or exchange of shares or other similar corporate change, or any distributions to common stockholders other than regular cash dividends; (9) to exclude the effects of stock based compensation and the

under generally accepted accounting principles; (11) to exclude the goodwill and intangible asset impairment charges that are required to be recorded under generally accepted accounting principles and (12) to exclude the effect of any other unusual, non-recurring gain or loss or other extraordinary item. In addition, the Board retains the discretion to reduce or eliminate the compensation or economic benefit due upon attainment of Performance Goals and to define the manner of calculating the Performance Criteria it selects to use for such Performance Period. Partial achievement of the specified criteria may result in the payment or vesting corresponding to the degree of achievement as specified in the Stock Award Document or the written terms of a Performance Cash Award.

- (kk) "**Performance Period**" means the period of time selected by the Board over which the attainment of one or more Performance Goals will be measured for the purpose of determining a Participant's right to and the payment of a Stock Award or a Performance Cash Award. Performance Periods may be of varying and overlapping duration, at the sole discretion of the Board.
  - (Il) "Performance Stock Award" means a Stock Award granted under the terms and conditions of Section 6(c)(i).
  - (mm) "Plan" means this 2014 Inducement Grant Equity Incentive Plan of The Rubicon Project, Inc.
  - (nn) "Restricted Stock Award" means an award of shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(a).
- (00) "Restricted Stock Award Document" means an Award Document evidencing the terms and conditions of a Restricted Stock Award grant. Each Restricted Stock Award Document will be subject to the terms and conditions of the Plan.
- (pp) "Restricted Stock Unit Award" means a right to receive shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(b).
- (qq) "Restricted Stock Unit Award Document" means an Award Document evidencing the terms and conditions of a Restricted Stock Unit Award grant. Each Restricted Stock Unit Award Document will be subject to the terms and conditions of the Plan.
  - (rr) "Securities Act" means the U.S. Securities Act of 1933, as amended.
- (ss) "Stock Appreciation Right" or "SAR" means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 5.
- (tt) "Stock Appreciation Right Award Document" means an Award Document evidencing the terms and conditions of a Stock Appreciation Right grant. Each Stock Appreciation Right Award Document will be subject to the terms and conditions of the Plan.
- (uu) "Stock Award" means any right to receive Common Stock granted under the Plan, including an Option, a Restricted Stock Award, a Restricted Stock Unit Award, a Stock Appreciation Right, a Performance Stock Award, or any Other Stock Award.

- (vv) "Stock Award Document" means an Award Document evidencing the terms and conditions of a Stock Award grant. Each Stock Award Document will be subject to the terms and conditions of the Plan.
- (ww) "Subsidiary" means, with respect to the Company, (i) any corporation of which more than 50% of the outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than 50%.
- (xx) "Ten Percent Stockholder" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any Affiliate.

END OF DOCUMENT

# THE RUBICON PROJECT, INC.

## 2014 INDUCEMENT GRANT EQUITY INCENTIVE PLAN

# RESTRICTED STOCK UNIT GRANT NOTICE

#### FOR EMPLOYEES

Notice is hereby given of the grant by The Rubicon Project, Inc. (the "Company") to the Participant named below (the "Participant") of a Restricted Stock Unit Award under the Company's 2014 Inducement Grant Equity Incentive Plan (the "Plan"), which is available on the Company intranet (Inside RP) and incorporated herein by reference. This Restricted Stock Unit Award is governed by this Notice (including any special terms and conditions set forth in any appendices attached hereto), and the Plan, and in the event of a conflict between the terms of this Notice and the Plan, the terms of the Plan shall control. By acceptance of the Restricted Stock Unit Award, and also by acceptance through performance of the vesting requirements and the Shares issuable upon vesting, Participant agrees to the terms and conditions set forth in this Notice (including any special terms and conditions set forth in any appendices attached hereto) and the Plan. Capitalized terms used but not defined in this Notice shall have the meanings given to them in the Plan.

The Restricted Stock Unit Award consists of the number of Restricted Stock Units set forth below (the "Restricted Stock Units" or "RSUs"). Each RSU represents the right to receive one share (a "Share") of the Company's Common Stock, par value \$0.00001 (the "Common Stock"), subject to vesting as set forth below and to the terms and conditions of the Plan and this Notice, as follows:

Participant Name:	
Number of Restricted Stock Units:	
Issuance Date:	
<b>Vesting Commencement Date:</b>	
Vesting Schedule:	

For purposes of this Notice, "Vesting Date" means each May 15 and November 15, and a complete calendar month will begin on the first day of each calendar month and end on the last day of that calendar month. Subject to the Notice and any Separate Agreement (as defined below), and subject to any acceleration provisions in the Plan:

- (i) on the first Vesting Date that is on or immediately following the first anniversary of the Vesting Commencement Date (the "First Vesting Date"), there shall vest a number of the RSUs equal to the sum of (A) 25% of the total number of RSUs and (B) a number of RSUs equal to the product of 2.0833% of the total number of RSUs and the number of complete calendar months, if any, elapsed during the period beginning on the first anniversary of the Vesting Commencement Date and ending on the First Vesting Date;
- (ii) on each of the six Vesting Dates next succeeding the First Vesting Date, there shall vest an additional number of RSUs equal to 12.5% of the total number of RSUs, except that the number of RSUs vesting on the last of such six succeeding Vesting Dates will be less than 12.5% of the total number of RSUs if and to the extent that the number of RSUs Vesting on the First

Vesting Date exceeded 25% of the total number of RSUs;

- (iii) except as provided in <u>Section 2</u> below in connection with a termination of Continuous Service without Cause or due to death or Disability, no RSUs will vest before the First Vesting Date, and vesting of RSUs will occur only on Vesting Dates, without any ratable vesting for periods of time between Vesting Dates; and
- (iv) if the application of one of the vesting percentages set forth above results in the vesting of a fractional Share, the number of Shares that shall become vested on such Vesting Date shall be rounded to the nearest whole Share, provided that the number of Shares paid to Participant on the final Vesting Date shall be adjusted as appropriate to compensate for rounding on previous Vesting Dates so that the total number of Shares paid is equal to the total number of Shares subject to the RSUs, as such Shares may be adjusted pursuant to Section 9 of the Plan.

Subject to <u>Section 2</u> below, if Participant ceases to remain in Continuous Service for any or no reason before Participant vests in any of the Restricted Stock Units, all unvested Restricted Stock Units and Participant's right to acquire any Shares of Common Stock hereunder will immediately terminate and be forfeited. However, notwithstanding anything herein to the contrary, the vesting of the Restricted Stock Units shall be subject to any vesting acceleration provisions applicable to the Restricted Stock Units contained in the Plan and/or any employment or service agreement, offer letter, severance agreement, or any other agreement between Participant and the Company or any Affiliate (such agreement, a "Separate Agreement"). Furthermore, under all circumstances, the vesting of Restricted Stock Units shall be subject to the satisfaction of Participant's obligations as set forth in Section 6(b).

The Restricted Stock Unit Award is subject to the terms and conditions, and the representations of Participant, set forth below and including any special terms and conditions set forth in any appendices attached hereto.

## 1. Vesting of RSUs and Payment of Shares.

- (a) <u>Prior to Vesting</u>. Prior to vesting and actual payment on any vested Restricted Stock Unit, such Restricted Stock Unit will represent an unsecured obligation of the Company, for which there is no trust and no obligation other than to make payment as contemplated by this Notice and the Plan. Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Restricted Stock Units, or any Shares deliverable hereunder unless and until such RSUs have vested in the manner set forth in the Vesting Schedule above and the underlying Shares have been issued and recorded on the records of the Company or its transfer agents or registrars. No adjustment shall be made for any dividends (ordinary or extraordinary, whether cash, securities, or other property) or distributions or other rights for which the record date is prior to the date Shares are issued, except as provided in Section 9 of the Plan.
- (b) <u>Vesting</u>. Each Restricted Stock Unit represents the right to receive payment on the date it vests in the form of one Share. Subject to <u>Section 3</u> and the next paragraph, any Restricted Stock Units that vest will be paid to Participant in whole Shares as soon as practicable after vesting, but in each such case within the period ending no later than the fifteenth (15<sup>th</sup>) day of the third (3<sup>rd</sup>) month following the end of the calendar year, or if later, the end of the Company's tax year, in either case that includes the vesting date. In no event will Participant be permitted, directly or indirectly, to specify the taxable year of payment of any Restricted Stock Units payable under this Notice. Any distribution or delivery of Shares to be made to Participant will, if Participant is then deceased, be made to Participant's designated beneficiary, or if no beneficiary survives Participant, the administrator or executor of Participant's estate. Any such transfere must furnish the Company with written notice of his or her status as transferee and evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer. After RSUs have vested in the manner set forth in the Vesting Schedule above and the underlying Shares have been issued and recorded on the records of the Company or its transfer agents or registrars, Participant will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares.
- (c) 409A. Notwithstanding anything in the Plan, this Notice, or any Separate Agreement to the contrary, if the vesting of the balance, or some lesser portion of the balance, of the Restricted Stock Units is accelerated in connection with the termination of Participant's Continuous Service (provided that such termination is a "separation from service" within the meaning of Section 409A, as determined by the Company), other than due to death, and if (x) Participant is a "specified employee" within the meaning of Section 409A at the time of the termination of Participant's Service and (y) the payment of such accelerated Restricted Stock Units will result in the imposition of additional tax under Section 409A if paid to Participant on or within the six (6) month period following the termination of Participant's Service, then the payment of such accelerated Restricted Stock Units will not be made until the date that is six (6) months and one (1) day following the date of termination of Participant's Continuous Service, unless Participant dies following the date his or her Continuous Service terminates, in which case, the RSUs will be paid in Shares to Participant's estate as soon as practicable following his or her death. It is the intent of this Notice that the grant of Restricted Stock Units and any Shares issuable upon vesting of the Restricted Stock Units be exempt from the requirements of Section 409A to the greatest extent provided under the regulations promulgated so that none of the Restricted Stock Units or Shares issuable upon vesting of RSUs will be subject to the additional tax imposed under Section 409A, and any ambiguities herein will be interpreted to so comply. To the extent that any Restricted Stock Units or any Shares issuable under the terms of any Restricted Stock Units are determined to be subject to the requirements of Section 409A, it is the intent of this Notice that the award comply with Section 409A, and any ambiguities will be interpreted to so comply. For purposes of this Notice, "Section

Revenue Service guidance thereunder, as each may be amended from time to time.

2. Forfeiture Upon Termination of Continuous Service. Except as otherwise provided in the Vesting Schedule set forth above in this Notice or in a Separate Agreement, but notwithstanding any contrary provision of this Notice, if Participant ceases to remain in Continuous Service at any time for any reason other than (i) a termination of Continuous Service by the Company without Cause on or after the First Vesting Date, or (ii) Participant's death or Disability on or after the First Vesting Date, the then-unvested Restricted Stock Units will thereupon terminate and be forfeited at no cost to the Company and Participant will have no further rights with respect to such forfeited Restricted Stock Units or any underlying Shares. Upon a termination of Continuous Service by the Company without Cause on or after the First Vesting Date, or due to Participant's death or Disability on or after the First Vesting Date, a number of additional Restricted Stock Units shall become vested as of the date of such termination of Continuous Service equal to the product obtained by multiplying the number of RSUs scheduled to vest on the scheduled Vesting Date next succeeding the date of termination of Continuous Service and a fraction, the numerator of which is the number of full months from the Vesting Date immediately preceding the date of termination of Continuous Service, and the denominator of which is the number of full months from the Vesting Date immediately preceding the date of termination of Continuous Service to the scheduled Vesting Date next following the date of termination of Continuous Service to the scheduled Vesting Date next following the date of termination of Continuous Service. For these purposes, a full month means the period from the date of one calendar month to the same date the next calendar month (e.g., from May 15 to June 15), or the last day of the next calendar month if the date is the 29th, 30th, or 31st and the next calendar month does not have at least 29, 30 or 31 days, as the case may be. Any Restr

# 3. Tax Consequences, Withholding, and Liability.

(a) Participant understands that Participant may suffer adverse tax consequences as a result of the grant or vesting of the Restricted Stock Units and issuance and/or disposition of the Shares. Participant understands that the actual tax consequences associated with the Restricted Stock Units and Shares are complicated and depend, in part, on Participant's specific situation and may also depend on the resolution of currently uncertain tax law and other variables not within the control of the Company. THEREFORE, PARTICIPANT SHOULD SEEK INDEPENDENT ADVICE REGARDING THE APPLICABLE PROVISIONS OF THE FEDERAL TAX LAW AND THE TAX LAWS OF ANY MUNICIPALITY, STATE OR NON-U.S. JURISDICTION TO WHICH PARTICIPANT IS SUBJECT. By accepting (through performance) the Restricted Stock Units and any Shares, Participant acknowledges and agrees that Participant has either consulted with a competent tax advisor independent of the Company to obtain tax advice concerning the RSUs and Shares in light of Participant's specific situation or has had the opportunity to consult with such a tax advisor and has chosen not to do so. Neither the Company nor any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents, any written or oral advice or representation regarding the U.S. federal, state, local or non-U.S. tax consequences of the receipt, ownership and vesting of the Restricted Stock Units, the other transactions contemplated by this Notice, or the value of the Company or the RSUs or Shares at any time. With respect to such matters, Participant relies solely on P

(b) Participant (and not the Company) shall be responsible for Participant's own tax liability that may arise as a result of the receipt, ownership and vesting of the Restricted Stock Units, the

issuance of Shares pursuant to the Restricted Stock Units, or the other transactions contemplated by this Notice. Pursuant to such procedures as the Plan administrator may specify from time to time, the Company shall satisfy its obligations to pay withholding taxes or other tax deposits in connection with the receipt, ownership and/or vesting of the Restricted Stock Units, the issuance of Shares pursuant to the Restricted Stock Units, or the other transactions contemplated by this Notice in accordance with applicable law or regulation (the "Tax Obligations"). If amounts paid by the Company in respect of Tax Obligations are less than Participant's tax obligations, Participant is solely responsible for any additional taxes due. If amounts paid by the Company in respect of Tax Obligations exceed Participant's tax obligations, Participant's sole recourse will be against the relevant taxing authorities, and the Company and its Affiliates will have no obligation to issue additional Shares or pay cash to Participant in respect thereof. Participant is responsible for determining Participant's actual income tax liabilities and making appropriate payments to the relevant taxing authorities to fulfill Participant's tax obligations and avoid interest and penalties.

(c) Payment by the Company or its Affiliate of the Tax Obligations will result in a commensurate obligation of Participant to pay, or cause to be paid, to the Company or its Affiliate, in accordance with Section 8(g) of the Plan, the amount of Tax Obligations so paid, and the Company shall not be required to issue any of the Shares or any interest in the Shares unless and until Participant has satisfied this obligation. To the extent determined appropriate by the Company in its discretion, it shall have the right (but not the obligation) to cause Participant to satisfy any or all Tax Obligations by withholding and retaining Shares otherwise deliverable to Participant having an aggregate Fair Market Value equal to the amount of such Tax Obligations. If, at the time Shares are to be issued, those Shares are not freely tradeable on a national securities exchange or market system (and for this purpose, a blackout pursuant to the Company's insider trading policy will not be considered to render the Shares not freely tradeable), Participant may in Participant's sole discretion satisfy the Tax Obligations by electing to have the Company withhold and retain such number of Shares otherwise deliverable to Participant, and/or by surrendering such number of Shares already delivered to Participant, having an aggregate Fair Market Value equal to the amount of such Tax Obligations. In order to satisfy the Tax Obligations, the Company will not withhold the amount of such Tax Obligations from Participant's paycheck(s) and/or any other amounts payable to Participant unless the net proceeds from any automatic sale of certain shares of the Restricted Stock as set forth in Section 3(d) below are not sufficient to satisfy such Tax Obligations in their entirety.

(d) In the event that (i) Participant is not subject to the requirements of Section 16 of the Securities Exchange Act of 1934 on a date that the risk of forfeiture to the Company as described in this Notice lapses with respect to some or all of the Restricted Stock Units ("Lapse Date") and (ii) Participant incurs a tax liability on such Lapse Date as a result of such lapse, then the Applicable Percentage (as defined below) of the Shares issuable pursuant to the Restricted Stock Units with respect to which the risk of forfeiture shall have lapsed on the Lapse Date, shall be sold within an administratively reasonable period of time on or after the Lapse Date by a broker selected or approved by the Company at such fees and pursuant to such rules and process as the Company may reasonably approve. Participant will bear the brokerage fees and other costs associated with sales and related transmission of funds. The net proceeds from such sale shall be remitted to the relevant tax authorities as determined by the Company for Participant's benefit in the amounts directed by the Company, or paid to the Company in reimbursement of any Tax Obligations paid by the Company, and any remaining net proceeds shall be delivered to Participant or a brokerage account maintained for Participant. For these purposes the "Applicable Percentage" means, in the Company's sole discretion, an amount reasonably expected to be required to satisfy any or all Tax Obligations and selling expenses. Participant shall have no right to affect or influence any adjustments that the Company may elect to make to the Applicable Percentage for this purpose. There is no assurance that the price at which Shares sold pursuant to this

Section 3(d) will equal the value at which Shares vesting on the Lapse Date are taxed.

4. No Guarantee of Continued Service. THE VESTING OF THE RESTRICTED STOCK UNITS PURSUANT TO THE VESTING SCHEDULE APPLICABLE THERETO IS EARNED ONLY BY CONTINUOUS SERVICE AT THE WILL OF THE COMPANY (OR THE AFFILIATE OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED A RESTRICTED STOCK UNIT AWARD OR ACQUIRING SHARES UPON VESTING OF RESTRICTED STOCK UNITS. THIS NOTICE, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE APPLICABLE TO RESTRICTED STOCK UNITS DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT TO PROVIDE SERVICE FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE AFFILIATE OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S CONTINUOUS SERVICE AT ANY TIME, FOR ANY REASON OR NO REASON, WITH OR WITHOUT NOTICE, AND WITH OR WITHOUT CAUSE.

#### 5. Participant Representations.

- (a) Participant acknowledges that (i) Participant was and is free to use professional advisors of Participant's choice in connection with this Notice and grant of the Restricted Stock Units, that Participant understands this Notice and the meaning and consequences of receiving grants of RSUs and Shares issued upon vesting of RSUs; (ii) Participant has reviewed and understands this Notice and the Plan; (iii) receipt of the RSUs and any Shares issued upon vesting of the RSUs is voluntary and Participant is accepting the RSUs and any Shares issued freely and without coercion or duress; and (iv) Participant has not received and is not relying, and will not rely, upon any advice, representations or assurances made by or on behalf of the Company or any of its Affiliates or any employee of or counsel to the Company or any of its Affiliates regarding any tax or other effects or implications of the RSUs or Shares or other matters contemplated by this award of Restricted Stock Units.
- (b) Participant is aware of the Company's business affairs and financial condition and understands that an investment in the Shares involves a high degree of risk. Participant has not received and is not relying, and will not rely, upon any advice, representations or assurances made by or on behalf of the Company or any of its Affiliates or any employee of or counsel to the Company or any of its Affiliates regarding the Company's prospects or the value of the RSUs or Shares.

#### 6. Additional Conditions to Issuance of Stock.

(a) <u>Legal and Regulatory Compliance</u>. The issuance of Shares upon or after vesting of the Restricted Stock Units shall be subject to compliance with all applicable requirements of federal, state or foreign law with respect to such securities. If at any time the Company determines, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary as a condition to the issuance of Shares to Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. If the Company determines that the issuance of any Shares will violate federal securities laws or other applicable laws or regulations or the requirements of any exchange or market system upon which the Shares are listed, the Company may defer issuance until the earliest date at which the Company reasonably anticipates that the issuance of Shares will no longer cause such violation. Accordingly, Participant may not be able to receive Shares

when desired even though the Restricted Stock Units have vested. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority, but the inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance of any Shares shall relieve the Company of any liability in respect of the failure to issue such Shares as to which such requisite authority shall not have been obtained. As a condition to the issuance of Shares, the Company may require Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company. Without limiting the foregoing, if at the time of vesting of any Restricted Stock Units, there is not in effect under the Securities Act of 1933, as amended (the "Securities Act"), a registration statement covering the Shares to be issued, and available for delivery a prospectus meeting the requirements of Section 10(a)(3) of the Securities Act, Participant shall, if required by the Company, as a condition to vesting and issuance of the Shares, make appropriate representations in a form satisfactory to the Company to support issuance of the Shares in compliance with applicable laws and regulations, including to the effect that such Shares will not be sold other than (A) pursuant to an effective registration statement under the Securities Act, or an applicable exemption from the registration requirements of such Act; (B) in compliance with all applicable state securities laws and regulations; and (C) in compliance with all terms and conditions of the Plan, this Notice, and any other written agreement between Participant and the Company or any of its Affiliates.

(b) Obligations to the Company. As a condition to receipt and vesting of any Restricted Stock Units and issuance of Shares as a result of vesting, Participant must enter into the Company's Intellectual Property Assignment and Confidential Information Agreement, or a similar or successor agreement for the protection of the Company's intellectual property and confidential information, in form specified by the Company (the "Proprietary Interests Agreement"), if Participant has not already done so, and Participant's acceptance of Restricted Stock Units and any Shares will constitute Participant's agreement to the Proprietary Interests Agreement. If Participant breaches in any material respect the Proprietary Interests Agreement or any other contract between Participant and the Company, or Participant's common law duty of confidentiality or trade secret protection, or any Company policy prohibiting misappropriation of property or any illegal or fraudulent acts, the Company may suspend any vesting of any Restricted Stock Units or issuance of any Shares pending Participant's cure of such breach, and if such breach cannot be cured or is not cured to the Company's reasonable satisfaction within such time not less than twenty (20) days as the Company may specify, the Company may terminate any Restricted Stock Units for which Shares have not been issued and will have no obligation to issue any Shares in respect of any such terminated Restricted Stock Units or to provide any consideration to Participant in respect thereof.

## 7. Handling of Shares; Restrictive Legends and Stop-Transfer Orders.

- (a) <u>Book Entries</u>. The Company will cause the Shares to be recorded in book entry or other electronic form and reflected in records maintained by or for the Company.
- (b) <u>Legends</u>. Each data base entry representing any Shares may be endorsed with legends substantially as set forth below, as well as such other legends as the Company may deem appropriate to implement the provisions of this Notice or comply with applicable laws and regulations and Company policies:

THE SHARES REPRESENTED HEREBY ARE SUBJECT TO RESTRICTIONS ON TRANSFER FOR A PERIOD OF TIME FOLLOWING THE EFFECTIVE DATE OF

ANY UNDERWRITTEN PUBLIC OFFERING OF THE COMPANY'S SECURITIES SET FORTH IN AN AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES AND MAY NOT BE SOLD OR OTHERWISE DISPOSED OF BY THE HOLDER PRIOR TO THE EXPIRATION OF SUCH PERIOD WITHOUT THE CONSENT OF THE COMPANY OR THE MANAGING UNDERWRITER.

- (c) <u>Stop-Transfer Notices</u>. In order to ensure compliance with the restrictions referred to herein and Company policies, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (d) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Notice or any other agreement to which the Shares are subject or any laws governing the Shares or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferree to whom such Shares shall have been so transferred.
- 8. Restrictions on Transfer. Except as otherwise expressly provided in this Notice, the Restricted Stock Units will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of the Restricted Stock Units, or upon any attempted sale under any execution, attachment or similar process, the affected RSUs will become null and void. The Company may impose such restrictions, conditions or limitations as it determines appropriate as to the timing and manner of any resales by the Participant or other subsequent transfers by the Participant of any vested Shares, including without limitation (a) restrictions under an insider trading policy, (b) restrictions designed to delay and/or coordinate the timing and manner of sales by Participant and other holders, and (c) restrictions as to the use of a specified brokerage firm for such resales or other transfers.
- 9. Lock-Up. In connection with any underwritten public offering by the Company of its equity securities pursuant to a registration statement filed under the Securities Act, upon the request of the Company or the underwriters managing such offering, during the Lock-up Period (as defined below) Participant shall not, without the prior written consent of the Company or its underwriters, directly or indirectly sell, make any short sale of, loan, hypothecate, pledge, offer, grant or sell any option or other contract for the purchase of, purchase any option or other contract for the sale of, enter into any swap, hedging or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of, or otherwise dispose of or transfer, or agree to engage in any of the foregoing transactions with respect to, any Shares or other securities into which the Shares may be converted or that are issued in respect of the Shares (other than those included in the registration). For this purpose, the "Lock-up Period" means such period of time after the effective date of the registration as is requested by the Company or the underwriters; provided that such period shall not exceed 180 days (or such additional period as may reasonably be requested by the Company or such underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports or (ii) analyst recommendations and opinions, including (without limitation) the restrictions set forth in Rule 2711(f)(4) of the National Association of Securities Dealers and Rule 472(f)(4) of the New York Stock Exchange, as amended, or any similar successor rules). The Company's underwriters shall be beneficiaries of the provision set forth in this Section 9, and Participant shall execute and deliver such agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In addition, if requested by the Compan

or the underwriters of Common Stock (or other securities) of the Company, Participant shall provide, within ten (10) days of such request, such information as may be required or reasonably requested by the Company or the underwriters in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Securities Act. The obligations described in this Section 9 shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a Commission Rule 145 transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions with respect to the Shares (or other securities) subject to the foregoing restriction until the end of said one hundred and eighty (180) day (or other) period. Participant agrees, and will cause any transferee to agree, that any transferee of the award of Restricted Stock Units shall be bound by this Section 9.

# 10. Additional Agreements.

(a) <u>Electronic Delivery</u>. The Company may, in its sole discretion, decide to deliver any documents related to Restricted Stock Units or Shares by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to administration of this Notice, the RSUs and the Shares through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.

(b) Personal Information. To facilitate the administration of the Plan and any successor plan and the terms of this Notice, it may be necessary for the Company (or its payroll administrators) to collect, hold and process certain personal information about Participant, including, but not limited to, Participant's name, home address and telephone number, date of birth, social insurance number or other identification number, salary, nationality, job title, any shares of Company Common Stock owned, relationship to the Company, details of all awards issued under the Plan or any predecessor or successor plan or any other entitlement to shares of Company Common Stock awarded, canceled, exercised, vested, unvested or outstanding in Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("Data") and to transfer this Data to certain third parties such as transfer agents, stock plan administrators, and brokers with whom Participant or the Company may elect to deposit any Shares. Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of Participant's Data for the exclusive purpose of implementing, administering and managing Participant's participation in the Plan and any predecessor and successor plan. Participant understands that Data will be transferred to the Company's transfer agent, broker, administrative agents or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan and any predecessor and successor plan. Participant understands that the recipients of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than Participant's country. Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker, administrative agents, and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan or any predecessor or successor plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participant's participation in the Plan or any predecessor or successor plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage Participant's participation in the Plan or any predecessor or successor plan. The

Participant understands if he or she resides outside the United States, he or she may, at any time, view Data, request additional information about the storage and processing of Data, require any necessary amendments to Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources representative. Further, Participant understands that he or she is providing the consents herein on a purely voluntary basis. If Participant does not consent, or if Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Company will not be adversely affected; the only consequence of refusing or withdrawing Participant's consent is that the Company would not be able to grant Restricted Stock Units or other equity awards or administer or maintain such awards. Therefore, Participant understands that refusing or withdrawing his or her consent may affect Participant's ability to participate in the Plan or any successor plan. For more information on the consequences of Participant's refusal to consent or withdrawal of consent, Participant understands that he or she may contact his or her local human resources representative. Finally, upon request of the Company, Participant agrees to provide an executed data privacy consent form to the Company (or any other agreements or consents that may be required by the Company) that the Company may deem necessary to obtain under the data privacy laws in Participant's country, either now or in the future. Participant understands that he or she will not be able to participate in the Plan if he or she fails to execute any such consent or agreement.

- (c) <u>Proprietary Information</u>. Participant agrees that all financial and other information relating to the Company furnished to Participant constitutes "Proprietary Information" that is the property of the Company. Participant shall hold in confidence and not disclose or, except within the scope of Participant's Continuous Service, use any Proprietary Information. Participant shall not be obligated under this paragraph with respect to information Participant can document is or becomes readily publicly available without restriction through no fault of Participant. Upon termination of Participant's Continuous Service, Participant shall promptly return to Company all items containing or embodying Proprietary Information (including all copies). This paragraph supplements, but does not limit, any other agreement between Participant and the Company, or any applicable law, related to protection, ownership, or use of the Company's information or property.
- (d) <u>Consideration</u>. The Restricted Stock Units and Shares are issued in consideration of services provided by Participant and/or other benefit to the corporation within the meaning of Section 152 of the General Corporation Law of the State of Delaware; Participant is not required to make any cash payment to the Company in respect of issuance of Restricted Stock Units or Shares.

# 11. General.

- (a) No Waiver; Remedies. Either party's failure to enforce any provision of this Notice shall not in any way be construed as a waiver of any such provision, or prevent that party from thereafter enforcing such provision and each and every other provision of this Notice. The rights granted both parties herein are cumulative and shall not constitute a waiver of either party's right to assert all other legal remedies available to it under the circumstances.
- (b) <u>Successors and Assigns</u>. The terms of this Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth, the terms of this Notice shall be binding upon Participant and Participant's heirs, executors, administrators, successors and assigns. The rights and obligations of Participant under this Notice may only be assigned with the prior written consent of the Company.

- (c) <u>Notices</u>. Any notice hereunder shall be in writing (which shall include electronic transmission) and shall be deemed received (i) the business day following electronic verification of receipt if sent electronically, (ii) upon personal delivery to the party to whom the notice is directed, (iii) the business day following deposit with a reputable overnight courier, or (iv) five days after deposit in the U.S. mail, First Class with postage prepaid. Notice shall be addressed to the Company at its principal executive office and to Participant at the address that he or she most recently provided to the Company. Participant agrees that it is Participant's responsibility to notify the Company of any changes to his or her mailing address so that Participant may receive any shareholder information to be delivered by regular mail.
- (d) Interpretation. Headings herein are for convenience of reference only, do not constitute a part of this Notice, and will not affect the meaning or interpretation of this Notice. References herein to Sections are references to the referenced Section hereof, unless otherwise specified. The Board or its Committee will have the power to interpret the Plan and this Notice and to adopt such rules for the administration, interpretation and application of the Plan and this Notice as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Restricted Stock Units have vested). All actions taken and all interpretations and determinations made by the Board or its Committee in good faith will be final and binding upon Participant, the Company and all other interested persons. Neither the Board or its Committee nor any person acting on behalf of the Board or its Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Notice.
- (e) <u>Modifications to Notice</u>. Modifications to this Notice can be made only in an express written contract executed by a duly authorized officer of the Company and shall not require the consent of the Participant unless such modification would materially adversely affect the rights of the Participant hereunder. Notwithstanding anything to the contrary in the Plan or this Notice, the Company reserves the right to revise this Notice as it deems necessary or advisable, in its sole discretion and without the consent of Participant, to comply with Section 409A or to otherwise avoid imposition of any additional tax or income recognition under Section 409A in connection to this award of Restricted Stock Units.
- (f) Governing Law; Severability. This Notice is governed by the internal substantive laws, but not the choice of law rules, of Delaware. If any provision of this Notice becomes or is declared by a court or arbitrator having jurisdiction over a dispute hereunder to be illegal, unenforceable or void, such provision shall be amended to the extent necessary to conform to applicable law so as to be valid and enforceable and to achieve, to the extent possible, the economic, business and other purposes of such illegal, unenforceable, or void provision or, if such provision cannot be so amended without materially altering the intention of the parties, then such provision shall deleted from this Notice and the remainder of this Notice shall continue in full force and effect.
- (g) Entire Agreement. The Plan and this Notice, along with any Separate Agreement (to the extent applicable), form a contract and constitute the entire understanding between Participant and the Company with respect to the RSUs and the Shares issuable upon vesting of the RSUs and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect thereto.
- (h) Appendix. The award of RSUs shall be subject to any additional terms and conditions for Non-U.S. Employees set forth in Appendix A attached hereto ("Appendix A") and any special terms and conditions for Participant's country set forth in Appendix B attached hereto ("Appendix B"). Moreover, if Participant relocates to one of the countries included in Appendix B, the

special terms and conditions for such country will apply to Participant to the extent the Company determines that the application of such terms and conditions is necessary or advisable for legal or administrative reasons. Appendix A and Appendix B constitute part of the Notice.		
(i) <u>Imposition of Other Requirements</u> . The Company reserves the right to impose other requirements on Participant's participation in the Plan, on the RSUs and on any Shares acquired under the Plan, to the extent the Company determines it is necessary or advisable for legal or administrative reasons, and to require Participant to sign any additional agreements or undertakings that may be necessary to accomplish the foregoing.		
Dated:	THE RUBICON PROJECT, INC.	
	Ву:	
	Name:	
	Title:	

## APPENDIX A

# ADDITIONAL TERMS AND CONDITIONS OF RESTRICTED STOCK UNIT GRANT FOR NON-U.S. EMPLOYEES

- 1. Terms of Plan Participation for Non-U.S. Participants. Participant understands that this Appendix A contains additional terms and conditions that, together with the Plan and the Notice, govern Participant's participation in the Plan if Participant is working or resident in a country other than the United States. Participant further understands that Participant's participation in the Plan also will be subject to any terms and conditions for Participant's country set forth in Appendix B attached hereto. Capitalized terms used but not defined in this Appendix A shall have the same meanings assigned to them in the Plan and/or Notice.
  - 2. Tax Consequences, Withholding, and Liability. The following provision supplements Section 3 of the Notice:

By accepting (through performance) the RSUs and any Shares, Participant authorizes the Company and/or the Subsidiary or Affiliate employing or retaining Participant (the "Employer"), or their respective agents, at the Company's discretion, to satisfy the obligations with regard to all Tax Obligations by one or a combination of the methods set forth in Section 8(g) of the Plan. If Participant is or becomes subject to Tax Obligations in more than one jurisdiction, Participant acknowledges that the Company and/or the Employer (or former employer, as applicable) may be required to withhold or account for Tax Obligations in more than one jurisdiction.

If the Tax Obligations are satisfied by withholding in Shares, for tax purposes, Participant is deemed to have been issued the full number of Shares subject to the vested RSUs, notwithstanding that a number of the Shares are held back solely for the purpose of paying the Tax Obligations.

- 3. Nature of Grant. By accepting (through performance) the RSUs and any Shares, Participant acknowledges, understands and agrees that:
- (a) the grant of the RSUs is voluntary and occasional and does not create any contractual or other right to receive future grants of RSUs, or benefits in lieu of RSUs, even if RSUs have been granted in the past;
  - (b) all decisions with respect to future RSU or other grants, if any, will be at the sole discretion of the Company;
- (c) the RSUs and any Shares acquired under the Plan, and the income and value of the same, are not intended to replace any pension rights or compensation;
- (d) the RSUs and any Shares acquired under the Plan, and the income and value of the same, are not part of Participant's normal or expected compensation for any purposes including, but not limited to, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement benefits or payments or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or the Employer or any Affiliate;

- (e) the future value of the Shares underlying the RSUs is unknown, indeterminable, and cannot be predicted with certainty;
- (f) if the underlying Shares do not increase in value, the RSUs will have no value;
- (g) for purposes of the RSUs, Participant's Continuous Service will be considered terminated as of the date Participant is no longer actively providing services to the Company or the Employer (regardless of the reason for such termination and whether or not later found to be invalid or in breach of employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment or service agreement, if any), and unless otherwise expressly provided in this Notice or determined by the Company, (i) Participant's right to vest in the RSUs under the Plan, if any, will terminate as of such date and will not be extended by any notice period (e.g., Participant's period of service would not include any contractual notice period or any period of "garden leave" or similar period mandated under employment laws in the jurisdiction where Participant is employed or the terms of Participant's employment agreement, if any); the Board or Committee shall have the exclusive discretion to determine when Participant is no longer actively providing services for purposes of the RSUs (including whether Participant may still be considered to be providing services while on a leave of absence);
- (h) no claim or entitlement to compensation or damages shall arise from forfeiture of the RSUs resulting from the termination of Participant's Continuous Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment or service agreement, if any), and in consideration of the grant of the RSUs to which Participant is otherwise not entitled, Participant irrevocably agrees never to institute any claim against the Company, the Employer or any Affiliate, waives his or her ability, if any, to bring any such claim, and releases the Company, the Employer and any Affiliate from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, Participant shall be deemed irrevocably to have agreed not to pursue such claim and agrees to execute any and all documents necessary to request dismissal or withdrawal of such claim;
- (i) unless otherwise provided in the Plan or by the Company in its discretion, the RSUs and the benefits evidenced by the Notice do not create any entitlement to have the RSUs or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the Shares; and
- (j) neither the Company, the Employer nor any Affiliate shall be liable for any foreign exchange rate fluctuation between Participant's local currency and the U.S. dollar that may affect the value of the RSUs or of any amounts due to Participant pursuant to the vesting of the RSUs or the subsequent sale of any Shares acquired upon vesting.
- 4. <u>Venue</u>. For purposes of litigating any dispute that arises under the Notice, the parties hereby submit to and consent to the jurisdiction of the State of California, and agree that such litigation will be conducted in the courts of Los Angeles County, California, or the federal courts for the United States for the Central District of California, and no other courts, where this award of RSUs is made and/or to be performed.
- 5. <u>Insider Trading</u>. By participating in the Plan, Participant agrees to comply with the Company's policy on insider trading (to the extent that it is applicable to Participant). Further, Participant acknowledges that Participant's country of residence may also have laws or regulations governing insider

trading and that such law or regulations may impose additional restrictions on Participant's ability to participate in the Plan (e.g., acquiring or selling Shares) and that Participant is solely responsible for complying with such laws or regulations.

6. <u>Language</u>. If the Notice or any other document related to the Plan has been translated into a language other than English and the meaning of the translated version is different than the English version, the English version will control.

### APPENDIX B

## COUNTRY-SPECIFIC PROVISIONS FOR NON-U.S. EMPLOYEES

### Terms and Conditions

This Appendix B includes additional terms and conditions that govern the RSUs granted to Participant under the Plan if Participant works or resides in one of the countries listed below. If Participant is a citizen or resident of a country other than the one in which Participant currently is working (or if Participant is considered as such for local law purposes), or if Participant transfers employment or residence to another country after RSUs have been granted to Participant under the Plan, the Company, in its discretion, will determine the extent to which the terms and conditions herein will be applicable to Participant.

Capitalized terms used but not otherwise defined herein shall have the meaning given to such terms in the Plan, the Notice or Appendix A.

### **Notifications**

This Appendix B also includes information regarding securities laws, exchange controls and certain other issues of which Participant should be aware with respect to his or her participation in the Plan. The information is based on the securities, exchange control and other laws in effect in the respective countries as of November 2014. Such laws are often complex and change frequently. As a result, the Company recommends that Participant not rely on the information in this Appendix B as the only source of information relating to the consequences of his or her participation in the Plan because the information included herein may be out of date at the time that Participant acquires Shares under the Plan or subsequently sells such Shares.

In addition, the information contained herein is general in nature and may not apply to Participant's particular situation and the Company is not in a position to assure Participant of any particular result. Accordingly, Participant is advised to seek appropriate professional advice as to how the relevant laws in Participant's country may apply to his or her individual situation.

Finally, if Participant is a citizen or resident of a country other than the one in which Participant currently is working or residing (or if Participant is considered as such for local law purposes), or if Participant transfers employment or residence to another country after RSUs have been granted to Participant under the Plan, the information contained herein may not be applicable to Participant in the same manner.

### **AUSTRALIA**

## Notifications

Securities Law Information. If Shares are acquired under the Plan and subsequently offered for sale to a person or entity resident in Australia, such offer may be subject to disclosure requirements under Australian law. Participants should obtain legal advice regarding any applicable disclosure requirements prior to making any such offer.

### **BRAZIL**

### Terms and Conditions

Compliance with Law. By participating in the Plan, Participant agrees to comply with all applicable Brazilian laws and to pay any and all applicable taxes associated with the acquisition and sale of Shares acquired under the Plan, or the receipt of any dividends in the future.

### **Notifications**

**Exchange Control Information**. Participants who are residents or domiciled in Brazil must submit a declaration of assets and rights held outside of Brazil, including Shares acquired under the Plan, to the Central Bank if the aggregate value of such assets and rights is at least US\$100,000. Participants should consult their personal legal advisors for further details regarding this requirement.

### **CANADA**

## Terms and Conditions

Labor Law Acknowledgement. This provision replaces Section 3(h) of Appendix A and supplements Section 2 of the Notice:

for purposes of the RSUs, Participant's Continuous Service will be considered terminated as of the earlier of: (i) the date on which Participant's employment with the Company and/or the Employer is terminated; (ii) the date on which Participant receives a written notice of termination of Continuous Service regardless of any notice period or period of pay in lieu of such notice required under any employment laws in Participant's country (including, without limitation, statutory law, regulatory law, and/or common law), even if such law is otherwise applicable to Participant's benefits from the Company and/or the Employer; or (iii) the date on which Participant is no longer actively providing services to the Company and/or the Employer (regardless of the reason for such termination and regardless of whether it is later found to be invalid), and unless otherwise expressly provided in this Notice or determined by the Company, Participant's right to vest in the RSUs under the Plan, if any, will terminate as of such date; the Board or Committee shall have the exclusive discretion to determine when Participant is no longer actively providing services for purposes of the RSUs (including whether Participant may still be considered to be providing services while on a leave of absence);

### **Notifications**

**Securities Law Information.** Shares acquired under the Plan may result in Canadian securities laws issues if such Shares are sold through a broker other than the designated broker or if the sale does not take place through the facilities of a stock exchange outside Canada on which the Shares are listed (*i.e.*, the New York Stock Exchange).

**Tax Reporting Obligation**. Foreign property (including Shares acquired under the Plan and possibly the RSUs) must be reported on Form T1135 (Foreign Income Verification Statement) if the total value of foreign property exceeds C\$100,000 at any time during the year. Participants should consult their personal tax advisors for further details regarding this requirement.

### **FRANCE**

## Terms and Conditions

**Language Consent.** By participating in the Plan, Participant confirms having read and understood the documents relating to the RSUs and his or her participation in the Plan (*i.e.*, the Plan and this Notice), which were provided to Participant in the English language. Participant accepts the terms of these documents accordingly.

Consentement Relatif à la Langue Utilisée. En participant au Plan, le Participant confirme avoir lu et compris les documents relatifs aux RSUs et à sa participation au Plan (à savoir, le Plan et le présent Avis) qui lui ont été communiqués en langue anglaise. Le Participant accepte les termes de ces documents en connaissance de cause.

### Notifications

**Foreign Asset/Account Reporting Information.** Participants in France must declare any foreign bank investment, or brokerage account opened, used or closed during the fiscal year to the French tax authorities when filing their annual tax returns. Participants should consult their personal tax advisors for details regarding this requirement.

## **GERMANY**

## Notifications

**Exchange Control Information**. Cross-border payments in excess of €12,500 in connection with the acquisition or sale of securities (*e.g.*, transfer of proceeds from the sale of Shares into Germany) must be reported electronically to the German Federal Bank. The online filing portal may be accessed at the website of the German Federal Bank. Participants should consult their personal tax advisors for details regarding this requirement.

### **ITALY**

### Terms and Conditions

**Data Privacy**. This provision replaces in its entirety <u>Section 10(b)</u> of the Notice:

Participant understands that the Company may hold certain personal information about Participant, including Participant's name, home address and telephone number, date of birth, social insurance number or other identification number, salary, nationality, job title, any shares of stock or directorships Participant holds in the Company, details of the Plan or any other entitlement to shares of stock awarded, canceled, exercised, vested, unvested or outstanding in Participant's favor ("Data"), for the exclusive purpose of implementing, administering and managing Participant's participation in the Plan.

Participant also understands that providing the Company with the Data is necessary for the performance of the Plan and that Participant's refusal to provide such Data would make it impossible for the Company to perform its contractual obligations and may affect Participant's ability to participate in the Plan. The Controller of personal data processing is The Rubicon Project, Inc., with

registered offices at 12181 Bluff Creek Drive, Playa Vista, CA 90094 U.S.A., and, pursuant to D.lgs 196/2003, its representative in Italy is The Rubicon Project S.r.l. with registered offices at Corsa Giaccomo Matteotti 7, CAP 20121 Milano, Italy.

Participant understands that Participant's Data will not be publicized, but it may be transferred to Morgan Stanley Smith Barney, Equity Administration Solutions, Inc., their respective affiliates and other financial institutions or brokers involved in the management and administration of the Plan. Participant further understands that the Company and/or its Affiliates will transfer Data amongst themselves as necessary for the purpose of implementation, administration and management of Participant's participation in the Plan, and that the Affiliates may each further transfer Data to third parties assisting the Company in the implementation, administration and management of the Plan, including any requisite transfer to Morgan Stanley Smith Barney, Equity Administration Solutions, Inc., their respective affiliates, or another third party with whom Participant may elect to deposit any Shares acquired under the Plan. Such recipients may receive, possess, use, retain and transfer the Data in electronic or other form, for the purposes of implementing, administering and managing Participant's participation in the Plan. Participant understands that these recipients may be located in the European Economic Area, or elsewhere, such as the U.S. Should the Company exercise its discretion in suspending all necessary legal obligations connected with the management and administration of the Plan, it will delete Participant's Data as soon as it has accomplished all the necessary legal obligations connected with the management and administration of the Plan.

Participant understands that Data processing related to the purposes specified above shall take place under automated or non-automated conditions, anonymously when possible, that comply with the purposes for which Data are collected and with such confidentiality and security provisions as set forth by applicable Italian data privacy laws and regulations, with specific reference to Legislative Decree no. 196/2003.

The processing activity, including communication, the transfer of Participant's Data abroad, including outside of the European Economic Area, as herein specified and pursuant to applicable Italian data privacy laws and regulations, does not require Participant's consent thereto as the processing is necessary to performance of contractual obligations related to implementation, administration and management of the Plan. Participant understands that, pursuant to Section 7 of the Legislative Decree no. 196/2003, Participant has the right to, including but not limited to, access, delete, update, ask for rectification of Participant's Data and cease, for legitimate reason, the Data processing. Furthermore, Participant is aware that Participant's Data will not be used for direct marketing purposes. In addition, the Data provided can be reviewed and questions or complaints can be addressed by contacting the Company.

Plan Document Acknowledgment. In participating in the Plan, Participant acknowledges that he or she has received a copy of the Plan and this Notice and has reviewed the Plan and this Notice in their entirety and fully understands and accepts all provisions of the Plan and this Notice. Participant further acknowledges that Participant has read and specifically and expressly approves the sections of the Notice and Appendix A addressing (i) Forfeiture Upon Termination of Continuous Service (Section 2 of the Notice), (ii) Tax Consequences, Withholding, and Liability (Section 3 of the Notice), (iii) Governing Law; Severability (Section 11(f) of the Notice), (iv) Imposition of Other Requirements (Section 11(i) of the Notice); (v) Nature of Grant (Section 3 of Appendix A), (vi) Venue (Section 4 of Appendix A), (vii) Language (Section 6 of Appendix A), and (vii) the Data Privacy section set forth above in this Appendix B.

Notifications

**Exchange Control Information.** Participants are required to report investments held abroad or foreign financial assets (*e.g.*, cash, Shares) that may generate income taxable in Italy on an annual tax return (UNICO Form, RW Schedule) or on a special form if no tax return is due, irrespective of their value. The same reporting duties apply to Italian residents who are beneficial owners of the investments, even if they do not directly hold investments abroad or foreign assets.

**Foreign Asset/Account Reporting Information**. A tax on the value of any financial assets held outside of Italy by Italian residents will apply at an annual rate of 0.2% for fiscal year 2014. The taxable amount will be the fair market value of the financial assets, assessed at the end of the calendar year in the place where the financial assets are held, using the documentation issued by the local broker. Participants should consult their personal tax advisors for details regarding this requirement.

## **JAPAN**

### Notifications

**Foreign Asset/Account Reporting Information.** Participants holding assets outside of Japan (*e.g.*, Shares acquired under the Plan) with a value exceeding \$50,000,000 (as of December 31 each year) are required to comply with annual tax reporting obligations with respect to such assets. Participants should consult their personal tax advisors for details regarding this requirement.

## **SINGAPORE**

## Notifications

Securities Law Information. The grant of RSUs under the Plan is being made pursuant to the "Qualifying Person" exemption under section 273(1)(f) of the Singapore Securities and Futures Act (Chapter 289, 2006 Ed.) ("SFA"). The Plan has not been lodged or registered as a prospectus with the Monetary Authority of Singapore. Further, the RSUs granted under the Plan are subject to section 257 of the SFA and Participant is not permitted to sell, or offer to sell, any Shares in Singapore unless such sale or offer is made pursuant to the exemptions under Part XIII Division (1) Subdivision (4) (other than section 280) of the SFA.

**Director Notification Obligation.** Directors, associate directors or shadow directors of a Singapore Subsidiary or Affiliate are subject to certain notification requirements under the Singapore Companies Act. Among these requirements is an obligation to notify such entity in writing within two business days of any of the following events: (i) the acquisition or disposal of an interest (*e.g.*, RSUs granted under the Plan or Shares) in the Company or any Subsidiary or Affiliate, (ii) any change in previously-disclosed interests (*e.g.*, sale of Shares), or (iii) becoming a director, associate director or shadow director of a Subsidiary or Affiliate in Singapore, if the individual holds such an interest at that time.

## UNITED KINGDOM

## Terms and Conditions

**Tax Obligations.** The following provision supplements <u>Section 3 of the Notice</u> as supplemented by <u>Section 2 of Appendix A</u> (Tax Consequences, Withholding, and Liability):

If payment or withholding of any income tax liability arising in connection with Participant's participation in the Plan is not made by Participant to the Employer within ninety (90) days of the end of the U.K. tax year during which the event giving rise to the income tax liability occurs or such other period specified in Section 222(1)(c) of the U.K. Income Tax (Earnings and Pensions) Act 2003 (the "**Due Date**"), Participant understands and agrees that the amount of any uncollected income tax will constitute a loan owed by Participant to the Company and/or the Employer, effective on the Due Date. Participant further understands and agrees that the loan will bear interest at the then-current official rate of Her Majesty's Revenue and Customs ("**HMRC**"), it will be immediately due and repayable by Participant, and the Company and/or the Employer may recover it at any time thereafter by any of the means referred to in the Plan or this Notice.

Notwithstanding the foregoing, if Participant is a director or an executive officer of the Company (within the meaning of such terms for purposes of Section 13(k) of the Exchange Act), Participant will not be eligible for such a loan to cover the income tax liability. In the event that Participant is a director or executive officer and the income tax is not collected from or paid by Participant by the Due Date, Participant understands that the amount of any uncollected income tax may constitute an additional benefit to Participant on which additional income tax and National Insurance Contributions will be payable. Participant understands and agrees that Participant will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or the Employer (as appropriate) for the value of any employee National Insurance Contributions ("NICs") due on this additional benefit which the Company or the Employer may recover from Participant by any of the means referred to in the Plan or this Notice.

**Joint Election for Transfer of Liability for Employer National Insurance Contributions.** The following provision supplements <u>Section 3 of the Notice</u> as supplemented by <u>Section 2 of Appendix A</u> (Tax Consequences, Withholding, and Liability):

As a condition of participation in the Plan and the issuance of Shares upon vesting of the RSUs, Participant agrees to accept any liability for secondary Class 1 NICs that may be payable by the Company or the Employer in connection with the RSUs and any event giving rise to Tax Obligations (the "Employer NICs"). The Employer NICs may be collected by the Company or the Employer using any of the methods described in the Plan or this Notice.

Without prejudice to the foregoing, Participant agrees to execute a joint election with the Company and/or the Employer (a "Joint Election"), the form of such Joint Election being formally approved by HMRC, and any other consent or elections required to accomplish the transfer of the Employer NICs liability to Participant. Participant further agrees to execute such other elections as may be required by any successor to the Company and/or the Employer for the purpose of continuing the effectiveness of Participant's Joint Election. If Participant does not complete the Joint Election prior to vesting of the RSUs, or if approval of the Joint Election is withdrawn by HMRC and a new Joint Election is not entered into, the RSUs shall become null and void and will not vest, without any liability to the Company, the Employer or any Affiliate.

## THE RUBICON PROJECT, INC.

### 2014 INDUCEMENT GRANT EQUITY INCENTIVE PLAN

### STOCK OPTION GRANT NOTICE FOR EMPLOYEES

Notice is hereby given of the grant by The Rubicon Project, Inc. (the "Company") to the Participant named below (the "Participant") of an Option award as described below (the "Option") under the Company's 2014 Inducement Grant Equity Incentive Plan (the "Plan"). The Option gives Participant the right to purchase the number of shares (each a "Share") of the Company's Common Stock, par value \$0.00001 (the "Common Stock") set forth below at the exercise price set forth below and subject to vesting as set forth below. The Option is governed by and subject to this Notice (including any special terms and conditions set forth in any appendices attached hereto), which include various agreements and representations by Participant and the Plan (which is available on the Company intranet (Inside RP) and incorporated herein by reference). In the event of a conflict between the terms of this Notice and the Plan, the terms of the Plan shall control. By acceptance of the Option, and also by acceptance through performance of the vesting requirements and by exercising the Option, Participant agrees to the terms and conditions set forth in this Notice (including any special terms and conditions set forth in any appendices attached hereto) and the Plan. Capitalized terms used but not defined in this Notice shall have the meanings given to them in the Plan.

Participant Name:	
Number of Shares Subject to Option:	
Issuance Date:	
Type of Option:	Nonstatutory Stock Option
Exercise Price:	\$ per share
Vesting Commencement Date:	

**Expiration Date**: Subject to any separate written agreement between the Company and Participant, and subject to earlier termination as described below, the Option will expire and cease to be exercisable on the tenth anniversary of the Issuance Date.

**Exercise**: The Option may be exercised only to the extent vested. Exercise is effected by Participant's delivery of written notice to the Company in the manner determined by the Company specifying the exercise date and number of Shares to be purchased, together with payment of the exercise price for the Shares purchased. The exercise price must be paid in cash unless the Company, in its discretion, allows another form of payment specified in the Plan.

## Vesting Schedule:

Subject to any vesting acceleration provisions applicable to the Options contained in the Plan and/or any employment or service agreement, offer letter, severance agreement, or any other agreement between Participant and the Company or any Affiliate (such agreement, a "Separate Agreement"):

- (i) the Option shall vest (i) with respect to 25% of the underlying Shares on the first anniversary of the Vesting Commencement Date (the "First Vesting Date"), and (ii) with respect to the remaining 75% of the underlying Shares in 36 equal consecutive monthly installments thereafter, each consisting of 1/48 of the initial grant, on the same day of each calendar month following the First Vesting Date as the day of the month on which the Vesting Commencement Date occurs, provided that vesting is subject to Continuous Service and vesting will not occur on a particular scheduled vesting date if the Participant is not in Continuous Service on that scheduled vesting date; (ii) no vesting will occur before the first scheduled vesting date, and vesting will occur only on scheduled vesting dates, without any ratable vesting for periods of time between vesting dates;
- (ii) vesting will be suspended during the portion of any leave of absence (LOA) Participant has in excess of 90 days, and if Participant returns to work following such a LOA, then an amount of time equal to the period that vesting was suspended, and vesting dates that occurred within that time period, will be added to the end of the originally scheduled vesting period to give Participant an opportunity to vest in the Shares that would have vested during the period that vesting was suspended. Subject to Continuous Service, vesting will occur on each such additional vesting date in the amount of Shares not vested on the corresponding vesting date during the period of the suspension. However, in no case will the vesting period extend beyond the Expiration Date; and
- (iii) subject to Section 2 below, cessation of Participant's Continuous Service for any or no reason before the Option vests in full will result in cessation of vesting of the Option.

Furthermore, under all circumstances, the vesting of Options shall be subject to the satisfaction of Participant's obligations as set forth in Section 6(b).

### 1. Prior to Exercise.

Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of the Option, whether or not vested; stockholder rights accrue only in respect of Shares that have been issued by the Company and recorded on the records of the Company or its transfer agents or registrars following proper exercise of the Option. No adjustment shall be made for any dividends (ordinary or extraordinary, whether cash, securities, or other property) or distributions or other rights for which the record date is prior to the date Shares are issued following proper exercise of the Option.

2. Forfeiture Upon Termination of Continuous Service. Except as otherwise provided in the vesting schedule set forth above in this Notice or in a Separate Agreement, if Participant ceases to remain in Continuous Service at any time for any reason, the then-unvested portion of the Option will thereupon terminate and may not be exercised. After termination of Participant's Continuous Service for any reason or no reason, Participant (or in the case of Participant's death, Participant's heirs or estate) may exercise the Option, but only to the extent vested at the time of or as a result of termination of Participant's Continuous Service and not previously exercised, until the earlier of (i) the Expiration Date, or (ii) the close of business on the 90th day after termination of Participant's Continuous Service, or the 180th day if termination of Continuous Service is due to Participant's death or Disability, and after the Expiration Date or the 90th or 180th day after termination of Participant's Continuous Service, as the case may be, the Option will terminate and be forfeited at no cost to the Company and Participant will have no further rights with respect thereto.

## 3. Tax Consequences, Withholding, and Liability.

(a) Participant understands that Participant may suffer adverse tax consequences as a result of the grant, vesting or exercise of the Option and issuance and/or disposition of the Shares. Participant understands that the actual tax consequences associated with the Option and Shares are complicated and depend, in part, on Participant's specific situation and may also depend on the resolution of currently uncertain tax law and other variables not within the control of the Company. THEREFORE, PARTICIPANT SHOULD SEEK INDEPENDENT ADVICE REGARDING THE APPLICABLE PROVISIONS OF THE FEDERAL TAX LAW AND THE TAX LAWS OF ANY MUNICIPALITY, STATE OR NON-U.S. JURISDICTION TO WHICH PARTICIPANT IS SUBJECT. By accepting (through performance) the Option and by its exercise, Participant acknowledges and agrees that Participant has either consulted with a competent tax advisor independent of the Company to obtain tax advice concerning the Option and Shares in light of Participant's specific situation or has had the opportunity to consult with such a tax advisor and has chosen not to do so. Neither the Company nor any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents, any written or oral advice or representation regarding the U.S. federal, state, local or non-U.S. tax consequences of the receipt, vesting and exercise of the Option, the other transactions contemplated by this Notice, or the value of the Company or the Options or Shares at any time. With respect to such matters, Participant relies solely on Participant's own advisors

(b) Participant (and not the Company) shall be responsible for Participant's own tax liability that may arise as a result of the receipt, vesting and exercise of the Option, or the other transactions contemplated by this Notice. Pursuant to such procedures as the Plan administrator may specify from time to time, the Company shall satisfy its obligations to pay withholding taxes or other tax deposits in connection with the receipt, ownership and/or vesting of the Option, the issuance of Shares upon exercise of the Option, or the other transactions contemplated by this Notice in accordance with

applicable law or regulation (the "Tax Obligations"). If amounts paid by the Company in respect of Tax Obligations are less than Participant's tax obligations, Participant is solely responsible for any additional taxes due. If amounts paid by the Company in respect of Tax Obligations exceed Participant's tax obligations, Participant's sole recourse will be against the relevant taxing authorities, and the Company and its Affiliates will have no obligation to Participant in respect thereof. Participant is responsible for determining Participant's actual income tax liabilities and making appropriate payments to the relevant taxing authorities to fulfill Participant's tax obligations and avoid interest and penalties.

- (c) Payment by the Company or its Affiliate of the Tax Obligations will result in a commensurate obligation of Participant to pay, or cause to be paid, to the Company or its Affiliate, in accordance with Section 8(g) of the Plan, the amount of Tax Obligations so paid, and the Company shall not be required to issue any Shares unless and until Participant has satisfied this obligation. To the extent determined appropriate by the Company in its discretion, it shall have the right (but not the obligation) to cause Participant to satisfy any or all Tax Obligations by withholding and retaining Shares otherwise deliverable to Participant having an aggregate Fair Market Value equal to the amount of such Tax Obligations.
- 4. No Guarantee of Continued Service. THE VESTING OF THE OPTION PURSUANT TO THE VESTING SCHEDULE APPLICABLE THERETO IS EARNED ONLY BY CONTINUOUS SERVICE AT THE WILL OF THE COMPANY (OR THE AFFILIATE OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED THE OPTION OR ACQUIRING SHARES UPON EXERCISE OF THE OPTION. THIS NOTICE, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE APPLICABLE TO THE OPTION DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT TO PROVIDE SERVICE FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE AFFILIATE OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S CONTINUOUS SERVICE AT ANY TIME, FOR ANY REASON OR NO REASON, WITH OR WITHOUT NOTICE, AND WITH OR WITHOUT CAUSE UNLESS OTHERWISE PROVIDED IN A SEPARATE AGREEMENT.

# 5. Participant Representations.

- (a) Participant acknowledges that (i) Participant was and is free to use professional advisors of Participant's choice in connection with this Notice and grant of the Option, that Participant understands this Notice and the meaning and consequences of receiving the Option and Shares issued upon exercise of the Option; (ii) Participant has reviewed and understands this Notice and the Plan; (iii) receipt of the Option and any Shares issued upon exercise is voluntary and Participant is accepting the Option and any Shares issued upon exercise freely and without coercion or duress; and (iv) Participant has not received and is not relying, and will not rely, upon any advice, representations or assurances made by or on behalf of the Company or any of its Affiliates or any employee of or counsel to the Company or any of its Affiliates regarding any tax or other effects or implications of the Option, its exercise, receipt of Shares, or other matters contemplated by this award of Options.
- (b) Participant is aware of the Company's business affairs and financial condition and understands that an investment in the Shares involves a high degree of risk. Participant has not received and is not relying, and will not rely, upon any advice, representations or assurances made by or on behalf of the Company or any of its Affiliates or any employee of or counsel to the Company or any of its Affiliates regarding the Company's prospects or the value of the Option or Shares issuable upon

exercise.

### 6. Additional Conditions to Issuance of Stock.

(a) Legal and Regulatory Compliance. The issuance of Shares upon or after exercise of the Option shall be subject to compliance with all applicable requirements of federal, state or foreign law with respect to such securities. If at any time the Company determines, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary as a condition to the issuance of Shares to Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. If the Company determines that the issuance of any Shares will violate federal securities laws or other applicable laws or regulations or the requirements of any exchange or market system upon which the Shares are listed, the Company may defer issuance until the earliest date at which the Company reasonably anticipates that the issuance of Shares will no longer cause such violation. Accordingly, Participant may not be able to receive Shares when desired even though Participant has requested to exercise the Option. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority, but the inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance of any Shares shall relieve the Company of any liability in respect of the failure to issue such Shares as to which such requisite authority shall not have been obtained. As a condition to the issuance of Shares, the Company may require Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company. Without limiting the foregoing, if at the time of exercise of the Option, there is not in effect under the Securities Act of 1933, as amended (the "Securities Act"), a registration statement covering the Shares to be issued, and available for delivery a prospectus meeting the requirements of Section 10(a)(3) of the Securities Act, Participant shall, if required by the Company, as a condition to exercise of the Option and issuance of the Shares, make appropriate representations in a form satisfactory to the Company to support issuance of the Shares in compliance with applicable laws and regulations, including to the effect that such Shares will not be sold other than (A) pursuant to an effective registration statement under the Securities Act, or an applicable exemption from the registration requirements of such Act; (B) in compliance with all applicable state securities laws and regulations; and (C) in compliance with all terms and conditions of the Plan, this Notice, and any other written agreement between Participant and the Company or any of its Affiliates.

(b) Obligations to the Company. As a condition to receipt of the Options and issuance of Shares as a result of exercise, Participant must enter into the Company's Intellectual Property Assignment and Confidential Information Agreement, or a similar or successor agreement for the protection of the Company's intellectual property and confidential information, in form specified by the Company (the "Proprietary Interests Agreement"), if Participant has not already done so, and Participant's acceptance of the Option and any Shares issued upon exercise will constitute Participant's agreement to the Proprietary Interests Agreement. If Participant breaches in any material respect the Proprietary Interests Agreement or any other contract between Participant and the Company, or Participant's common law duty of confidentiality or trade secret protection, or any Company policy prohibiting misappropriation of property or any illegal or fraudulent acts, the Company may suspend any vesting and/or exercise of the Option and/or issuance of any Shares pending Participant's cure of such breach, and if such breach cannot be cured or is not cured to the Company's reasonable satisfaction within such time not less than twenty (20) days as the Company may specify, the Company may terminate the

Option to the extent not exercised and will have no obligation to issue any Shares in respect of the terminated Option or to provide any consideration to Participant in respect thereof.

## 7. Handling of Shares; Restrictive Legends and Stop-Transfer Orders.

- (a) <u>Book Entries</u>. The Company will cause the Shares issuable upon exercise of the Option to be recorded in book entry or other electronic form and reflected in records maintained by or for the Company.
- (b) <u>Legends</u>. Each data base entry representing any Shares issuable upon exercise of the Option may be endorsed with legends substantially as set forth below, as well as such other legends as the Company may deem appropriate to implement the provisions of this Notice or comply with applicable laws and regulations and Company policies:

THE SHARES REPRESENTED HEREBY ARE SUBJECT TO RESTRICTIONS ON TRANSFER FOR A PERIOD OF TIME FOLLOWING THE EFFECTIVE DATE OF ANY UNDERWRITTEN PUBLIC OFFERING OF THE COMPANY'S SECURITIES SET FORTH IN AN AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES AND MAY NOT BE SOLD OR OTHERWISE DISPOSED OF BY THE HOLDER PRIOR TO THE EXPIRATION OF SUCH PERIOD WITHOUT THE CONSENT OF THE COMPANY OR THE MANAGING UNDERWRITER.

- (c) <u>Stop-Transfer Notices</u>. In order to ensure compliance with the restrictions referred to herein and Company policies, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (d) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Notice or any other agreement to which the Shares are subject or any laws governing the Shares or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferree to whom such Shares shall have been so transferred.
- 8. Restrictions on Transfer. Except as otherwise expressly provided in this Notice, the Option will not in whole or part be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of any interest in the Option, or upon any attempted sale under any execution, attachment or similar process, the affected Option will become null and void without further obligation to Participant. The Company may impose such restrictions, conditions or limitations as it determines appropriate as to the timing and manner of any resales by the Participant or other subsequent transfers by the Participant of any Shares issued upon the exercise of the Option, including without limitation (a) restrictions under an insider trading policy, (b) restrictions designed to delay and/or coordinate the timing and manner of sales by Participant and other holders, and (c) restrictions as to the use of a specified brokerage firm for such resales or other transfers.
- 9. Lock-Up. In connection with any underwritten public offering by the Company of its equity securities pursuant to a registration statement filed under the Securities Act, upon the request of the Company or the underwriters managing such offering, during the Lock-up Period (as defined below)

Participant shall not, without the prior written consent of the Company or its underwriters, directly or indirectly sell, make any short sale of, loan, hypothecate, pledge, offer, grant or sell any option or other contract for the purchase of, purchase any option or other contract for the sale of, enter into any swap, hedging or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of, or otherwise dispose of or transfer, or agree to engage in any of the foregoing transactions with respect to, any Shares or other securities into which the Shares may be converted or that are issued in respect of the Shares (other than those included in the registration). For this purpose, the "Lock-up Period" means such period of time after the effective date of the registration as is requested by the Company or the underwriters; provided that such period shall not exceed 180 days (or such additional period as may reasonably be requested by the Company or such underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports or (ii) analyst recommendations and opinions, including (without limitation) the restrictions set forth in Rule 2711(f)(4) of the National Association of Securities Dealers and Rule 472(f)(4) of the New York Stock Exchange, as amended, or any similar successor rules). The Company's underwriters shall be beneficiaries of the provision set forth in this Section 9, and Participant shall execute and deliver such agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In addition, if requested by the Company or the underwriters of Common Stock (or other securities) of the Company, Participant shall provide, within ten (10) days of such request, such information as may be required or reasonably requested by the Company or the underwriters in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Securities Act. The obligations described in this Section 9 shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a Commission Rule 145 transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions with respect to the Shares (or other securities) subject to the foregoing restriction until the end of said one hundred and eighty (180) day (or other) period. Participant agrees, and will cause any transferee to agree, that any transferee of the Option shall be bound by this Section 9.

### 10. Additional Agreements.

- (a) <u>Electronic Delivery</u>. The Company may, in its sole discretion, decide to deliver any documents related to the Option or Shares by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to administration of this Notice, the Option and the Shares through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
- (b) <u>Personal Information</u>. To facilitate the administration of the Plan and any successor plan and the terms of this Notice, it may be necessary for the Company (or its payroll administrators) to collect, hold and process certain personal information about Participant, including, but not limited to, Participant's name, home address and telephone number, date of birth, social insurance number or other identification number, salary, nationality, job title, any shares of Company Common Stock owned, relationship to the Company, details of all awards issued under the Plan or any predecessor or successor plan or any other entitlement to shares of Company Common Stock awarded, canceled, exercised, vested, unvested or outstanding in Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("Data") and to transfer this Data to certain third parties such as transfer agents, stock plan administrators, and brokers with whom Participant or the Company may elect to deposit any Shares. Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of Participant's Data for the exclusive

purpose of implementing, administering and managing Participant's participation in the Plan and any predecessor and successor plan. Participant understands that Data will be transferred to the Company's transfer agent, broker, administrative agents or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan and any predecessor and successor plan. Participant understands that the recipients of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than Participant's country. Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker, administrative agents, and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan or any predecessor or successor plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participant's participation in the Plan or any predecessor or successor plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage Participant's participation in the Plan or any predecessor or successor plan. The Participant understands if he or she resides outside the United States, he or she may, at any time, view Data, request additional information about the storage and processing of Data, require any necessary amendments to Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources representative. Further, Participant understands that he or she is providing the consents herein on a purely voluntary basis. If Participant does not consent, or if Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Company will not be adversely affected; the only consequence of refusing or withdrawing Participant's consent is that the Company would not be able to grant Options or other equity awards or administer or maintain such awards. Therefore, Participant understands that refusing or withdrawing his or her consent may affect Participant's ability to participate in the Plan or any successor plan. For more information on the consequences of Participant's refusal to consent or withdrawal of consent, Participant understands that he or she may contact his or her local human resources representative. Finally, upon request of the Company, Participant agrees to provide an executed data privacy consent form to the Company (or any other agreements or consents that may be required by the Company) that the Company may deem necessary to obtain under the data privacy laws in Participant's country, either now or in the future. Participant understands that he or she will not be able to participate in the Plan if he or she fails to execute any such consent or agreement.

(c) <u>Proprietary Information</u>. Participant agrees that all financial and other information relating to the Company furnished to Participant constitutes "Proprietary Information" that is the property of the Company. Participant shall hold in confidence and not disclose or, except within the scope of Participant's Continuous Service, use any Proprietary Information. Participant shall not be obligated under this paragraph with respect to information Participant can document is or becomes readily publicly available without restriction through no fault of Participant. Upon termination of Participant's Continuous Service, Participant shall promptly return to Company all items containing or embodying Proprietary Information (including all copies). This paragraph supplements, but does not limit, any other agreement between Participant and the Company, or any applicable law, related to protection, ownership, or use of the Company's information or property.

(d) <u>Consideration</u>. The Option and Shares issued upon exercise are issued in consideration of services provided by Participant and/or other benefit to the corporation within the meaning of Section 152 of the General Corporation Law of the State of Delaware; Participant is not required to make any cash payment to the Company in respect of issuance of Options, but is required to

pay the exercise price listed in the Notice prior to the issuance of Shares.

### 11. General.

- (a) No Waiver; Remedies. Either party's failure to enforce any provision of this Notice shall not in any way be construed as a waiver of any such provision, or prevent that party from thereafter enforcing such provision and each and every other provision of this Notice. The rights granted both parties herein are cumulative and shall not constitute a waiver of either party's right to assert all other legal remedies available to it under the circumstances.
- (b) <u>Successors and Assigns</u>. The terms of this Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth, the terms of this Notice shall be binding upon Participant and Participant's heirs, executors, administrators, successors and assigns. The rights and obligations of Participant under this Notice may only be assigned with the prior written consent of the Company.
- (c) <u>Notices</u>. Any notice hereunder shall be in writing (which shall include electronic transmission) and shall be deemed received (i) the business day following electronic verification of receipt if sent electronically, (ii) upon personal delivery to the party to whom the notice is directed, (iii) the business day following deposit with a reputable overnight courier, or (iv) five days after deposit in the U.S. mail, First Class with postage prepaid. Notice shall be addressed to the Company at its principal executive office and to Participant at the address that he or she most recently provided to the Company. Participant agrees that it is Participant's responsibility to notify the Company of any changes to his or her mailing address so that Participant may receive any shareholder information to be delivered by regular mail.
- (d) Interpretation. Headings herein are for convenience of reference only, do not constitute a part of this Notice, and will not affect the meaning or interpretation of this Notice. References herein to Sections are references to the referenced Section hereof, unless otherwise specified. The Board or its Committee will have the power to interpret the Plan and this Notice and to adopt such rules for the administration, interpretation and application of the Plan and this Notice as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of the extent, if any, to which the Option has vested). All actions taken and all interpretations and determinations made by the Board or its Committee in good faith will be final and binding upon Participant, the Company and all other interested persons. Neither the Board or its Committee nor any person acting on behalf of the Board or its Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Notice.
- (e) <u>Modifications to Notice</u>. Modifications to this Notice can be made only in an express written contract executed by a duly authorized officer of the Company and shall not require the consent of the Participant unless such modification would materially adversely affect the rights of the Participant hereunder. Notwithstanding anything to the contrary in the Plan or this Notice, the Company reserves the right to revise this Notice as it deems necessary or advisable, in its sole discretion and without the consent of Participant, to comply with Section 409A or to otherwise avoid imposition of any additional tax or income recognition under Section 409A in connection to the Option.
- (f) Governing Law; Severability. This Notice is governed by the internal substantive laws, but not the choice of law rules, of Delaware. If any provision of this Notice becomes or is declared by a court or arbitrator having jurisdiction over a dispute hereunder to be illegal, unenforceable or void, such provision shall be amended to the extent necessary to conform to applicable law so as to be valid and

enforceable and to achieve, to the extent possible, the economic, business and other purposes of such illegal, unenforceable, or void provision or, if such provision cannot be so amended without materially altering the intention of the parties, then such provision shall be deleted from this Notice and the remainder of this Notice shall continue in full force and effect.

- (g) Entire Agreement. The Plan and this Notice, along with any Separate Agreement (to the extent applicable), form a contract and constitute the entire understanding between Participant and the Company with respect to the Option and the Shares issuable upon exercise of the Option and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect thereto.
- (h) Appendix. The award of Options shall be subject to any additional terms and conditions for Non-U.S. Employees set forth in Appendix A attached hereto ("Appendix A") and any special terms and conditions for Participant's country set forth in Appendix B attached hereto ("Appendix B"). Moreover, if Participant relocates to one of the countries included in Appendix B, the special terms and conditions for such country will apply to Participant to the extent the Company determines that the application of such terms and conditions is necessary or advisable for legal or administrative reasons. Appendix A and Appendix B constitute part of the Notice.

1 1	reements or undertakings that may be necessary to accomplish the foregoing.
Dated:	THE RUBICON PROJECT, INC.
	Ву:
	Name:
	Title:

(i) Imposition of Other Requirements. The Company reserves the right to impose other requirements on Participant's participation in the Plan, on

### APPENDIX A

## ADDITIONAL TERMS AND CONDITIONS OF STOCK OPTION GRANT FOR NON-U.S. EMPLOYEES

- 1. Terms of Plan Participation for Non-U.S. Participants. Participant understands that this Appendix A contains additional terms and conditions that, together with the Plan and the Notice, govern Participant's participation in the Plan if Participant is working or resident in a country other than the United States. Participant further understands that Participant's participation in the Plan also will be subject to any terms and conditions for Participant's country set forth in Appendix B attached hereto. Capitalized terms used but not defined in this Appendix A shall have the same meanings assigned to them in the Plan and/or Notice.
  - 2. Tax Consequences, Withholding, and Liability. The following provision supplements Section 3 of the Notice:

By accepting (through performance) the Option and by its exercise, Participant authorizes the Company and/or the Subsidiary or Affiliate employing or retaining Participant (the "Employer"), or their respective agents, at the Company's discretion, to satisfy the obligations with regard to all Tax Obligations by one or a combination of the methods set forth in Section 8(g) of the Plan. If Participant is or becomes subject to Tax Obligations in more than one jurisdiction, Participant acknowledges that the Company and/or the Employer (or former employer, as applicable) may be required to withhold or account for Tax Obligations in more than one jurisdiction.

If the Tax Obligations are satisfied by withholding in Shares, for tax purposes, Participant is deemed to have been issued the full number of Shares subject to the exercised Options, notwithstanding that a number of the Shares are held back solely for the purpose of paying the Tax Obligations.

- 3. Nature of Grant. By accepting (through performance) the Option and by its exercise, Participant acknowledges, understands and agrees that:
- (a) the grant of the Option is voluntary and occasional and does not create any contractual or other right to receive future grants of options, or benefits in lieu of options, even if options have been granted in the past;
  - (b) all decisions with respect to future option or other grants, if any, will be at the sole discretion of the Company;
- (c) the Option and any Shares acquired under the Plan, and the income and value of the same, are not intended to replace any pension rights or compensation;
- (d) the Option and any Shares acquired under the Plan, and the income and value of the same, are not part of Participant's normal or expected compensation for any purposes including, but not limited to, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement benefits or payments or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or the Employer or any Affiliate;

- (e) the future value of the Shares underlying the Option is unknown, indeterminable, and cannot be predicted with certainty;
- (f) if the underlying Shares do not increase in value, the Option will have no value;
- (g) if Participant exercises the Option and acquires Shares, the value of such Shares may increase or decrease in value, even below the exercise price;
- (h) for purposes of the Option, Participant's Continuous Service will be considered terminated as of the date Participant is no longer actively providing services to the Company or the Employer (regardless of the reason for such termination and whether or not later found to be invalid or in breach of employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment or service agreement, if any), and unless otherwise expressly provided in this Notice or determined by the Company, (i) Participant's right to vest in the Option under the Plan, if any, will terminate as of such date and will not be extended by any notice period (e.g., Participant's period of service would not include any contractual notice period or any period of "garden leave" or similar period mandated under employment laws in the jurisdiction where Participant is employed or the terms of Participant's employment agreement, if any); and (ii) the period (if any) during which Participant may exercise the Option after such termination of Continuous Service will commence on such date and will not be extended by any notice period mandated under employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment or service agreement, if any; the Board or Committee shall have the exclusive discretion to determine when Participant is no longer actively providing services for purposes of the Option (including whether Participant may still be considered to be providing services while on a leave of absence);
- (i) no claim or entitlement to compensation or damages shall arise from forfeiture of the Option resulting from the termination of Participant's Continuous Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment or service agreement, if any), and in consideration of the grant of the Option to which Participant is otherwise not entitled, Participant irrevocably agrees never to institute any claim against the Company, the Employer or any Affiliate, waives his or her ability, if any, to bring any such claim, and releases the Company, the Employer and any Affiliate from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, Participant shall be deemed irrevocably to have agreed not to pursue such claim and agrees to execute any and all documents necessary to request dismissal or withdrawal of such claim;
- (j) unless otherwise provided in the Plan or by the Company in its discretion, the Option and the benefits evidenced by the Notice do not create any entitlement to have the Option or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the Shares; and
- (k) neither the Company, the Employer nor any Affiliate shall be liable for any foreign exchange rate fluctuation between Participant's local currency and the U.S. dollar that may affect the value of the Option or of any amounts due to Participant pursuant to the exercise of the Option or the subsequent sale of any Shares acquired upon exercise.
- 4. <u>Venue</u>. For purposes of litigating any dispute that arises under the Notice, the parties hereby submit to and consent to the jurisdiction of the State of California, and agree that such litigation will be conducted in the courts of Los Angeles County, California, or the federal courts for the United

States for the Central District of California, and no other courts, where this award of Options is made and/or to be performed.

- 5. <u>Insider Trading</u>. By participating in the Plan, Participant agrees to comply with the Company's policy on insider trading (to the extent that it is applicable to Participant). Further, Participant acknowledges that Participant's country of residence may also have laws or regulations governing insider trading and that such law or regulations may impose additional restrictions on Participant's ability to participate in the Plan (*e.g.*, acquiring or selling Shares) and that Participant is solely responsible for complying with such laws or regulations.
- 6. <u>Language</u>. If the Notice or any other document related to the Plan has been translated into a language other than English and the meaning of the translated version is different than the English version, the English version will control.

### APPENDIX B

## COUNTRY-SPECIFIC PROVISIONS FOR NON-U.S. EMPLOYEES

#### Terms and Conditions

This Appendix B includes additional terms and conditions that govern the Options granted to Participant under the Plan if Participant works or resides in one of the countries listed below. If Participant is a citizen or resident of a country other than the one in which Participant currently is working (or if Participant is considered as such for local law purposes), or if Participant transfers employment or residence to another country after Options have been granted to Participant under the Plan, the Company, in its discretion, will determine the extent to which the terms and conditions herein will be applicable to Participant.

Capitalized terms used but not otherwise defined herein shall have the meaning given to such terms in the Plan, the Notice or Appendix A.

## Notifications

This Appendix B also includes information regarding securities laws, exchange controls and certain other issues of which Participant should be aware with respect to his or her participation in the Plan. The information is based on the securities, exchange control and other laws in effect in the respective countries as of November 2014. Such laws are often complex and change frequently. As a result, the Company recommends that Participant not rely on the information in this Appendix B as the only source of information relating to the consequences of his or her participation in the Plan because the information included herein may be out of date at the time that Participant acquires Shares under the Plan or subsequently sells such Shares.

In addition, the information contained herein is general in nature and may not apply to Participant's particular situation and the Company is not in a position to assure Participant of any particular result. Accordingly, Participant is advised to seek appropriate professional advice as to how the relevant laws in Participant's country may apply to his or her individual situation.

Finally, if Participant is a citizen or resident of a country other than the one in which Participant currently is working or residing (or if Participant is considered as such for local law purposes), or if Participant transfers employment or residence to another country after Options have been granted to Participant under the Plan, the information contained herein may not be applicable to Participant in the same manner.

## **AUSTRALIA**

## **Terms and Conditions**

Right to Exercise. If the Option vests when the Fair Market Value per Share is equal to or less than the exercise price for the Option, Participant shall not be permitted to exercise the vested Option. The vested Option may be exercised only starting on the business day following the first day on which the Fair Market Value per Share exceeds the exercise price for the Option. Furthermore, notwithstanding the Expiration Date set forth in the Notice, this Option shall automatically expire in the event the Option has not become exercisable pursuant to the preceding sentence within six years and 11 months following the Issuance Date. For the avoidance of doubt, this provision applies also to any unvested Options held by Participants who transfer to Australia after the grant of the Option, as determined by the Company in its

sole discretion.

## Notifications

Securities Law Information. If Shares are acquired under the Plan and subsequently offered for sale to a person or entity resident in Australia, such offer may be subject to disclosure requirements under Australian law. Participants should obtain legal advice regarding any applicable disclosure requirements prior to making any such offer.

### **BRAZIL**

# Terms and Conditions

Compliance with Law. By participating in the Plan, Participant agrees to comply with all applicable Brazilian laws and to pay any and all applicable taxes associated with the acquisition and sale of Shares acquired under the Plan, or the receipt of any dividends in the future.

## Notifications

**Exchange Control Information**. Participants who are residents or domiciled in Brazil must submit a declaration of assets and rights held outside of Brazil, including Shares acquired under the Plan, to the Central Bank if the aggregate value of such assets and rights is at least US\$100,000. Participants should consult their personal legal advisors for further details regarding this requirement.

### **CANADA**

#### Terms and Conditions

Labor Law Acknowledgement. This provision replaces Section 3(h) of Appendix A and supplements Section 2 of the Notice:

for purposes of the Option, Participant's Continuous Service will be considered terminated as of the earlier of: (i) the date on which Participant's employment with the Company and/or the Employer is terminated; (ii) the date on which Participant receives a written notice of termination of Continuous Service regardless of any notice period or period of pay in lieu of such notice required under any employment laws in Participant's country (including, without limitation, statutory law, regulatory law, and/or common law), even if such law is otherwise applicable to Participant's benefits from the Company and/or the Employer; or (iii) the date on which Participant is no longer actively providing services to the Company and/or the Employer (regardless of the reason for such termination and regardless of whether it is later found to be invalid), and unless otherwise expressly provided in this Notice or determined by the Company, (i) Participant's right to vest in the Option under the Plan, if any, will terminate as of such date; and (ii) the period (if any) during which Participant may exercise the Option after such termination of Continuous Service will commence on such date and will not be extended by any notice period mandated under employment laws in the jurisdiction where Participant is employed or providing services, or the terms of Participant's employment agreement, if any; the Board or Committee shall have the exclusive discretion to determine when Participant is no longer actively providing services for purposes of the Option (including whether Participant may still be considered to be providing services while on a leave of absence);

### **Notifications**

**Securities Law Information**. Shares acquired under the Plan may result in Canadian securities laws issues if such Shares are sold through a broker other than the designated broker or if the sale does not take place through the facilities of a stock exchange outside Canada on which the Shares are listed (*i.e.*, the New York Stock Exchange).

**Tax Reporting Obligation**. Foreign property (including Shares acquired under the Plan and possibly the Option) must be reported on Form T1135 (Foreign Income Verification Statement) if the total value of foreign property exceeds C\$100,000 at any time during the year. Participants should consult their personal tax advisors for further details regarding this requirement.

## **FRANCE**

### Terms and Conditions

**Language Consent.** By participating in the Plan, Participant confirms having read and understood the documents relating to the Options and his or her participation in the Plan (*i.e.*, the Plan and this Notice), which were provided to Participant in the English language. Participant accepts the terms of these documents accordingly.

Consentement Relatif à la Langue Utilisée. En participant au Plan, le Participant confirme avoir lu et compris les documents relatifs aux Options et à sa participation au Plan (à savoir, le Plan et le présent Avis) qui lui ont été communiqués en langue anglaise. Le Participant accepte les termes de ces documents en connaissance de cause.

# Notifications

**Foreign Asset/Account Reporting Information.** Participants in France must declare any foreign bank investment, or brokerage account opened, used or closed during the fiscal year to the French tax authorities when filing their annual tax returns. Participants should consult their personal tax advisors for details regarding this requirement.

## **GERMANY**

### Notifications

**Exchange Control Information.** Cross-border payments in excess of €12,500 in connection with the purchase or sale of securities (*e.g.*, transfer of proceeds from the sale of Shares into Germany) must be reported electronically to the German Federal Bank. The online filing portal may be accessed at the website of the German Federal Bank. Participants should consult their personal tax advisors for details regarding this requirement.

## **ITALY**

### Terms and Conditions

**Data Privacy**. This provision replaces in its entirety <u>Section 10(b)</u> of the Notice:

Participant understands that the Company may hold certain personal information about Participant, including Participant's name, home address and telephone number, date of birth, social insurance number or other identification number, salary, nationality, job title, any shares of stock or directorships Participant holds in the Company, details of the Plan or any other entitlement to shares of stock awarded, canceled, exercised, vested, unvested or outstanding in Participant's favor ("Data"), for the exclusive purpose of implementing, administering and managing Participant's participation in the Plan.

Participant also understands that providing the Company with the Data is necessary for the performance of the Plan and that Participant's refusal to provide such Data would make it impossible for the Company to perform its contractual obligations and may affect Participant's ability to participate in the Plan. The Controller of personal data processing is The Rubicon Project, Inc., with registered offices at 12181 Bluff Creek Drive, Playa Vista, CA 90094 U.S.A., and, pursuant to D.lgs 196/2003, its representative in Italy is The Rubicon Project S.r.l. with registered offices at Corsa Giaccomo Matteotti 7, CAP 20121 Milano, Italy.

Participant understands that Participant's Data will not be publicized, but it may be transferred to Morgan Stanley Smith Barney, Equity Administration Solutions, Inc., their respective affiliates and other financial institutions or brokers involved in the management and administration of the Plan. Participant further understands that the Company and/or its Affiliates will transfer Data amongst themselves as necessary for the purpose of implementation, administration and management of Participation in the Plan, and that the Affiliates may each further transfer Data to third parties assisting the Company in the implementation, administration and management of the Plan, including any requisite transfer to Morgan Stanley Smith Barney, Equity Administration Solutions, Inc., their respective affiliates, or another third party with whom Participant may elect to deposit any Shares acquired under the Plan. Such recipients may receive, possess, use, retain and transfer the Data in electronic or other form, for the purposes of implementing, administering and managing Participant's participation in the Plan. Participant understands that these recipients may be located in the European Economic Area, or elsewhere, such as the U.S. Should the Company exercise its discretion in suspending all necessary legal obligations connected with the management and administration of the Plan, it will delete Participant's Data as soon as it has accomplished all the necessary legal obligations connected with the management and administration of the Plan.

Participant understands that Data processing related to the purposes specified above shall take place under automated or non-automated conditions, anonymously when possible, that comply with the purposes for which Data are collected and with such confidentiality and security provisions as set forth by applicable Italian data privacy laws and regulations, with specific reference to Legislative Decree no. 196/2003.

The processing activity, including communication, the transfer of Participant's Data abroad, including outside of the European Economic Area, as herein specified and pursuant to applicable Italian data privacy laws and regulations, does not require Participant's consent thereto as the processing is necessary to performance of contractual obligations related to implementation, administration and management of the Plan. Participant understands that, pursuant to Section 7 of the Legislative Decree no. 196/2003, Participant has the right to, including but not limited to, access, delete, update, ask for rectification of Participant's Data and cease, for legitimate reason, the Data processing. Furthermore, Participant is aware that Participant's Data will not be used for direct marketing purposes. In addition, the Data provided can be reviewed and questions or complaints can be addressed by contacting the Company.

Plan Document Acknowledgment. In participating in the Plan, Participant acknowledges that he or she has received a copy of the Plan and this Notice and has reviewed the Plan and this Notice in their entirety and fully understands and accepts all provisions of the Plan and this Notice. Participant further acknowledges that Participant has read and specifically and expressly approves the sections of the Notice and Appendix A addressing (i) Forfeiture Upon Termination of Continuous Service (Section 2 of the Notice), (ii) Tax Consequences, Withholding, and Liability (Section 3 of the Notice), (iii) Governing Law; Severability (Section 11(f) of the Notice), (iv) Imposition of Other Requirements (Section 11(i) of the Notice); (v) Nature of Grant (Section 3 of Appendix A), (vi) Venue (Section 4 of Appendix A), (vii) Language (Section 6 of Appendix A), and (vii) the Data Privacy section set forth above in this Appendix B.

### **Notifications**

**Exchange Control Information**. Participants are required to report investments held abroad or foreign financial assets (*e.g.*, cash, Shares) that may generate income taxable in Italy on an annual tax return (UNICO Form, RW Schedule) or on a special form if no tax return is due, irrespective of their value. The same reporting duties apply to Italian residents who are beneficial owners of the investments, even if they do not directly hold investments abroad or foreign assets.

Foreign Asset/Account Reporting Information. A tax on the value of any financial assets held outside of Italy by Italian residents will apply at an annual rate of 0.2% for fiscal year 2014. The taxable amount will be the fair market value of the financial assets, assessed at the end of the calendar year in the place where the financial assets are held, using the documentation issued by the local broker. Participants should consult their personal tax advisors for details regarding this requirement.

## **JAPAN**

## Notifications

Exchange Control Information. If Participant remits more than \(\frac{4}{30}\) million for the purchase of Shares in a single transaction, Participant must file a Payment Report with the Ministry of Finance (through the Bank of Japan or the bank carrying out the transaction). The precise reporting requirements vary depending on whether the relevant payment is made through a bank in Japan. If Participant intends to acquire Shares whose value exceeds \(\frac{4}{100}\) million in a single transaction, Participant must also file a Report Concerning Acquisition of Shares ("Securities Acquisition Report") with the Ministry of Finance through the Bank of Japan within 20 days of acquiring the Shares. The forms to make these reports can be acquired from the Bank of Japan.

A Payment Report is required independently from a Securities Acquisition Report. Therefore, if the total amount that Participant pays upon a one-time transaction for exercising the Option and acquiring Shares exceeds ¥100 million, Participant must file both a Payment Report and a Securities Acquisition Report.

**Foreign Asset/Account Reporting Information.** Participants holding assets outside of Japan (*e.g.*, Shares acquired under the Plan) with a value exceeding \$50,000,000 (as of December 31 each year) are required to comply with annual tax reporting obligations with respect to such assets. Participants should consult their personal tax advisors for details regarding this requirement.

### **SINGAPORE**

### Notifications

Securities Law Information. The grant of Options under the Plan is being made pursuant to the "Qualifying Person" exemption under section 273(1)(f) of the Singapore Securities and Futures Act (Chapter 289, 2006 Ed.) ("SFA"). The Plan has not been lodged or registered as a prospectus with the Monetary Authority of Singapore. Further, the Options granted under the Plan are subject to section 257 of the SFA and Participant is not permitted to sell, or offer to sell, any Shares in Singapore unless such sale or offer is made pursuant to the exemptions under Part XIII Division (1) Subdivision (4) (other than section 280) of the SFA.

**Director Notification Obligation.** Directors, associate directors or shadow directors of a Singapore Subsidiary or Affiliate are subject to certain notification requirements under the Singapore Companies Act. Among these requirements is an obligation to notify such entity in writing within two business days of any of the following events: (i) the acquisition or disposal of an interest (*e.g.*, options granted under the Plan or Shares) in the Company or any Subsidiary or Affiliate, (ii) any change in previously-disclosed interests (*e.g.*, sale of Shares), or (iii) becoming a director, associate director or shadow director of a Subsidiary or Affiliate in Singapore, if the individual holds such an interest at that time.

### UNITED KINGDOM

### Terms and Conditions

**Tax Obligations.** The following provision supplements <u>Section 3 of the Notice</u> as supplemented by <u>Section 2 of Appendix A</u> (Tax Consequences, Withholding, and Liability):

If payment or withholding of any income tax liability arising in connection with Participant's participation in the Plan is not made by Participant to the Employer within ninety (90) days of the end of the U.K. tax year during which the event giving rise to the income tax liability occurs or such other period specified in Section 222(1)(c) of the U.K. Income Tax (Earnings and Pensions) Act 2003 (the "Due Date"), Participant understands and agrees that the amount of any uncollected income tax will constitute a loan owed by Participant to the Company and/or the Employer, effective on the Due Date. Participant further understands and agrees that the loan will bear interest at the then-current official rate of Her Majesty's Revenue and Customs ("HMRC"), it will be immediately due and repayable by Participant, and the Company and/or the Employer may recover it at any time thereafter by any of the means referred to in the Plan or this Notice.

Notwithstanding the foregoing, if Participant is a director or an executive officer of the Company (within the meaning of such terms for purposes of Section 13(k) of the Exchange Act), Participant will not be eligible for such a loan to cover the income tax liability. In the event that Participant is a director or executive officer and the income tax is not collected from or paid by Participant by the Due Date, Participant understands that the amount of any uncollected income tax may constitute an additional benefit to Participant on which additional income tax and National Insurance Contributions may be payable. Participant understands and agrees that Participant will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or the Employer (as appropriate) for the value of any employee National Insurance Contributions ("NICs") due on this additional benefit which the Company or the Employer may recover from Participant by any of the means referred to in the Plan or this Notice.

**Joint Election for Transfer of Liability for Employer National Insurance Contributions.** The following provision supplements <u>Section 3 of the Notice</u> as supplemented by <u>Section 2 of Appendix A</u> (Tax Consequences, Withholding, and Liability):

As a condition of participation in the Plan and the exercise of the Option, Participant agrees to accept any liability for secondary Class 1 NICs that may be payable by the Company or the Employer in connection with the Option and any event giving rise to Tax Obligations (the "Employer NICs"). The Employer NICs may be collected by the Company or the Employer using any of the methods described in the Plan or this Notice.

Without prejudice to the foregoing, Participant agrees to execute a joint election with the Company and/or the Employer (a "Joint Election"), the form of such Joint Election being formally approved by HMRC, and any other consent or elections required to accomplish the transfer of the Employer NICs liability to Participant. Participant further agrees to execute such other elections as may be required by any successor to the Company and/or the Employer for the purpose of continuing the effectiveness of Participant's Joint Election. If Participant does not complete the Joint Election prior to exercise of the Option, or if approval of the Joint Election is withdrawn by HMRC and a new Joint Election is not entered into, the Option shall become null and void and will not vest or become exercisable, without any liability to the Company, the Employer or any Affiliate.

## THE RUBICON PROJECT, INC.

# 2014 INDUCEMENT GRANT EQUITY INCENTIVE PLAN

## RESTRICTED STOCK GRANT NOTICE

### FOR EMPLOYEES

Notice is hereby given of the grant to the employee (the "Participant") named below of a Restricted Stock Award. By receipt and acknowledgment of the Restricted Stock Award, and also by acceptance through performance of the vesting requirements, Participant agrees to the terms and conditions set forth in this Notice and the Company's 2014 Inducement Grant Equity Incentive Plan (the "Plan"), which is hereby incorporated by reference, and confirms that receipt of the Restricted Stock Award is voluntary. This Restricted Stock Award is governed by this Notice and the Plan, and in the event of a conflict between the terms of this Notice and the Plan, the terms of the Plan shall control. Capitalized terms used but not defined in this Notice shall have the meanings given to them in the Plan.

The Company hereby grants to Participant a Restricted Stock Award under the Plan (the "Restricted Stock"). Each share of Restricted Stock represents the right to receive one share (a "Share") of the Company's Common Stock, par value \$0.00001 ("Common Stock"), subject to vesting as set forth below and to the terms and conditions of the Plan and this Notice, as follows:

Participant Name:	
Issuance Date:	
Vesting Commencement Date:	
Number of Shares of Restricted Stock:	
Vesting Schedule:	

vesting senedure.

For purposes of this Notice, "Vesting Date" means each May 15 and November 15, and a complete calendar month will begin on the first day of each calendar month and end on the last day of that calendar month. Subject to the Notice and any Separate Agreement (as defined below), and subject to any acceleration provisions in the Plan:

- (i) on the first Vesting Date that is on or immediately following the first anniversary of the Vesting Commencement Date (the "First Vesting Date"), there shall vest a number of Shares of Restricted Stock equal to the sum of (A) 25% of the total number of Shares of Restricted Stock and (B) a number of Shares of Restricted Stock equal to the product of 2.0833% of the total number of Shares of Restricted Stock and the number of complete calendar months, if any, elapsed during the period beginning on the first anniversary of the Vesting Commencement Date and ending on the First Vesting Date;
- (ii) on each of the six Vesting Dates next succeeding the First Vesting Date, there shall vest an additional number of Shares of Restricted Stock equal to 12.5% of the total number of Shares of Restricted Stock, except that the number of Shares of Restricted Stock vesting on the last of such six succeeding Vesting Dates will be less than 12.5% of the total number of Shares of Restricted Stock if and to the extent that the number of Shares of Restricted Stock vesting on the First Vesting Date exceeded 25% of the total number of Shares of Restricted Stock;

- (iii) except as provided in Section 7 below in connection with a termination of Continuous Service without Cause or due to death or Disability, no Shares of Restricted Stock will vest before the First Vesting Date, and vesting of Shares of Restricted Stock will occur only on Vesting Dates, without any ratable vesting for periods of time between Vesting Dates; and
- (iv) in the event that the application of one of the vesting percentages set forth above results in the vesting of a fractional Share, the number of Shares that shall become vested on such Vesting Date shall be rounded to the nearest whole Share.

Subject to Section 7 below, if Participant ceases to remain in Continuous Service for any or no reason before Participant vests in any of the Shares of Restricted Stock, all unvested Restricted Stock and Participant's right to acquire any Shares of Common Stock hereunder will immediately terminate and be forfeited. However, notwithstanding anything herein to the contrary, the vesting of the Restricted Stock shall be subject to any vesting acceleration provisions applicable to the Restricted Stock contained in the Plan and/or any employment or service agreement, offer letter, severance agreement, or any other agreement between Participant and the Company or any Affiliate or Subsidiary (such agreement, a "Separate Agreement"). Furthermore, under all circumstances, the vesting of Restricted Stock shall be subject to the satisfaction of Participant's obligations as set forth in Section 13(b).

Participant acknowledges receipt of a copy of the Plan and represents that Participant has reviewed the Plan and this Notice in their entirety, has had an opportunity to obtain the advice of Participant's separate independent counsel prior to receipt and acknowledgment of this award, and fully understands this Notice and the Plan. Participant further acknowledges that this Notice and the Plan (including any exhibits to each document) and any Separate Agreement (if applicable) set forth the entire understanding between Participant and the Company regarding the Restricted Stock and the Shares issuable upon vesting of the Restricted Stock and supersede all prior oral and written agreements with respect thereto, including, but not limited to, any other agreement or understanding between Participant and the Company relating to Participant's Continuous Service and any termination thereof, compensation, or rights, claims or interests in or to the Restricted Stock or the Shares.

### RESTRICTED STOCK TERMS

- 1. Grant of Restricted Stock. The Company hereby grants to the Participant named in the Notice of Grant a Restricted Stock Award under the Plan, subject to all of the terms and conditions in this Notice, the Plan, and any Separate Agreement (if applicable), all of which is incorporated herein by reference. As of the Issuance Date set forth in a Notice of Grant, the Company issues to you the number of shares of Common Stock as set forth in that Notice of Grant subject to the vesting requirements set forth in that Notice of Grant (each, a "Share" and collectively, the "Shares"). All shares of Restricted Stock issued hereunder shall be held in escrow by an authorized officer of the Company in accordance with the terms of the Joint Escrow Instructions attached hereto as Exhibit A. Participant will have no right to the release of any Shares from the escrow created by the Joint Escrow Instructions (the "Escrow") unless and until the Shares have vested in the manner set forth in Section 4 and the restrictions in Section 15 shall have lapsed.
- 2. Company's Obligation to Pay. Each share of Restricted Stock represents the right to receive an unrestricted Share of Common Stock on the date it vests. Participant will have no right to the lapse of restrictions on any Restricted Stock unless and until the Restricted Stock has vested in the manner set forth in Section 4.

## 3. Participant Representations.

- (a) Participant acknowledges that (i) Participant was and is free to use professional advisors of Participant's choice in connection with this Notice and any grant of Restricted Stock, that Participant understands this Notice and the meaning and consequences of receiving grants of Restricted Stock and Shares released from the Escrow upon vesting of Restricted Stock; and (ii) Participant has not received and is not relying, and will not rely, upon any advice, representations or assurances made by or on behalf of the Company or any of its Affiliates or any employee of or counsel to the Company or any of its Affiliates regarding any tax or other effects or implications of the Restricted Stock or Shares or other matters contemplated by this award of Restricted Stock.
- (b) Participant is aware of the Company's business affairs and financial condition and understands that an investment in the Shares involves a high degree of risk. Participant is aware of the restrictions on transferability of the Restricted Stock, including that Participant may not be able to sell or dispose of them or use them as collateral for loans.
- (c) If at the time of issuance of any Restricted Stock, there is not in effect under the Securities Act of 1933, as amended (the "Securities Act") a registration statement covering the Shares to be issued, and available for delivery a prospectus meeting the requirements of Section 10(a)(3) of the Securities Act, Participant shall, if required by the Company, as a condition to issuance of the Shares, (i) deliver to the Company Participant's Investment Representation Statement in the form attached hereto as <a href="Exhibit B">Exhibit B</a>; and/or (ii) make appropriate representations in a form satisfactory to the Company that such Shares will not be sold other than (A) pursuant to an effective registration statement under the Securities Act, or an applicable exemption from the registration requirements of such Act; (B) in compliance with all applicable state securities laws and regulations; and (C) in compliance with all terms and conditions of the Plan, this Notice, and any other written agreement between Participant and the Company or any of its Affiliates.
- **4.** Vesting Schedule. Except as provided in <u>Section 6</u>, and subject to <u>Section 7</u>, the Restricted Stock will vest in accordance with the vesting schedule set forth above in this Notice.

- 5. Lock-Up. In connection with any underwritten public offering by the Company of its equity securities pursuant to a registration statement filed under the Securities Act, upon the request of the Company or the underwriters managing such offering, during the Lock-up Period (as defined below) Participant shall not, without the prior written consent of the Company or its underwriters, directly or indirectly sell, make any short sale of, loan, hypothecate, pledge, offer, grant or sell any option or other contract for the purchase of, purchase any option or other contract for the sale of, enter into any swap, hedging or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of, or otherwise dispose of or transfer, or agree to engage in any of the foregoing transactions with respect to, any Shares or other securities into which the Shares may be converted or that are issued in respect of the Shares (other than those included in the registration). For this purpose, the "Lock-up Period" means such period of time after the effective date of the registration as is requested by the Company or the underwriters; provided that such period shall not exceed 180 days (or such additional period as may reasonably be requested by the Company or such underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports or (ii) analyst recommendations and opinions, including (without limitation) the restrictions set forth in Rule 2711(f)(4) of the National Association of Securities Dealers and Rule 472(f)(4) of the New York Stock Exchange, as amended, or any similar successor rules). The Company's underwriters shall be beneficiaries of the provision set forth in this Section 5, and Participant shall execute and deliver such agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In addition, if requested by the Company or the underwriters of Common Stock (or other securities) of the Company, Participant shall provide, within ten (10) days of such request, such information as may be required or reasonably requested by the Company or the underwriters in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Securities Act. The obligations described in this Section 5 shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a Commission Rule 145 transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions with respect to the Shares (or other securities) subject to the foregoing restriction until the end of said one hundred and eighty (180) day (or other) period. Participant agrees, and will cause any transferee to agree, that any transferee of some or all of the Shares shall be bound by this Section 5.
- **6. Release from Escrow after Vesting.** Subject to Section 9, any restrictions that lapse with respect to shares of Restricted Stock upon vesting will lapse with respect to whole Shares.

It is the intent of this Agreement that the issuance of Restricted Shares be exempt from the requirements of Section 409A pursuant to the regulations promulgated so that none of the Shares granted under the award of Restricted Stock will be subject to the additional tax imposed under Section 409A, and any ambiguities herein will be interpreted to so comply. For purposes of this Agreement, "Section 409A" means Section 409A of the Code, and any proposed, temporary or final Treasury Regulations and Internal Revenue Service guidance thereunder, as each may be amended from time to time.

7. Forfeiture Upon Termination of Continuous Service. Except as otherwise provided in the vesting schedule set forth above in this Notice or in a Separate Agreement, but notwithstanding any contrary provision of this Notice, if Participant ceases to remain in Continuous Service at any time for any reason other than due to a termination of Continuous Service by the Company without Cause or due to Participant's death or Disability, in either case on or after the first vesting date for the Restricted Stock, the then-unvested Restricted Stock will thereupon terminate, be forfeited and automatically reacquired by the Company at no cost to the Company and Participant will have no further rights with respect to such forfeited Shares. Upon a termination of Continuous Service by the Company without Cause or due to

Participant's death or Disability, in either case on or after the first vesting date for Restricted Stock, a number of additional Shares of Restricted Stock shall become vested as of the date of such termination of Continuous Service equal to the product obtained by multiplying the number of Shares of Restricted Stock scheduled to vest on the scheduled vesting date next succeeding the date of termination of Continuous Service and a fraction, the numerator of which is the number of full months from the vesting date immediately preceding the date of termination of Continuous Service to the date of termination of Continuous Service to the date of termination of Continuous Service to the scheduled vesting date next following the date of termination of Continuous Service. For these purposes, a full month means the period from the date of one calendar month to the same date the next calendar month (e.g. from May 15 to June 15), or the last day of the next calendar month if the date is the 29th, 30th, or 31st and the next calendar month does not have at least 29, 30 or 31 days, as the case may be. Any Restricted Stock remaining unvested after such pro rata acceleration of vesting shall terminate, be forfeited and reacquired by the Company at no cost to the Company and Participant will have no further rights with respect to such forfeited Shares.

8. Death of Participant. Any distribution or delivery of Shares to be made to Participant pursuant to this Notice (including the Joint Escrow Instructions) will, if Participant is then deceased, be made to Participant's designated beneficiary, or if no beneficiary survives Participant, the administrator or executor of Participant's estate. Any such transferee must furnish the Company with written notice of his or her status as transferee and evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.

### 9. Tax Consequences, Withholding, and Liability.

- (a) Participant understands that Participant may suffer adverse tax consequences as a result of the grant or vesting of the Restricted Stock and issuance and/or disposition of the Shares. Participant understands that the actual tax consequences associated with the Restricted Stock and Shares are complicated and depend, in part, on Participant's specific situation and may also depend on the resolution of currently uncertain tax law and other variables not within the control of the Company. THEREFORE, PARTICIPANT SHOULD SEEK INDEPENDENT ADVICE REGARDING THE APPLICABLE PROVISIONS OF THE FEDERAL TAX LAW AND THE TAX LAWS OF ANY MUNICIPALITY, STATE OR FOREIGN COUNTRY TO WHICH PARTICIPANT IS SUBJECT. By receiving and acknowledging this grant of Restricted Stock, Participant acknowledges and agrees that Participant has either consulted with a competent tax advisor independent of the Company to obtain tax advice concerning the Restricted Stock and Shares in light of Participant's specific situation or has had the opportunity to consult with such a tax advisor and has chosen not to do so. Neither the Company nor any of its employees, counsel, or agents has provided to Participant, and Participant has not relied upon from the Company or any of its employees, counsel, or agents, any written or oral advice or representation regarding the U.S. federal, state, local and foreign tax consequences of the receipt, ownership and vesting of the Restricted Stock, the other transactions contemplated by this Notice, or the value of the Company or the Restricted Stock at any time. With respect to such matters, Participant relies solely on Participant's own advisors.
- (b) Participant (and not the Company) shall be responsible for Participant's own tax liability that may arise as a result of the receipt, ownership and vesting of the Restricted Stock, the issuance of Shares pursuant to the award of Restricted Stock, or the other transactions contemplated by this Notice. Pursuant to such procedures as the Plan administrator may specify from time to time, the Company shall satisfy its obligations to pay withholding taxes or other tax deposits in connection with the

receipt, ownership and/or vesting of the Restricted Stock, the issuance of Shares pursuant to the award of Restricted Stock, or the other transactions contemplated by this Notice in the minimum amount required to satisfy such obligations in accordance with applicable law or regulation (the "Tax Obligations"). If amounts paid by the Company in respect of Tax Obligations are less than Participant's tax obligations, Participant is solely responsible for any additional taxes due. If amounts paid by the Company in respect of Tax Obligations exceed Participant's tax obligations, Participant's sole recourse will be against the relevant taxing authorities, and the Company and its Affiliates will have no obligation to issue additional Shares or pay cash to Participant in respect thereof. Participant is responsible for determining Participant's actual income tax liabilities and making appropriate payments to the relevant taxing authorities to fulfill Participant's tax obligations and avoid interest and penalties.

(c) Payment by the Company of the Tax Obligations will result in a commensurate obligation of Participant to pay, or cause to be paid, to the Company or its Affiliate the amount of Tax Obligations so paid, and the Company shall not be required to release any of the affected Shares from the Escrow and the Company shall not be obligated to deliver any pecuniary interest in the Shares unless and until Participant has satisfied this obligation. The Plan administrator, in its sole discretion and pursuant to such procedures as it may specify from time to time, may permit Participant to satisfy the Tax Obligations, in whole or in part (without limitation) by any of the following means or any combination of two or more of the following means: (i) paying cash, (ii) having the Company withhold and retain Shares otherwise deliverable to Participant having a Fair Market Value equal to the amount of such Tax Obligations, (iii) having the Company withhold the amount of such Tax Obligations from Participant's paycheck(s), (iv) delivering to the Company already vested and owned Shares having a Fair Market Value equal to such Tax Obligations, or (v) selling such number of such Shares otherwise deliverable to Participant having an aggregate Fair Market Value equal to the amount of the Tax Obligations through such means as the Company may determine in its sole discretion (whether through a broker or otherwise). To the extent determined appropriate by the Company in its discretion, it shall have the right (but not the obligation) to cause Participant to satisfy any or all Tax Obligations by withholding and retaining Shares otherwise deliverable to Participant having an aggregate Fair Market Value equal to the amount of such Tax Obligations. If, at the time Shares are to be issued, those Shares are not freely tradeable on a national securities exchange or market system (and for this purpose, a blackout pursuant to the Company's insider trading policy will not be considered to render the Shares not freely tradeable), Participant may in Participant's sole discretion satisfy the Tax Obligations by electing to have the Company withhold and retain such number of Shares otherwise deliverable to Participant, and/or by surrendering such number of Shares already delivered to Participant, having an aggregate Fair Market Value equal to the amount of such Tax Obligations. In order to satisfy the Tax Obligations, the Company will not withhold the amount of such Tax Obligations from Participant's paycheck(s) and/or any other amounts payable to Participant unless the net proceeds from any automatic sale of certain shares of the Restricted Stock as set forth in Section (9)(d) below are not sufficient to satisfy such Tax Obligations in their entirety.

(d) In the event that (i) Participant is not subject to the requirements of Section 16 of the Securities Exchange Act of 1934 on a date that the risk of forfeiture to the Company as described in this Notice lapses with respect to some or all of the Restricted Stock ("Lapse Date") and (ii) Participant incurs a tax liability on such Lapse Date as a result of such lapse, then the Applicable Percentage (as defined below) of the Shares issuable pursuant to the award of Restricted Stock with respect to which the risk of forfeiture shall have lapsed on the Lapse Date, shall be sold within an administratively reasonable period of time on or after the Lapse Date by a broker selected or approved by the Company at such fees and pursuant to such rules and process as the Company may reasonably approve. Participant will bear the brokerage fees and other costs associated with sales and related transmission of funds. The net proceeds from such sale shall be remitted to the relevant tax authorities as determined by the Company for Participant's benefit in the amounts directed by the Company, or paid to the Company in reimbursement

of any Tax Obligations paid by the Company, and any remaining net proceeds shall be delivered to Participant or a brokerage account maintained for Participant. For these purposes the "Applicable Percentage" means forty-five percent (45%), provided that the Company may in its discretion from time to time adjust the Applicable Percentage to an amount reasonably expected to be required to satisfy any or all Tax Obligations and selling expenses. Participant shall have no right to affect or influence any adjustments that the Company may elect to make to the Applicable Percentage for this purpose.

- (e) Under Section 83(a) of the Code, Participant will generally be taxed on the shares of Restricted Stock subject to this award on the date(s) such shares of Restricted Stock vest and the forfeiture restrictions lapse, based on their Fair Market Value on such date, at ordinary income rates subject to payroll and withholding tax and tax reporting, as applicable. Under Section 83(b) of the Code, Participant may elect to be taxed on the shares of Restricted Stock on the Issuance Date, based upon their Fair Market Value on such date, at ordinary income rates subject to payroll and withholding tax and tax reporting, as applicable. If Participant elects to accelerate the date on which Participant is taxed on the shares of Restricted Stock under Section 83(b), an election (an "83(b) Election") to such effect must be filed with the Internal Revenue Service within 30 days from the Issuance Date and applicable withholding taxes must be paid to the Company at that time. The foregoing is only a summary of the federal income tax laws that apply to the shares of Restricted Stock under this Agreement and does not purport to be complete. If Participant determines to make an 83(b) Election, it is Participant's responsibility to file such an election with the Internal Revenue Service within the 30-day period after the Issuance Date, to deliver to the Company a signed copy of the 83(b) Election, to file an additional copy of such election form with Participant's federal income tax return for the calendar year in which the Issuance Date occurs, and to pay applicable withholding taxes to the Company at the time that the 83(b) Election is filed with the Internal Revenue Service.
- 10. Rights as Stockholder. Neither Participant nor any person claiming under or through Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares granted hereunder unless and until such Shares have been issued and recorded on the records of the Company or its transfer agents or registrars. No adjustment shall be made for any dividends (ordinary or extraordinary, whether cash, securities, or other property) or distributions or other rights for which the record date is prior to the date Shares are issued, except as provided in Section 12. After such issuance and recordation, Participant will have all the rights of a stockholder of the Company with respect to voting such Shares and receipt of dividends and distributions on such Shares. Any dividends or distributions payable with respect to unvested Restricted Stock will be subject to the same vesting requirements and restrictions as the Shares of Common Stock underlying the Restricted Stock with respect to which they are paid and shall be held in escrow by an authorized officer of the Company in accordance with the terms of the Joint Escrow Instructions attached hereto as Exhibit A.
- 11. No Guarantee of Continued Service. PARTICIPANT ACKNOWLEDGES AND AGREES THAT THE VESTING OF THE RESTRICTED STOCK PURSUANT TO THE VESTING SCHEDULE APPLICABLE THERETO IS EARNED ONLY BY CONTINUOUS SERVICE AT THE WILL OF THE COMPANY (OR THE AFFILIATE OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) AND NOT THROUGH THE ACT OF BEING HIRED, BEING GRANTED A RESTRICTED STOCK AWARD OR ACQUIRING UNRESTRICTED SHARES UPON VESTING OF RESTRICTED STOCK. PARTICIPANT FURTHER ACKNOWLEDGES AND AGREES THAT THIS NOTICE, THE TRANSACTIONS CONTEMPLATED HEREUNDER AND THE VESTING SCHEDULE APPLICABLE TO RESTRICTED STOCK DO NOT CONSTITUTE AN EXPRESS OR IMPLIED PROMISE OF CONTINUED ENGAGEMENT TO PROVIDE SERVICE FOR THE VESTING PERIOD, FOR ANY PERIOD, OR AT ALL, AND SHALL NOT INTERFERE IN ANY WAY WITH PARTICIPANT'S RIGHT OR THE RIGHT OF THE COMPANY (OR THE AFFILIATE

OR SUBSIDIARY EMPLOYING OR RETAINING PARTICIPANT) TO TERMINATE PARTICIPANT'S CONTINUOUS SERVICE AT ANY TIME, FOR ANY REASON OR NO REASON, WITH OR WITHOUT NOTICE, AND WITH OR WITHOUT CAUSE.

12. Capital Structure Adjustments. Except as otherwise provided herein, appropriate and proportionate adjustments shall be made in the number and class of Shares (or any other securities or other property as to which the Shares may be exchanged for, converted into, or otherwise transferred) subject to the award of Restricted Stock in the event of a stock dividend, stock split, reverse stock split, recapitalization, reorganization, merger, consolidation, separation, or like change in the capital structure of the Company that directly affects the class of shares to which such Shares belong.

### 13. Additional Conditions to Issuance of Stock.

(a) Legal and Regulatory Compliance. The issuance of Shares shall be subject to compliance with all applicable requirements of federal, state or foreign law with respect to such securities. If at any time the Company determines, in its discretion, that the listing, registration or qualification of the Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary as a condition to the issuance of Shares to Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. If the Company determines that the issuance of any Shares will violate federal securities laws or other applicable laws or regulations or the requirements of any exchange or market system upon which the Shares are listed, the Company may defer issuance until the earliest date at which the Company reasonably anticipates that the issuance of Shares will no longer cause such violation. The Company will make all reasonable efforts to meet the requirements of any such state or federal law or securities exchange and to obtain any such consent or approval of any such governmental authority, but the inability of the Company to obtain from any regulatory body having jurisdiction the authority, if any, deemed by the Company's legal counsel to be necessary to the lawful issuance of any Shares shall relieve the Company of any liability in respect of the failure to issue such Shares as to which such requisite authority shall not have been obtained. As a condition to the issuance of Shares, the Company may require Participant to satisfy any qualifications that may be necessary or appropriate, to evidence compliance with any applicable law or regulation and to make any representation or warranty with respect thereto as may be requested by the Company.

(b) Obligations to the Company. As a condition to issuance of any shares of Restricted Stock, Participant must enter into the Company's Intellectual Property Assignment and Confidential Information Agreement, or a similar or successor agreement for the protection of the Company's intellectual property and confidential information, in form specified by the Company (the "Proprietary Interests Agreement"), if Participant has not already done so, and Participant's acceptance of Restricted Stock and any Shares will constitute Participant's agreement to the Proprietary Interests Agreement. If Participant breaches in any material respect the Proprietary Interests Agreement or any other contract between Participant and the Company, or Participant's common law duty of confidentiality or trade secret protection, or any Company policy prohibiting misappropriation of property or any illegal or fraudulent acts, the Company may suspend any vesting of any Restricted Stock pending Participant's cure of such breach, and if such breach cannot be cured or is not cured to the Company's reasonable satisfaction within such time not less than twenty (20) days as the Company may specify, the Company may require Participant to forfeit any unvested Restricted Stock and will have no obligation to provide any consideration to Participant in respect thereof.

14. Handling of Shares; Restrictive Legends and Stop-Transfer Orders.

- (a) <u>Certificates or Book Entries</u>. The Company may in its discretion issue physical certificates representing Shares, or cause the Shares to be recorded in book entry or other electronic form and reflected in records maintained by or for the Company. The Secretary of the Company, or such other escrow holder as the Secretary may appoint, shall retain physical custody of any certificate representing Shares that are subject to restrictions on transfer under <u>Section 15</u> of this Agreement.
- (b) <u>Legends</u>. Each certificate or data base entry representing any Shares may be endorsed with legends substantially as set forth below, as well as such other legends as the Company may deem appropriate to implement the provisions of this Notice or comply with applicable laws and regulations:

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO RESTRICTIONS ON TRANSFER FOR A PERIOD OF TIME FOLLOWING THE EFFECTIVE DATE OF ANY UNDERWRITTEN PUBLIC OFFERING OF THE COMPANY'S SECURITIES SET FORTH IN AN AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES AND MAY NOT BE SOLD OR OTHERWISE DISPOSED OF BY THE HOLDER PRIOR TO THE EXPIRATION OF SUCH PERIOD WITHOUT THE CONSENT OF THE COMPANY OR THE MANAGING UNDERWRITER.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN VESTING REQUIREMENTS AND RESTRICTIONS ON TRANSFER SET FORTH IN A RESTRICTED STOCK AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH REQUIREMENTS AND RESTRICTIONS IN FAVOR OF THE ISSUER OR ITS ASSIGNEE(S) ARE BINDING ON THE TRANSFEREES OF THESE SHARES.

- (c) <u>Stop-Transfer Notices</u>. In order to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.
- (d) <u>Refusal to Transfer</u>. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Notice or any other agreement to which the Shares are subject or any laws governing the Shares or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferree to whom such Shares shall have been so transferred.
- 15. Restrictions on Transfer. Except as otherwise expressly provided in this Notice, the Restricted Stock and the rights and privileges conferred by this Notice will not be transferred, assigned, pledged or hypothecated in any way (whether by operation of law or otherwise) and will not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of the Restricted Stock or any right or privilege conferred by this Notice, or upon any attempted sale under any execution, attachment or similar process, this grant and the rights and privileges conferred hereby immediately will become null and void. The Company may impose such restrictions, conditions or limitations as it determines appropriate as to the timing and manner of any resales by the Participant or other subsequent transfers by the Participant of any vested Shares, including without limitation (a) restrictions under an insider trading policy, (b) restrictions designed to delay and/or coordinate the timing and manner of sales by Participant and other holders, and

(c) restrictions as to the use of a specified brokerage firm for such resales or other transfers.

### 16. Additional Agreements.

- (a) <u>Electronic Delivery</u>. The Company may, in its sole discretion, decide to deliver any documents related to Restricted Stock or Shares by electronic means or request Participant's consent to participate in the Plan by electronic means. Participant hereby consents to receive such documents by electronic delivery and agrees to administration of this Notice, the Restricted Stock and the Shares through any on-line or electronic system established and maintained by the Company or another third party designated by the Company.
- (b) Personal Information. To facilitate the administration of the Plan and any successor plan and the terms of this Notice, it may be necessary for the Company (or its payroll administrators) to collect, hold and process certain personal information about Participant, including, but not limited to, Participant's name, home address and telephone number, date of birth, social insurance number or other identification number, salary, nationality, job title, any shares of Company Common Stock owned, relationship to the Company, details of all awards issued under the Plan or any predecessor or successor plan or any other entitlement to shares of Company Common Stock awarded, canceled, exercised, vested, unvested or outstanding in Participant's favor, for the exclusive purpose of implementing, administering and managing the Plan ("Data") and to transfer this Data to certain third parties such as transfer agents, stock plan administrators, and brokers with whom Participant or the Company may elect to deposit any Shares. Participant hereby explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of Participant's Data for the exclusive purpose of implementing, administering and managing Participant's participation in the Plan and any predecessor and successor plan. Participant understands that Data will be transferred to the Company's transfer agent, broker, administrative agents or such other stock plan service provider as may be selected by the Company in the future, which is assisting the Company with the implementation, administration and management of the Plan and any predecessor and successor plan. Participant understands that the recipients of the Data may be located in the United States or elsewhere, and that the recipients' country (e.g., the United States) may have different data privacy laws and protections than Participant's country. Participant understands that if he or she resides outside the United States, he or she may request a list with the names and addresses of any potential recipients of the Data by contacting his or her local human resources representative. The Participant authorizes the Company, the Company's broker, administrative agents, and any other possible recipients which may assist the Company (presently or in the future) with implementing, administering and managing the Plan or any predecessor or successor plan to receive, possess, use, retain and transfer the Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participant's participation in the Plan or any predecessor or successor plan. The Participant understands that Data will be held only as long as is necessary to implement, administer and manage Participant's participation in the Plan or any predecessor or successor plan. The Participant understands if he or she resides outside the United States, he or she may, at any time, view Data, request additional information about the storage and processing of Data, require any necessary amendments to Data or refuse or withdraw the consents herein, in any case without cost, by contacting in writing his or her local human resources representative. Further, Participant understands that he or she is providing the consents herein on a purely voluntary basis. If Participant does not consent, or if Participant later seeks to revoke his or her consent, his or her employment status or service and career with the Company will not be adversely affected; the only adverse consequence of refusing or withdrawing Participant's consent is that the Company would not be able to grant Restricted Stock or other equity awards or administer or maintain such awards. Therefore, Participant understands that refusing or withdrawing his or her consent may affect Participant's ability to participate in the Plan or any successor plan. For more information on the consequences of Participant's refusal to consent or withdrawal of consent, Participant understands that he

or she may contact his or her local human resources representative.

- (c) Proprietary Information. Participant agrees that all financial and other information relating to the Company furnished to Participant constitutes "Proprietary Information" that is the property of the Company. Participant shall hold in confidence and not disclose or, except within the scope of Participant's Continuous Service, use any Proprietary Information. Participant shall not be obligated under this paragraph with respect to information Participant can document is or becomes readily publicly available without restriction through no fault of Participant. Upon termination of Participant's Continuous Service, Participant shall promptly return to Company all items containing or embodying Proprietary Information (including all copies). This paragraph supplements, but does not limit, any other agreement between Participant and the Company, or any applicable law, related to protection, ownership, or use of the Company's information or property.
- (d) <u>Consideration</u>. Except as may otherwise be set forth in the applicable Notice of Grant, Restricted Stock and Shares are issued in consideration of services provided by Participant and/or other benefit to the corporation within the meaning of Section 152 of the General Corporation Law of the State of Delaware; Participant is not required to make any cash payment to the Company in respect of issuance of Restricted Stock or Shares.

## 17. General.

- (a) No Waiver; Remedies. Either party's failure to enforce any provision of this Notice shall not in any way be construed as a waiver of any such provision, or prevent that party from thereafter enforcing such provision and each and every other provision of this Notice. The rights granted both parties herein are cumulative and shall not constitute a waiver of either party's right to assert all other legal remedies available to it under the circumstances.
- (b) <u>Successors and Assigns</u>. The terms of this Notice shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth, the terms of this Notice shall be binding upon Participant and Participant's heirs, executors, administrators, successors and assigns. The rights and obligations of Participant under this Notice may only be assigned with the prior written consent of the Company.
- (c) <u>Notices</u>. Any notice hereunder shall be in writing (which shall include electronic transmission) and shall be deemed received (i) the business day following electronic verification of receipt if sent electronically, (ii) upon personal delivery to the party to whom the notice is directed, (iii) the business day following deposit with a reputable overnight courier, or (iv) five days after deposit in the U.S. mail, First Class with postage prepaid. Notice shall be addressed to the Company at its principal executive office and to Participant at the address that he or she most recently provided to the Company. Participant agrees that it is Participant's responsibility to notify the Company of any changes to his or her mailing address so that Participant may receive any shareholder information to be delivered by regular mail.
- (d) Interpretation. Headings herein are for convenience of reference only, do not constitute a part of this Notice, and will not affect the meaning or interpretation of this Notice. References herein to Sections are references to the referenced Section hereof, unless otherwise specified. The Board or its Committee will have the power to interpret the Plan and this Notice and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any Restricted Stock has vested). All actions taken and all interpretations and determinations made by the

Board or its Committee in good faith will be final and binding upon Participant, the Company and all other interested persons. Neither the Board or its Committee nor any person acting on behalf of the Board or its Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Plan or this Notice.

- (e) <u>Modifications to the Notice</u>. This Notice constitutes the entire understanding of the parties on the subjects covered. Modifications to this Notice can be made only in an express written contract executed by a duly authorized officer of the Company and shall not require the consent of the Participant unless such modification would materially adversely affect the rights of the Participant hereunder. Notwithstanding anything to the contrary in the Plan or this Notice, the Company reserves the right to revise this Notice as it deems necessary or advisable, in its sole discretion and without the consent of Participant, to comply with Section 409A or to otherwise avoid imposition of any additional tax or income recognition under Section 409A in connection to this award of Restricted Stock.
- (f) Governing Law; Severability. This Notice is governed by the internal substantive laws, but not the choice of law rules, of Delaware. If any provision of this Notice becomes or is declared by a court or arbitrator having jurisdiction over a dispute hereunder to be illegal, unenforceable or void, such provision shall be amended to the extent necessary to conform to applicable law so as to be valid and enforceable and to achieve, to the extent possible, the economic, business and other purposes of such illegal, unenforceable, or void provision or, if such provision cannot be so amended without materially altering the intention of the parties, then such provision shall deleted from this Notice and the remainder of this Notice shall continue in full force and effect.
- (g) Entire Agreement. The Plan is incorporated herein by reference. The Plan and this Notice (including the exhibits referenced herein), along with any Separate Agreement (to the extent applicable) constitute the entire understanding between the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof, and may not be modified adversely to the Participant's interest except by means of a writing signed by the Company and Participant. Participant has read and understands the terms and provisions of the Plan and this Notice, and agrees to all the terms and conditions of the Restricted Stock in accordance with the Plan and this Notice.

### **EXHIBIT A**

## EXPLANATORY COVER SHEET

## JOINT ESCROW INSTRUCTIONS

These Joint Escrow Instructions are intended for use with The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan Restricted Stock Agreement (the "Restricted Stock Agreement").

These Joint Escrow Instructions are used for issuances of shares of the Corporation's Common Stock subject to vesting ("Restricted Stock") pursuant to the Restricted Stock Agreement. The Restricted Stock is subject to forfeiture to the Corporation unless and until the Restricted Stock shall have vested in the manner set forth in the Restricted Stock Agreement and the restrictions set forth in the Restricted Stock Agreement shall have lapsed. The Restricted Stock is also subject to various restrictions on transfer as set forth in the Restricted stock Agreement until the time that the Common Stock is publicly traded and any lock-up period has expired or a Change in Control of the Corporation occurs. The Escrow Agent, generally the Secretary, Assistant Secretary or General Counsel of the Corporation, holds any stock certificate or other documentation representing the shares underlying the grant of Restricted Stock in escrow in a secure location. If the Corporation is holding the certificate or other documentation, please use the following procedures:

Get an originally signed copy of the Restricted Stock Agreement and the Joint Escrow Instructions.

Place these original documents, together with any **original stock certificate or other original documentation** representing the escrowed shares **and** a copy of the check used for payment (if applicable) in a secure (preferably locked) location. These documents should be delivered personally to the Escrow Agent. The documents should be in an envelope (one for each grantee) clearly labeled with the grantee's name and the grant number on the outside.

Place a note in any other files or records referring to the Restricted Stock Agreement that the original stock certificate or other documentation has been transferred to the secure location on a specific date. Put a copy of the stock certificate or other documentation, the Restricted Stock Agreement and the Joint Escrow Instructions in a separate file used for day to day administration of the 2014 Inducement Grant Equity Incentive Plan.

Calendar the expiration of the vesting on the administrative calendar so that the shares can be released from escrow in a timely manner. Confirm that the restrictions on transfer have lapsed before releasing any shares from escrow, even vested shares.

### JOINT ESCROW INSTRUCTIONS

[Title of Escrow Agent]

THE RUBICON PROJECT, INC.

12181 BLUFF CREEK DRIVE, 4TH FLOOR

LOS ANGELES, CALIFORNIA 90094

Dear Sir:

As Escrow Agent for both The Rubicon Project, Inc., a Delaware corporation ("Corporation"), and the undersigned grantee of stock of the Corporation ("Grantee"), you are hereby authorized and directed to hold the documents delivered to you pursuant to the terms of that certain The Rubicon Project, Inc. 2014 Inducement Grant Equity Incentive Plan Restricted Stock Agreement ("Agreement"), dated , 20 , to which a copy of these Joint Escrow Instructions is attached as Exhibit A, in accordance with the following instructions:

These Joint Escrow Instructions are used for issuances of shares of the Corporation's Common Stock subject to vesting ("Restricted Stock") pursuant to the Agreement. The Restricted Stock is subject to forfeiture to the Corporation unless and until the Restricted Stock shall have vested in the manner set forth in the Agreement and the restrictions set forth in the Agreement shall have lapsed. At such time, the shares underlying the Restricted Stock shall be released from escrow to the Grantee.

Any dividends or distributions payable with respect to unvested Restricted Stock will be subject to the same restrictions as the shares of Common Stock underlying the Restricted Stock with respect to which they are paid and will be deposited in the Escrow and held by the Escrow Agent, and will be released from the Escrow at the same time as the underlying shares of Restricted Stock.

In the event the Restricted Stock shall fail to vest as set forth in the Agreement, the Corporation or its assignee will give to Grantee and you a written notice specifying the number of shares of stock to be forfeited to the Corporation, the purchase price (if any), and the time for a closing hereunder at the principal office of the Corporation. Grantee and the Corporation hereby irrevocably authorize and direct you to close the transaction contemplated by such notice in accordance with the terms of said notice.

At the closing you are directed (a) to date any stock assignments necessary for the transfer in question, (b) to fill in the number of shares being transferred, and (c) to deliver same, together with any certificate or other documentation evidencing the shares of stock to be transferred, to the Corporation against the simultaneous delivery to you of the purchase price (if any) of the number of shares of stock being forfeited to the Corporation.

Grantee irrevocably authorizes the Corporation to deposit with you any certificates or other documentation evidencing shares of stock to be held by you hereunder and any additions and substitutions to said shares as specified in the Agreement. Grantee does hereby irrevocably constitute and appoint you as Grantee's attorney-in-fact and agent for the term of this escrow to execute with respect to such securities and other property all documents of assignment and/or transfer and all stock certificates or other documentation necessary or appropriate to make all securities negotiable and complete any transaction herein contemplated.

This escrow shall terminate upon vesting of the Restricted Stock but only if the restrictions placed on the Restricted Stock and described in Section 15 of the Agreement relating to restrictions on transfer shall have lapsed. At such time, the shares underlying the Restricted Stock shall be released to the Grantee but only upon Grantee's satisfaction of any and all Tax Obligations (as defined in the Agreement).

If at the time of termination of this escrow you should have in your possession any documents, securities, or other property belonging to Grantee, you shall deliver all of same to Grantee and shall be discharged of all further obligations hereunder; provided, however, that if at the time of termination of this escrow you are advised by the Corporation that the property subject to this escrow is the subject of a pledge or other security agreement, you shall deliver all such property to the pledgeholder or other person designated by the Corporation.

Except as otherwise provided in these Joint Escrow Instructions, your duties hereunder may be altered, amended, modified or revoked only by a writing signed by all of the parties hereto.

You shall be obligated only for the performance of such duties as are specifically set forth herein and may rely and shall be protected in relying or refraining from acting on any instrument reasonably believed by you to be genuine and to have been signed or presented by the proper party or parties or their assignees. You shall not be personally liable for any act you may do or omit to do hereunder as Escrow Agent or as attorney-in-fact for Grantee while acting in good faith and any act done or omitted by you pursuant to the advice of your own attorneys shall be conclusive evidence of such good faith.

You are hereby expressly authorized to disregard any and all warnings given by any of the parties hereto or by any other person or corporation, excepting only orders or process of courts of law, and are hereby expressly authorized to comply with and obey orders, judgments or decrees of any court. In case you obey or comply with any such order, judgment or decree of any court, you shall not be liable to any of the parties hereto or to any other person, firm or corporation by reason of such compliance, notwithstanding any such order, judgment or decree being subsequently reversed, modified, annulled, set aside, vacated or found to have been entered without jurisdiction.

You shall not be liable in any respect on account of the identity, authority or rights of the parties executing or delivering or purporting to execute or deliver the Agreement or any documents or papers deposited or called for hereunder.

You shall not be liable for the outlawing of any rights under any statute of limitations with respect to these Joint Escrow Instructions or any documents deposited with you.

Your responsibilities as Escrow Agent hereunder shall terminate if you shall cease to be [fill in title] of the Corporation or if you shall resign by written notice to each party. In the event of any such termination, the Corporation may appoint any officer or assistant officer of the Corporation as successor Escrow Agent and Grantee hereby confirms the appointment of such successor or successors as Grantee's attorney-in-fact and agent to the full extent of your appointment.

If you reasonably require other or further instruments in connection with these Joint Escrow Instructions or obligations in respect hereto, the necessary parties hereto shall cooperate in furnishing such instruments.

It is understood and agreed that should any dispute arise with respect to the delivery

and/or ownership or right of possession of the securities, you are authorized and directed to retain in your possession without liability to any person all or any part of said securities until such dispute shall have been settled either by mutual written agreement of the parties concerned or by a final order, decree or judgment of a court of competent jurisdiction after the time for appeal has expired and no appeal has been perfected, but you shall be under no duty whatsoever to institute or defend any such proceedings.

Any notice required or permitted hereunder shall be given in writing and shall be deemed effectively given upon personal delivery, delivery by express courier or five days after deposit in the United States Post Office, by registered or certified mail with postage and fees prepaid, addressed to each of the other parties hereunto entitled at the following addresses, or at such other addresses as a party may designate by ten days' advance written notice to each of the other parties hereto:

CORPORATION:	THE RUBICON PROJECT, INC.	
GRANTEE:		
ESCROW AGENT:	[ESCROW AGENT NAME]	

By signing these Joint Escrow Instructions you become a party hereto only for the purpose of said Joint Escrow Instructions; you do not become a party to the Agreement.

You shall be entitled to employ such legal counsel and other experts (including without limitation the firm of Gibson, Dunn & Crutcher LLP) as you may deem necessary properly to advise you in connection with your obligations hereunder. You may rely upon the advice of such counsel, and may pay such counsel reasonable compensation therefor. The Corporation shall be responsible for all fees generated by such legal counsel in connection with your obligations hereunder.

This instrument shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. It is understood and agreed that references to "you" or "your" herein refer to the original Escrow Agent and to any and all successor Escrow Agents. It is understood and agreed that the Corporation may at any time or from time to time assign its rights under the Agreement and these Joint Escrow Instructions in whole or in part.

This Agreement shall be governed by and interpret applied by the California courts to contracts made and to be per	ted and determined in accordance with the laws of the State of California, as such laws are rformed entirely in California by residents of that state.
	Very truly yours,
	THE RUBICON PROJECT, INC.
	Ву
	[NAME AND TITLE]
	GRANTEE:
ESCROW AGENT:	
[NAME]	

### **EXHIBIT B**

## INVESTMENT REPRESENTATION STATEMENT<sup>1</sup>

PARTICIPANT :

COMPANY : THE RUBICON PROJECT, INC.

SECURITY : COMMON STOCK

AMOUNT :

DATE :

In connection with the receipt of the above-listed Securities, the undersigned Participant represents to the Company the following:

- (a) Participant is aware of the Company's business affairs and financial condition and has acquired sufficient information about the Company to reach an informed and knowledgeable decision to acquire the Securities. Participant is acquiring these Securities for investment for Participant's own account only and not with a view to, or for resale in connection with, any "distribution" thereof within the meaning of the Securities Act of 1933, as amended (the "Securities Act").
- (b) Participant acknowledges and understands that the Securities constitute "restricted securities" under the Securities Act and have not been registered under the Securities Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of Participant's investment intent as expressed herein. In this connection, Participant understands that, in the view of the Securities and Exchange Commission, the statutory basis for such exemption may be unavailable if Participant's representation was predicated solely upon a present intention to hold these Securities for the minimum capital gains period specified under tax statutes, for a deferred sale, for or until an increase or decrease in the market price of the Securities, or for a period of one year or any other fixed period in the future. Participant further understands that the Securities must be held indefinitely unless they are subsequently registered under the Securities Act or an exemption from such registration is available. Participant further acknowledges and understands that the Company is under no obligation to register the Securities. Participant understands that the certificate evidencing the Securities shall be imprinted with any legend required under applicable securities laws and regulations.
- (c) Participant is familiar with the provisions of Rule 701 and Rule 144, each promulgated under the Securities Act, which, in substance, permit limited public resale of "restricted securities" acquired, directly or indirectly from the issuer thereof, in a non-public offering subject to the satisfaction of certain conditions. Rule 701 provides that if the issuer qualifies under Rule 701 at the time of the grant of Restricted Stock to Participant, the grant shall be exempt from registration under the Securities Act. In the event the Company becomes subject to the reporting requirements of Section 13 or

<sup>1</sup> Note: This Exhibit B will only be relevant for Restricted Stock granted before the Issuer S-8 is filed, and then only until such time as a resale S-8 is filed. Once the resale S-8 has been filed, then the securities will as a general rule, no longer be treated as "restricted securities" within the meaning of Rule 144, although for affiliates they will be "control securities" under the same rule.

15(d) of the Securities Exchange Act of 1934, ninety (90) days thereafter (or such longer period as any market stand-off agreement may require) the Securities exempt under Rule 701 may be resold, subject to the satisfaction of the applicable conditions specified by Rule 144, including in the case of affiliates (1) the availability of certain public information about the Company, (2) the amount of Securities being sold during any three (3) month period not exceeding specified limitations, (3) the resale being made in an unsolicited "broker's transaction", transactions directly with a "market maker" or "riskless principal transactions" (as those terms are defined under the Securities Exchange Act of 1934) and (4) the timely filing of a Form 144, if applicable.

If the Company does not qualify under Rule 701 at the time of grant of Restricted Stock, then the Securities may be resold in certain limited circumstances subject to the provisions of Rule 144, which may require (i) the availability of current public information about the Company; (ii) the resale to occur more than a specified period after the purchase and full payment (within the meaning of Rule 144) for the Securities; and (iii) in the case of the sale of Securities by an affiliate, the satisfaction of the conditions set forth in sections (2), (3) and (4) of the paragraph immediately above.

(d) Participant further understands that if all of the applicable requirements of Rule 701 or 144 are not satisfied, registration under the Securities Act, compliance with Regulation A, or some other registration exemption shall be required; and that, notwithstanding the fact that Rules 144 and 701 are not exclusive, the Staff of the Securities and Exchange Commission has expressed its opinion that persons proposing to sell private placement securities other than in a registered offering and otherwise than pursuant to Rules 144 or 701 shall have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk. Participant understands that no assurances can be given that any such other registration exemption shall be available in such event.

Signature		
Print Name		



December 19, 2014

The Rubicon Project, Inc. 12181 Bluff Creek Drive, 4th Floor Los Angeles, CA 90094

Re: The Rubicon Project, Inc. Registration Statement on Form S-8

#### Ladies and Gentlemen:

I have examined the Registration Statement on Form S-8 (the "Registration Statement") of The Rubicon Project, Inc., a Delaware corporation (the "Company"), to be filed with the Securities and Exchange Commission (the "Commission") pursuant to the Securities Act of 1933, as amended (the "Securities Act"), in connection with the offering by the Company of up to 318,685 shares of the Company's common stock, par value \$0.00001 per share ("Common Stock"), that are subject to issuance upon the exercise of options granted under the isocket, inc. 2009 Equity Incentive Plan (the "iSocket Plan") and 1,000,000 shares of Common Stock (together with the shares issuable under the iSocket Plan, the "Shares") that are subject to issuance pursuant to awards under the Company's 2014 Inducement Grant Equity Incentive Plan (the "Inducement Plan," and together with the iSocket Plan, each, a "Plan" and together, the "Plans"). The options issuable under the iSocket Plan were assumed by the Company pursuant to the terms of that certain Merger Agreement, dated as of November 17, 2014, pursuant to which a wholly owned subsidiary of the Company merged with and into isocket, inc. ("iSocket") with iSocket becoming a wholly owned subsidiary of the Company.

In arriving at the opinion expressed below, I have examined the originals, or photostatic or certified copies, of such records of the Company and certificates of officers of the Company and of public officials and such other documents as I have deemed relevant and necessary as the basis for the opinions set forth below. In my examination, I have assumed the genuineness of all signatures, the legal capacity and competency of all natural persons, the authenticity of all documents submitted to me as originals and the conformity to original documents of all documents submitted to me as copies.

Based upon the foregoing examination and in reliance thereon, and subject to the assumptions stated herein and in reliance on statements of fact contained in the documents that I have examined, I am of the opinion that the Shares, when issued



and sold in accordance with the terms set forth in the applicable Plan and against payment therefor, will be validly issued, fully paid and non-assessable.

This opinion is limited to the effect of the current state of the Delaware General Corporation Law and the facts as they currently exist. I assume no obligation to revise or supplement this opinion in the event of future changes in such laws or the interpretation thereof or such facts.

I consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, I do not thereby admit that I am within the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations of the Commission.

Very truly yours,	
/s/ Jonathan Feldman	
Jonathan Feldman	

# CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated March 4, 2014, except for the subsequent events described in the third and fourth paragraphs of Note 18 as to which the date is March 18, 2014, relating to the financial statements of The Rubicon Project, Inc. which appears in the Prospectus dated April 1, 2014 filed by The Rubicon Project, Inc. pursuant to Rule 424(b) under the Securities Act of 1933, relating to the Registration Statement on Form S-1 (No. 333-193739). We also consent to the reference to us under the heading "Experts" in the Prospectus incorporated by reference in this Registration Statement on Form S-8.

/s/ PricewaterhouseCoopers LLP

Los Angeles, California December 19, 2014